COLLECTIVE AGREEMENT

between

VANCOUVER ISLAND UNIVERSITY

and

B.C. GENERAL EMPLOYEES' UNION (BCGEU) Faculty – Local 702

Effective from April 1, 2022 to March 31, 2025

250130v2

TABLE OF CONTENTS

DEFINITIONS.		T
ARTICLE 1 - PI	REAMBLE	2
1.1	Purpose of Agreement	2
1.2	Future Legislation	2
1.3	Conflict with Regulations	2
1.4	Use of Singular Terms	2
1.5	Human Rights Act	
1.6	Reduction in Salary or Benefits	2
ARTICLE 2 - U	NION RECOGNITION AND RIGHTS	3
2.1	Bargaining Agent Recognition	
2.2	Correspondence	
2.3	No Other Agreement	
2.4	No Discrimination for Union Activity	
2.5	Recognition and Rights of Stewards	
2.6	Bulletin Boards	4
2.7	Union Insignia	4
2.8	Right to Refuse to Cross Picket Lines	4
2.9	Time Off for Union Business	4
2.10	Committee Representation	5
2.11	Selection of University President, Academic Administrators and Chairs	
2.12	Appointment of a Faculty member as a Dean	
ARTICLE 3 - U	NION SECURITY	7
ARTICLE 4 - CI	HECK-OFF OF UNION DUES	7
ARTICLE 5 - EI	MPLOYER AND UNION SHALL ACQUAINT NEW FACULTY MEMBERS	8
ARTICLE 6 - EI	MPLOYER'S RIGHTS	8
ARTICLE 7 - EI	MPLOYER-UNION RELATIONS	8
7.1	Representation at Meetings with the Employer	
7.2	Union Bargaining Committees	•••••
7.3		8
	5 5	
7.4	Union Representatives	8
7.4 7.5	Union Representatives Joint Administration and Dispute Resolution Committee	
7.5	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information	
7.5 7.6	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database	
7.5 7.6 ARTICLE 8 - R	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT	8 9 10 10 11
7.5 7.6 ARTICLE 8 - RI 8.1	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment	8 9 10 10 11
7.5 7.6 ARTICLE 8 - R 8.1 8.2	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code)	8 9 10 10 11 11
7.5 7.6 ARTICLE 8 - RI 8.1 8.2 8.3	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment	8 9 10 11 11 11 12
7.5 7.6 ARTICLE 8 - R 8.1 8.2 8.3 8.3 8.4	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code)	
7.5 7.6 ARTICLE 8 - R 8.1 8.2 8.3 8.4 8.5	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings	
7.5 7.6 ARTICLE 8 - RI 8.1 8.2 8.3 8.4 8.5 8.6	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings Rights of the Parties	
7.5 7.6 ARTICLE 8 - R 8.1 8.2 8.3 8.4 8.5 8.6 8.7	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings Rights of the Parties False Complaints, Breaches of Confidentiality and Retaliatory Action	
7.5 7.6 ARTICLE 8 - R 8.1 8.2 8.3 8.4 8.5 8.6 8.7 8.8	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings Rights of the Parties False Complaints, Breaches of Confidentiality and Retaliatory Action Discussion	
7.5 7.6 ARTICLE 8 - R 8.1 8.2 8.3 8.4 8.5 8.6 8.7	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings Rights of the Parties False Complaints, Breaches of Confidentiality and Retaliatory Action	
7.5 7.6 ARTICLE 8 - RI 8.1 8.2 8.3 8.4 8.5 8.6 8.7 8.8 8.9	Union Representatives Joint Administration and Dispute Resolution Committee Technical Information Human Resources Database ESPECTFUL WORKING ENVIRONMENT Statement of Commitment Discrimination and Harassment (BC Human Rights Code) Workplace Bullying and Personal Harassment Procedures for Discrimination and Harassment (BC Human Rights Code) Findings Rights of the Parties False Complaints, Breaches of Confidentiality and Retaliatory Action Discussion	

9.1 Definition

	9.2	Step 1	16
	9.3	Time Limits to Present Initial Grievance	16
	9.4	Step 2	16
	9.5	Time Limit to Reply at Step 2	16
	9.6	Step 3	
	9.7	Time Limit to Reply at Step 3	16
	9.8	Failure to Act	17
	9.9	Time Limit to Submit to Arbitration	17
	9.10	Amending of Time Limits	17
	9.11	Dismissal or Suspension Grievance	17
	9.12	Deviation from Grievance Procedure	17
	9.13	Policy Grievance	17
	9.14	Technical Objections to Grievances	
	9.15	Retroactive Settlements	
ΑΒΤΙΟΙ	F 10 - AF	RBITRATION	18
ANTICL	10.1	Notification	
	10.1	Appointment of a Single Arbitrator	
	10.2	Appointment of a Board of Arbitration	
	10.3	Failure to Appoint	
	10.4 10.5	Board Procedure	
	10.5	Amending Time Limits	
	10.0	Appointment of Arbitrator	
	10.7	Expedited Arbitrations	
ARTICL		ISMISSAL, SUSPENSION AND DISCIPLINE	
	11.1	Burden of Proof	
	11.2	Requirement of Writing	
	11.3	Dismissal and Suspension Grievance	
	11.4	Right to Grieve Other Disciplinary Action	
	11.5	Evaluation Reports	
	11.6	Union Representation	
	11.7	Access to Personnel Record	22
ARTICL	.E 12 - SE	ENIORITY, LAYOFF AND RECALL, SEVERANCE, BUMPING RIGHTS	22
	12.1	Seniority Defined	22
	12.2	Seniority List	23
	12.3	Loss of Seniority	23
	12.4	Same Seniority Credits	24
	12.5	Pre-Layoff Canvass	24
	12.6	Notice of Layoff	24
	12.7	Layoff, Recall, Bumping Rights of Regular Faculty Members	25
	12.8	Registry of Laid Off Faculty Members	26
	12.9	Severance Pay	29
	12.10	Reinstatement of a Regular Faculty Member	
	12.11	Retraining of Regular Faculty Members	29
	12.12	Continuation of IT Privileges for BCGEU Term Faculty	30
ARTICL	.E 13 - RE	GULARIZATION	30
	13.1	Regularization	

	ARTICLE 14 - W	/ORKLOAD	31
	14.1	Workload Defined	
	14.2	Contact Hours/Workweek	
	14.3	Curriculum Development and Professional Development	33
	14.4	Designated Headquarters	
	14.5	Meal Periods	
	14.6	Rest Periods	35
	14.7	MOU#1 Professional Development and Training	
	ARTICLE 15 - O	VERTIME	35
	15.1	Definitions	
	15.2	Authorization and Application of Overtime	
	15.3	Right to Refuse Overtime	
	15.4	Overtime Entitlement	
	15.5	Overtime Compensation	
	15.6	Overtime Rates	
	15.7	Overtime Exemption	
		AID HOLIDAYS	27
		Paid Holidays	
	16.1 16.2	•	
	16.2	Holidays Falling on Saturday or Sunday	
	16.3	Holiday Falling on a Day of Rest	
	16.4	Holiday Falling on a Scheduled Workday	
		Holiday Coinciding with a Day of Vacation	
	16.6	Paid Holiday Entitlement for Faculty Members Less than Full-time	
		NNUAL VACATIONS	
	17.1	Annual Vacation Entitlement	
	17.2	Utilization of Vacation Credits	
	17.3	Vacation Carry Over and Vacation Payout	
	17.4	Scheduled Vacations	
	17.5	Illness or Injury During Vacation	
	17.6	Vacation Credits Upon Death	
	ARTICLE 18 - S	PECIAL AND OTHER LEAVE	39
	18.1	Definitions	
	18.2	General Leave	39
	18.3	Seniority Accrual	40
	18.4	Retention of Status	40
	18.5	Benefits While on Leave	
	18.6	Bereavement Leave	
	18.7	Family Illness Leave	40
	18.8	Compassionate Care Leave	
	18.9	Special Leave	41
	18.10	Full-Time Union or Public Duties	
	18.11	Jury Duty and Court Appearances	
	18.12	Leave for Writing Examinations	
	18.13	Leave for Taking Courses	
	18.14	Elections	
18.15	Exchange Leave	42	
18.16		Courses42	
18.17	Leave of Absence f	or University Committees43	

24.6 24.7

18.19 On-the-Job Training. 43 18.20 Unassisted Leave/Reduced Workload 43 18.21 Foreign Contract Leave 44 18.22 Deferred Salary Leave 45 18.32 Leave for Domestic and Sexual Violence. 45 18.24 Cultural Leave for Indigenous Employees. 45 18.24 Cultural Leave for Indigenous Employees. 45 19.1 Maternity Leave 45 19.2 Parental Leave 45 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Senicity Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 49 20.3 Unsafety Committee 49 20.4 Injury Pay Provision 49 20.5 Pollution Control 49 20.6 Pollution Control 49 20.7 Investigation of Accident Victims 49 20.8 Pay Provi		18.18	Provincial Articulation	43
18.21 Foreign Contract Leave 44 18.22 Deferred Salary Leave 45 18.23 Leave for Domestic and Sexual Violence 45 18.24 Cultural Leave for Indigenous Employees 45 ARTICLE 19 - MATERNITY AND PARENTAL LEAVE 45 19.1 Maternity Leave 45 19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 48 19.5 Seniority Rights on Re-employment. 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits. 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions. 49 20.4 Injury Pay Provision. 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions		18.19	On-the-Job Training	43
18.22 Deferred Salary Leave 45 18.23 Leave for Domestic and Sexual Violence		18.20	Unassisted Leave/Reduced Workload	43
18.23 Leave for Domestic and Sexual Violence. 45 18.24 Cultural Leave for Indigenous Employees. 45 ARTICLE 19 - MATERNITY AND PARENTAL LEAVE 45 19.1 Maternity Leave 45 19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits. 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions. 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.4 Eligibility 50 22.1 Eligibility 50 22.2 Medicial Services Plan 51		18.21	Foreign Contract Leave	44
18.24 Cultural Leave for Indigenous Employees 45 ARTICLE 19 - MATERNITY AND PARENTAL LEAVE 45 19.1 Maternity Leave 45 19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.2 Health AND SAFETY 48 20.3 Unsafe Work Conditions 49 20.4 Ingiuny Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 20.8 Pay Provision 49 20.8 Pay Provision 51 22.4 Educat Services Plan. 51 22.5		18.22	Deferred Salary Leave	45
ARTICLE 19 - MATERNITY AND PARENTL LEAVE 45 19.1 Maternity Leave 45 19.2 Parental Leave 45 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 50 21. Eligibility 50 22.2 Medical Services		18.23	Leave for Domestic and Sexual Violence	45
19.1 Maternity Leave 45 19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.1 Conditions 48 20.1 Conditions 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 21.1 CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan. 51 22.4 Dental Care Plan.		18.24	Cultural Leave for Indigenous Employees	45
19.1 Maternity Leave 45 19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.1 Conditions 48 20.1 Conditions 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 21.1 CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan. 51 22.4 Dental Care Plan.	ΔΡΤΙΟ	IF 19 - M	ΙΔΤΕΡΝΙΤΥ ΔΝΟ ΡΔΡΕΝΤΔΙ Ι ΕΔΥΕ	45
19.2 Parental Leave 46 19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 19.7 Sick Leave Credits 48 20.1 Conditions 49 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 21.7 Investigation of Accidents 50 22.1 Eligibility 50 22.1 Eligibility 50 22.2 Health And DISURANCE BENEFITS 50 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22	/			
19.3 Benefits Continuation 47 19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 21.6 Pollution Control 49 22.7 Itigibility 50 22.1 Edical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5<			,	
19.4 Supplemental Employment Benefit for Maternity and Parental Leave 47 19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 21.1 Eligibility 50 22.1 Eligibility 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 53 </td <td></td> <td>-</td> <td></td> <td>-</td>		-		-
19.5 Seniority Rights on Re-employment 48 19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits 48 19.7 Sick Leave Credits 48 ARTICLE 20 - HEALTH AND SAFETY 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 50 21.1 Eligibility 50 22.1 Eligibility 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Heaith Benefits 51 <td< td=""><td></td><td></td><td></td><td></td></td<>				
19.6 Extension Maternity Leave 48 19.7 Sick Leave Credits. 48 ARTICLE 20 - HEALTH AND SAFETY 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 21. Eligibility 50 22.1 Eligibility 50 22.1 Eligibility Benefits 51 22.3 Extended Health Benefits 51 22.4 Dettal Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51<				
19.7 Sick Leave Credits 48 ARTICLE 20 - HEALTH AND SAFETY 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 21.7 Eligibility 50 22.1 Eligibility 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.4 Detail Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 53 52 22.7 Medical T				
ARTICLE 20 - HEALTH AND SAFETY 48 20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 50 21.1 Eligibility 50 22.1 Hedical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 <td< td=""><td></td><td></td><td>•</td><td></td></td<>			•	
20.1 Conditions 48 20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.7 Investigation of Accidents 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Di				
20.2 Health and Safety Committee 49 20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.9 Absence Without Pay 53 22.1 Izerination of Coverage 53 22.1 Joint Committee on Benefits Administration 53 22.10 Termination of Coverage 54	ARTIC	-	-	-
20.3 Unsafe Work Conditions 49 20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54				
20.4 Injury Pay Provision 49 20.5 Transportation of Accident Victims 49 20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.1 Ektended Health Benefits 51 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 52 2.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 <td></td> <td></td> <td>•</td> <td></td>			•	
20.5Transportation of Accident Victims4920.6Pollution Control4920.7Investigation of Accidents4920.8Pay Provisions49ARTICLE 21 - CONTRACTING OUT49ARTICLE 22 - HEALTH AND INSURANCE BENEFITS5022.1Eligibility5022.2Medical Services Plan5122.3Extended Health Benefits5122.4Dental Care Plan5122.5Group Life Insurance/Accidental Death and Dismemberment Benefit Plan5122.6Disability Benefits5222.7Medical Travel Expense Entitlement5322.8Coverage During Layoff5322.9Absence Without Pay5322.10Termination of Coverage5322.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES5424.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply5858				
20.6 Pollution Control 49 20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 2.7 Medical Travel Expense Entitlement 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 ARTICLE 24 - WAGES AND ALLOWANCES 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release		-		
20.7 Investigation of Accidents 49 20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan. 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan. 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 ARTICLE 24 - WAGES AND ALLOWANCES 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 Fi			•	
20.8 Pay Provisions 49 ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 53 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 ARTICLE 24 - WAGES AND ALLOWANCES 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 First Aid 55 24.5 Salary Placement				
ARTICLE 21 - CONTRACTING OUT 49 ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 ARTICLE 24 - WAGES AND ALLOWANCES 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 First Aid 55 24.5 Salary Placement 55 24.5 Salary Placement <td></td> <td>-</td> <td>•</td> <td></td>		-	•	
ARTICLE 22 - HEALTH AND INSURANCE BENEFITS 50 22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 ARTICLE 24 - WAGES AND ALLOWANCES 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 First Aid 55 24.5 Salary Placement 55 Clothing Supply 58		20.8	Pay Provisions	49
22.1 Eligibility 50 22.2 Medical Services Plan 51 22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 First Aid 55 24.5 Salary Placement 55 Clothing Supply 58				
22.2 Medical Services Plan	ARTIC	LE 21 - CO	ONTRACTING OUT	49
22.3 Extended Health Benefits 51 22.4 Dental Care Plan 51 22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan 51 22.6 Disability Benefits 52 22.7 Medical Travel Expense Entitlement 53 22.8 Coverage During Layoff 53 22.9 Absence Without Pay 53 22.10 Termination of Coverage 53 22.11 Joint Committee on Benefits Administration 53 22.12 ESA Sick Leave 54 ARTICLE 23 - COLLEGE PENSION PLAN 54 24.1 Wages 54 24.2 Paydays 54 24.3 Chairs Stipend and/or Release Time 55 24.4 First Aid 55 24.5 Salary Placement 55 Clothing Supply 58				
22.4 Dental Care Plan		LE 22 - H	EALTH AND INSURANCE BENEFITS	50
22.5Group Life Insurance/Accidental Death and Dismemberment Benefit Plan5122.6Disability Benefits5222.7Medical Travel Expense Entitlement5322.8Coverage During Layoff5322.9Absence Without Pay5322.10Termination of Coverage5322.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN4ARTICLE 24 - WAGES AND ALLOWANCES5424.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply5858		LE 22 - H 22.1	EALTH AND INSURANCE BENEFITS Eligibility	50 50
22.6 Disability Benefits		LE 22 - H 22.1 22.2	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan	50 50 51
22.7Medical Travel Expense Entitlement5322.8Coverage During Layoff5322.9Absence Without Pay5322.10Termination of Coverage5322.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply5858		LE 22 - H 22.1 22.2 22.3	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits	50 51 51
22.8Coverage During Layoff.5322.9Absence Without Pay.5322.10Termination of Coverage5322.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply5858		LE 22 - H 22.1 22.2 22.3 22.4	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan	50 50 51 51 51
22.9Absence Without Pay		LE 22 - HI 22.1 22.2 22.3 22.4 22.5	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan	50 51 51 51 51
22.10Termination of Coverage5322.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58		LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits	50 51 51 51 51 51
22.11Joint Committee on Benefits Administration5322.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLANARTICLE 24 - WAGES AND ALLOWANCES24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58		LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement	50 51 51 51 51 51 52 53
22.12ESA Sick Leave54ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES5424.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58		LE 22 - HI 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff.	50 51 51 51 51 51 52 53
ARTICLE 23 - COLLEGE PENSION PLAN54ARTICLE 24 - WAGES AND ALLOWANCES5424.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58		LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay	50 51 51 51 51 52 53 53 53
ARTICLE 24 - WAGES AND ALLOWANCES		LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay Termination of Coverage	50 51 51 51 51 52 53 53 53
24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58		LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay Termination of Coverage Joint Committee on Benefits Administration	50 51 51 51 51 53 53 53 53
24.1Wages5424.2Paydays5424.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58	ARTIC	LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay Termination of Coverage Joint Committee on Benefits Administration ESA Sick Leave	50 51 51 51 51 53 53 53 53 53
24.2Paydays	ARTIC	LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay Termination of Coverage Joint Committee on Benefits Administration ESA Sick Leave	50 51 51 51 52 53 53 53 53 53 54 54
24.3Chairs Stipend and/or Release Time5524.4First Aid5524.5Salary Placement55Clothing Supply58	ARTIC	LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W	EALTH AND INSURANCE BENEFITS Eligibility Medical Services Plan Extended Health Benefits Dental Care Plan Group Life Insurance/Accidental Death and Dismemberment Benefit Plan Disability Benefits Medical Travel Expense Entitlement Coverage During Layoff Absence Without Pay Termination of Coverage Joint Committee on Benefits Administration ESA Sick Leave DLIEGE PENSION PLAN	50 51 51 51 51 53 53 53 53 53 54 54 54
24.4First Aid	ARTIC	LE 22 - H 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W 24.1	EALTH AND INSURANCE BENEFITS	50 51 51 51 51 53 53 53 53 53 54 54 54
24.5Salary Placement	ARTIC	LE 22 - Hi 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W 24.1 24.2	EALTH AND INSURANCE BENEFITS	50 51 51 51 52 53 53 53 53 53 54 54 54
Clothing Supply 58	ARTIC	LE 22 - Hi 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W 24.1 24.2 24.3	EALTH AND INSURANCE BENEFITS	50 51 51 51 51 53 53 53 53 53 53 54 54 54 54 54 54
	ARTIC	LE 22 - Hi 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W 24.1 24.2 24.3 24.4	EALTH AND INSURANCE BENEFITS	50 51 51 51 51 52 53 53 53 53 53 54 54 54 54 54 55
	ARTIC ARTIC ARTIC	LE 22 - Hi 22.1 22.2 22.3 22.4 22.5 22.6 22.7 22.8 22.9 22.10 22.11 22.12 LE 23 - CO LE 24 - W 24.1 24.2 24.3 24.4 24.5	EALTH AND INSURANCE BENEFITS	50 51 51 51 51 52 53 53 53 53 53 54 54 54 54 54 55

37.1 37.2 37.3 37.4 37.5 37.6

	24.8	Health and Safety Apparel and Equipment	58
	24.9	Independent Study Stipend	
	24.10	Professional Association Fees	59
ARTIC	LE 25 - T	ECHNOLOGICAL CHANGE	59
ARTIC	LE 26 - P	ROGRAM TRANSFERS AND MERGERS	
	26.1	Notice of Program Transfer / Merger	
	26.2	Transfer/Merger Agreements	59
	26.3	Disputes	59
ARTIC	LE 27 - C	OPYRIGHT AND INTELLECTUAL PROPERTY	59
	27.1	Copyright Ownership	59
	27.2	Employer Rights to Materials Copyrighted by Faculty Member(s)	60
	27.3	Faculty Member Rights to Materials Copyrighted by the Employer	60
	27.4	Joint Review	60
ARTIC	LE 28 - D	ISTRIBUTED LEARNING	60
ARTIC	LE 29 - N	ON-INSTRUCTIONAL DUTY DAYS FOR FACULTY MEMBERS	61
ARTIC	LE 30 - P	OSTED JOB COMPETITIONS	61
	30.1	Job Postings	61
	30.2	Qualifications	61
	30.3	Transfers	61
	30.4	Term Appointment Vacancies	62
	30.5	Equity Deserving Groups	62
ARTIC	LE 31 - T	RANSFERS AND SECONDMENTS	62
	31.1	Short Term Transfers	
	31.2	Secondments	
ARTIC	LE 32 - T	RAVEL	62
	32.1	Travel on University Business	
ARTIC	le 33 - IN	IDEMNITY/INSURANCE	62
		LASS SIZE	
ARTIC	LF 35 - T	HE HIGH SCHOOL AT VIU	63
	35.1	Field Trips; Extra Curricular Programs and Activities; Student Teachers; Volunt	
	35.2	Principal's Teaching Assignment	
ARTIC	LE 36 - L/	ABOUR MANAGEMENT COMMITTEE	64
	36.1	Composition of Labour Management Committee	64
	36.2	Meetings	64
	36.3	Chairperson	64
	36.4	Committee Jurisdiction	64
	36.5	Labour Management Minutes	65
ARTIC	LE 37 - IN	ITERNATIONAL EDUCATION	65
General		65	
Expenses		65	
Health and	d Insuran	ce Benefits65	
Emergenci	ies and E	mergency Evacuation66	
Orientatio	n	66	
Return		66	

ARTICLE 38 -	TERM OF AGREEMENT	66
38.1	Duration	
38.2	Notice to Bargain	
38.3	Commencement of Bargaining	
38.4	Changes in Agreement	
38.5	Agreement to Continue in Force	
APPENDIX A	-	69
APPENDIX B		70
EMP	LOYEE AND FAMILY ASSISTANCE PROGRAM	
DEPA	ARTMENTAL LISTING	
APPENDIX F1	L	71
	STRY OF LAID OFF EMPLOYEES – FORM 1	
APPENDIX E2	2	72
	STRY OF LAID OFF EMPLOYEES – FORM 2	
		72
	ICAL TRAVEL REFERRAL BENEFIT	
APPENDIX G		75
DEN	TAL PLAN	75
APPENDIX H		75
	ILY MEMBERS FOR THE PURPOSE OF	-
	CLE 18.8 COMPASSIONATE CARE LEAVE	
BCG	EU Program Norms	
MEMORAND	UM OF UNDERSTANDING 1	78
For C	Courses and Programs Not Approved by Senate	
MEMORAND	UM OF UNDERSTANDING 2	80
-	Programs	
	-	
	NDERSTANDING 1	
Prior	· Learning Assessment	
LETTER OF U	NDERSTANDING 2	
	al Sick Leave and Partial Disability Benefits	
	NDERSTANDING 3	
Educ	ation Support and Professional Development Fund (ESPD)	
LETTER OF U	NDERSTANDING 4	84
Cost	of Living Adjustment	
	GREEMENT 1	Q.1
	ension of Seniority Credit Adjustments	
•	aculty members on LTD beyond Change of Definition	
10116		

LETTER OF AGREEMENT 2	85
LETTER OF AGREEMENT 3	85
Vacation Carry Over	
LETTER OF AGREEMENT 4	85
FTE Calculations	85

DEFINITIONS

"Collective Agreement" means the combination of provisions of the Common Agreement with local provisions that constitute a Collective Agreement between an institution and a local Union.

"Employee" means an active member of the bargaining unit.

"Employer" means Vancouver Island University (VIU).

"Faculty Member" means a person employed by VIU as an instructor, or in an equivalent position. For the purpose of this collective agreement, "faculty member" means "vocational instructor" as described in the Certification issued by the Labour Relations Board on October 22, 1974, and varied on September 11, 2008.

"FTE" is the value of a full-time equivalent workload per annum (see Appendix I).

"Joint Administration and Dispute Resolution Committee" or "JADRC" means the committee established under Article 7.4(a).

"Joint Labour-Management Committee" means a committee formed with equal representation from the Union and the Employer.

"Ministry" means the Ministry of Advanced Education.

"Post-Secondary Employers' Association" or "PSEA" means the employers' association established for post-secondary colleges, institutes, and special purpose, teaching universities under the Public Sector Employers' Act.

"Ratification" means the acceptance by BCGEU, Vancouver Island University and the PSEA of the terms of the agreement.

"Union" means BCGEU certified as the bargaining agent.

Workers' Compensation Board of British Columbia (WCB BC) - WorkSafeBC is the operating name of the Workers' Compensation Board of British Columbia, a statutory agency created by the Workers Compensation Act.

Appointment Status:

"Regular Appointment" - An appointment to a regular position established by Board authority for normal University operations or an appointment to regular status as a result of regularization language. These appointments are subject to a two (2) year probationary period.*

*Change to length of probationary period applies only to probationary periods that commence after the date of ratification of this collective agreement.

"Term Appointment" - An appointment of a specific duration not to exceed twelve (12) months. Such an appointment is not intended to lead to a regular appointment. However, under exceptional circumstances, a further appointment of up to twelve (12) months may be processed.

"Auxiliary Appointment" - An appointment for a period not to exceed thirty (30) assigned days of work. If an assignment is known to be in excess of thirty (30) assigned days of work prior to the appointment, it shall be posted as a term appointment. If an auxiliary appointment exceeds the thirty (30) assigned days, then that appointment shall be made a term appointment retroactively to its original start date. Upon approval from the Union, this appointment shall be confirmed and will not be subject to the posting

process. Auxiliary appointments are as and when required and not intended to be used for ongoing or repeating work.

Employment Status:

"Regular Faculty Member" - Regular faculty member means a faculty member holding a regular appointment.

"Term Faculty Member" - Term faculty member means a faculty member holding a term appointment.

"Auxiliary Faculty Member" - Auxiliary faculty member means a faculty member holding an auxiliary appointment.

Any of the above may be described as part-time if the assignment is less than 1.0 FTE per annum.

ARTICLE 1 - PREAMBLE

1.1 Purpose of Agreement

(a) The purpose of this Agreement is to establish and maintain orderly collective bargaining procedure between the Employer and the Union.

(b) In support of our values of Learning, Discovery, Engagement, Diversity and Sustainability the University will provide to the Union all of the budget submissions from the Deans and Directors as presented to the Vice Presidents.

(c) In order to promote the efficient and effective operation of the institution through the establishment and continuance of harmonious relations and working conditions established under the Collective Agreement, and to assist in the development and expansion of the public post-secondary system, the parties therefore agree to the following terms of contract.

1.2 Future Legislation

In the event that any future legislation renders null and void or materially alters any provision of the Agreement, the remaining provisions shall remain in effect for the term of the Agreement, and the parties hereto shall negotiate a mutually agreeable provision to be substituted for the provision so rendered null and void or materially altered.

1.3 Conflict with Regulations

In the event that there is a conflict between contents of this Agreement and any regulations by the Employer, or on behalf of the Employer, this Agreement shall take precedence over the said regulation.

1.4 Use of Singular Terms

Wherever the singular is used in this Agreement, the same shall be construed as meaning the plural if the context requires unless otherwise specifically stated.

1.5 Human Rights Act

The parties hereto subscribe to the principles of the "Human Rights Act of British Columbia".

1.6 Reduction in Salary or Benefits

No faculty member shall suffer reduction in salary or any benefit as a result of this Agreement.

ARTICLE 2 - UNION RECOGNITION AND RIGHTS

2.1 Bargaining Agent Recognition

(a) The Employer recognizes the B.C. General Employees' Union as the exclusive bargaining agent on behalf of faculty members employed by Vancouver Island University for whom BCGEU has been certified as bargaining agent by the Labour Relations Board of BC.

(b) When the Employer introduces a new program or alters any existing Vancouver Island University program(s), it shall notify the BCGEU in writing at the point a decision by the Provost and Vice-President Academic is made to implement. The written notice shall be provided to the BCGEU Staff Representative and the Bargaining Unit Chairperson and include details of the altered or new program(s).

(c) The Employer shall meet with the BCGEU within 7 days of the written notice to consult on the appropriate bargaining unit jurisdiction of the new or altered program(s) within the University. Should the Union disagree with the jurisdictional placement of the new or altered program(s), it may pursue the matter through the Labour Relations Board.

2.2 Correspondence

The Employer agrees that all correspondence between the Employer and the Union related to matters covered in this Agreement shall be sent to the BCGEU Staff Representative with a copy to the Bargaining Unit Chairperson.

The Employer agrees that a copy of any correspondence between the Employer or department official and any faculty member in the bargaining unit covered by this Agreement, pertaining to the interpretation or application of any clause in this Agreement, shall be forwarded to the BCGEU Staff Representative.

2.3 No Other Agreement

No faculty member covered by this Agreement shall be required or permitted to make a written or oral agreement with the Employer or its representatives which may conflict with the terms of this Agreement.

2.4 No Discrimination for Union Activity

The Employer and the Union agree that there shall be no discrimination, interference, restriction, or coercion exercised or practised with respect to any faculty member for reason of membership or activity in the Union.

2.5 Recognition and Rights of Stewards

The Employer recognizes the Union's right to select stewards to represent faculty members. The Employer and the Union will agree on the number of stewards, taking into account both operational and geographical considerations.

The Union agrees to provide the Employer with a list of the faculty members designated as stewards for each work unit.

A steward, or alternate, shall obtain the permission of the immediate supervisor before leaving work to perform duties as a steward. Such permission shall not be unreasonably withheld. On resuming normal duties, the steward shall notify the supervisor.

The duties of stewards shall include:

(a) investigation of complaints of an urgent nature;

(b) investigation of grievances and assisting any faculty member which the steward represents in preparing and presenting a grievance in accordance with the grievance procedure;

- (c) supervision of ballot-boxes and other related functions during ratification votes;
- (d) carrying out duties within the realm of safety responsibilities;
- (e) attending meetings called by Management.

2.6 Bulletin Boards

The Employer shall provide secure bulletin board facilities for Union use. The sites are to be determined by mutual agreement. The use of such bulletin board facilities shall be restricted to the business affairs of the Union. In addition, the Employer shall provide access to electronic communication (e.g. email list serve).

2.7 Union Insignia

A Union member shall have the right to wear or display the recognized insignia of the Union.

2.8 Right to Refuse to Cross Picket Lines

(a) All faculty members covered by this Agreement shall have the right to refuse to cross a picket line arising out of a dispute as defined in the relevant legislation. Any faculty member failing to report for duty shall be considered to be absent without pay.

(b) Failure to cross a picket line encountered in carrying out the Employer's business shall not be considered a violation of this Agreement nor shall it be grounds for disciplinary action.

2.9 Time Off for Union Business

(a) Without Pay -- Leave of absence without pay and without loss of seniority will be granted:

(1) to an elected or appointed representative of the Union to attend conventions of the Union and bodies to which the Union is affiliated;

(2) for elected or appointed representatives of the Union to attend to Union business which requires them to leave their premises of employment;

(3) for faculty members who are representatives of the Union on a bargaining committee to attend meetings of the bargaining committee.

(b) A regular faculty member who is on leave of absence without pay in an elected or appointed position of the Union shall have accrued seniority banked.

(c) With Pay -- Leave of absence with pay and without loss of seniority will be granted:

(1) to faculty members who are representatives of the Union to leave their employment to carry on negotiations with the Employer;

(2) to stewards, or their alternatives, to perform their duties pursuant to Section 2.5;

(3) to faculty members called to appear as witnesses before an Arbitration Board.

(4) Meetings between representatives of the Union and the Employer will be scheduled at times mutually agreeable to the parties. Reasonable effort shall be made to hold such meetings at times that do not conflict with assigned duties.

Where such meetings cannot be scheduled at times that do not conflict with assigned duties, the Employer will grant a leave of absence without loss of pay or other entitlements for the purpose of attending such meetings to the total equivalent of one-quarter full-time equivalent per annum.

Where such leave is granted, the Employer will replace the faculty member as necessary.

This clause may be utilized by the Union to ensure adequate representation by the Union with respect to issues that affect the Employer or the post-secondary system. To facilitate the administration of this provision, the Union will ensure that the Employer is advised of the eligible leaves to be taken.

The Union may designate a person(s) who will be entitled to Union leave under this Article and will advise the Employer of the amount of the leave to be taken. The amount of the entitlement is one quarter ($\frac{1}{2}$) of a full-time equivalent per annum, without loss of pay or other entitlement.

Costs arising from this provision will not be charged against the program area of the participating union representative.

This provision will not be utilized where existing Employer-paid release time arrangements exceed this one-quarter (¼) full-time equivalent entitlement.

It is understood that faculty members granted leave-of-absence pursuant to this Article shall receive their current rate of pay while on leave of absence with pay. Leave of absence granted under this Article shall include sufficient travel time. The Employer agrees that any of the above leaves of absence shall not unreasonably be withheld. To facilitate the administration of paragraph (a) of this section, when leave without pay is granted, the leave shall be given with pay and the Union shall reimburse the Employer for the appropriate salary costs, including travel time, incurred.

2.10 Committee Representation

Whenever representatives from the Union are required to be on any committee, the Union shall be consulted before representatives are appointed.

2.11 Selection of University President, Academic Administrators and Chairs

(a) Selection of University President

(1) Selection and appointment of University President is the responsibility of the University Board. It is recognized, that the President should be appointed only after wide consultation within (and perhaps outside) the University community. Therefore, in order to assist the University Board in this task, a search committee, which will include Union representation, will be constituted. Such representation will be appointed by the Union from the bargaining unit.

(2) Once the committee size is determined by the University Board, it is recommended that union representation be proportional to the bargaining units' respective memberships. In addition, it is recommended that there is no less than one BCGEU appointee in addition to the named appointees as set out in the VIU Board Bylaw #17 – Recruitment of the President (dated March 25, 2010).

(3) At the time of the creation of the search committee, the respective membership of each bargaining unit shall be confirmed on the basis of a count of employees. The accounting shall be used to establish each bargaining unit's respective membership stated in 2.11(a)(2).

(b) Selection of the Provost and Vice-President Academic, Vice Presidents, and Associate Vice Presidents with instructional responsibility

(1) Selection and appointment of senior academic positions, (e.g. The Provost and Vice President Academic, Vice Presidents, and Associate Vice Presidents with academic responsibilities) is the duty and responsibility of the President. It is recognized that such persons should be appointed only after wide consultation within the University community. Therefore, in order to assist the President in this task, a search committee, which will include Union representation, will be constituted.

(c) Appointment of other Academic Administrative Positions (e.g. Deans, Associate Deans, Academic Administrators and Directors with Academic Responsibilities)

(1) It is the responsibility of the search committee to advise the President on the selection and appointment of Academic Administrators. It is recognized that such persons should be appointed only after wide consultation within the University community, and specifically after consultation with the members of the Faculty or areas(s) affected or impacted by the position under consideration.

(2) In order to assist the University in this task, a search committee will be constituted which shall include BCGEU faculty representation in proportion to the union members affected or impacted by the position under consideration. The Employer representatives on the committee shall be at least one less than the total union members (CUPE, BCGEU, and VIUFA) on the committee.

(3) At the time of the creation of the search committee, the respective membership of each bargaining unit shall be confirmed on the basis of a count of faculty members in the affected area or Faculty. The accounting shall be used to establish each bargaining unit's respective membership stated in 2.11(c)(2).

(d) Selection and appointment of Program Chairs

(1) Selection and appointment of Program Chairs is the responsibility of the appropriate administrator. A Union representative, appointed by the Union and the appropriate administrator, will participate in the process of selection. The Union will consider conflicts of interest when appointing their representative.

(2) These appointments will be for a maximum of two (2) years with two (2) one (1)-year extensions possible at the discretion of the appropriate administrator.

(3) No faculty member will be appointed, or extended, as program chair without the agreement of the faculty member.

2.12 Appointment of a Faculty member as a Dean

(a) Dean's right to return to a BCGEU regular position.

(1) A Dean appointed from within the BCGEU regular membership shall have the right to return to a position similar to their original BCGEU regular position on the completion of their appointment provided the duration of the appointment, including renewals, does not exceed five (5) years.

(2) BCGEU regular seniority accrual will be paused for the period of the appointment.

(b) Instructional Assignment for a Dean

(1) The Dean may be assigned a BCGEU instructional workload in the program area(s), to a maximum of one (1) course per semester, provided such assignment does not result in a reduction to the instructional workload for existing BCGEU regular faculty members in the decanal program area(s).

(2) The instructional assignment for the Dean will result in no extra remuneration for doing so, including no BCGEU payment for that instruction.

ARTICLE 3 - UNION SECURITY

(a) All faculty members in the bargaining unit who, on October 22, 1974, were members of the Union, or thereafter become members of the Union, shall, as a condition of continued employment, maintain such membership.

(b) All faculty members hired on or after April 21, 1975, shall, as a condition of continued employment, become members of the Union, and maintain such membership upon completion of thirty (30) days as a faculty member.

ARTICLE 4 - CHECK-OFF OF UNION DUES

(a) The Employer shall, as a condition of employment, deduct from the biweekly wages of each faculty member in the bargaining unit, whether or not the faculty member is a member of the Union, the amount of the regular biweekly dues payable to the Union by a member of the Union. The faculty member shall sign the authorization form provided by the Employer.

(b) The Employer shall deduct from any faculty member who is a member of the Union any assessments levied in accordance with the Union Constitution and (or) Bylaws and owing by the faculty member to the Union.

(c) Deductions shall be made biweekly and membership dues or payments in lieu thereof shall be considered as owing in the month for which they are so deducted.

(d) All deductions shall be remitted to the President of the Union not later than twenty-eight (28) days after the date of deduction and the Employer shall also provide a list of names of those faculty members from whose salaries such deductions have been made, together with the amounts deducted from each faculty member.

(e) Before the Employer is obliged to deduct any amount under Section (a) of this Article, the Union must advise the Employer, in writing, of the amount of its regular biweekly dues. The amount so advised shall continue to be the amount to be deducted until changed by further written notice to the Employer, signed by the President of the Union. Upon receipt of such notice, such changed amount shall be the amount deducted.

(f) The Employer shall supply each faculty member, without charge, a receipt for income tax purposes in the amount of the deductions paid to the Union by the faculty member in the previous calendar year. Such receipts shall be provided to the faculty members prior to March 1 of the succeeding year.

ARTICLE 5 - EMPLOYER AND UNION SHALL ACQUAINT NEW FACULTY MEMBERS

(a) The Employer agrees to acquaint new faculty members with the fact that a Collective Agreement is in effect, and with the conditions of employment set out in the Articles dealing with Union security and dues check-off.

(b) A new faculty member shall be advised of the name and location of appropriate steward. Whenever the steward is employed in the same work area as the new faculty member, the immediate supervisor will introduce the faculty member to the steward, who will provide the faculty member with a copy of the Collective Agreement.

(c) Where operational requirements permit, the Employer agrees that a Union steward will be given an opportunity to meet with each new faculty member within regular working hours, without loss of pay, for fifteen (15) minutes, sometime during the first thirty (30) days of employment, for the purpose of acquainting the new faculty member with the benefits and duties of Union membership and the faculty member's responsibilities and obligations to the Union.

(d) A copy of the offer letter to the successful applicant will be forwarded to the BCGEU Staff Representative and the Bargaining Unit Chairperson after the close of each competition.

ARTICLE 6 - EMPLOYER'S RIGHTS

The Union acknowledges that the management and directing of faculty members in the bargaining unit is retained by the Employer, except as this Agreement otherwise specifies.

ARTICLE 7 - EMPLOYER-UNION RELATIONS

7.1 Representation at Meetings with the Employer

No employee or group of employees shall undertake to represent the Union at meetings with the Employer without the proper authorization of the Union. To implement this, the Union shall supply the Employer with the names of its officers and, similarly, the Employer shall supply the Union with a list of its supervisory or other personnel with whom the Union may be required to transact business.

7.2 Union Bargaining Committees

A bargaining committee shall be appointed by the Union and shall consist of up to three (3) people representing the Union. The Union reserves the right to use up to three (3) additional non-instructional persons at any one time for technical information or advice. The Union shall advise the Employer of its appointees to this committee.

7.3 Union Representatives

The Employer agrees that access to its premises will be granted to members of the staff of the Union when dealing or negotiating with the Employer, as well as for the purpose of investigating and assisting in the settlement of a grievance.

Members of Union staff shall notify the designated supervisory official, in advance, of their intention and their purpose for entering and shall not interfere with the operation of the department or section concerned.

In order to facilitate the orderly, as well as the confidential investigation of grievances, the Employer will make available to Union representatives or stewards temporary use of an office or similar facility.

7.4 Joint Administration and Dispute Resolution Committee

(a) Formation and Composition

The Parties to this agreement will participate in the sector's Joint Administration and Dispute Resolution Committee (JADRC) consisting of five (5) representatives of the employers and five (5) representatives of the Provincial Bargaining Council.

(b) Operation

Meetings of JADRC shall be held as needed. A meeting shall be called within twenty (20) days of the written request of either the employers or the Provincial Bargaining Council unless mutually agreed otherwise. A minimum of six (6) representatives with equal representation from the Common Parties will constitute a quorum. JADRC will set its own procedures and protocols. All decisions of JADRC will be mutual decisions between the employers and the Provincial Bargaining Council and will be recorded or confirmed in writing.

(c) Purpose

The purpose of JADRC is to:

(1) Assist in the administration of collective agreements.

(2) Provide a forum for dialogue between the parties respecting issues impacting labour relations.

(3) Provide a means for resolving issues pertaining to the implementation, interpretation and resolution of matters arising from the Common Agreement.

- (4) Appoint an Arbitrator(s) as applicable for Common Agreement Dispute Resolution
- (5) Develop strategies to reduce arbitration and related costs.
- (d) Common Agreement Dispute Resolution

Where a dispute arises concerning the interpretation, application, operation or alleged violation of the Common Agreement, the parties will refer the dispute to JADRC using the Dispute Referral Form at Appendix D to this Agreement. Such referral would occur after the local grievance procedure is exhausted or deemed completed by agreement of the parties.

JADRC will act as the registrar for referred disputes and will forward the matter to an Arbitrator, within thirty (30) calendar days of the receipt of the dispute by JADRC's designated registrar. The agreed to list shall be as set out by JADRAC. The arbitrators shall be chosen in rotation from the list. A list of the arbitrators shall be provided to the Labour Management Committee once a year

Notwithstanding the referral of a dispute to an Arbitrator, the parties may mutually agree to request that JADRC attempt to resolve the matter through a pre-hearing discussion at the JADRC level. Where JADRC reaches a mutual decision on a matter referred, the decision will be final and binding upon the parties.

Prior to an arbitral hearing, and in the absence of any JADRC decision, the parties may resolve a dispute which relates to the interpretation, application, operation or allege violation of the Common Agreement. The resolution is without prejudice or precedent.

(e) Process and Costs

Arbitral decisions shall be rendered within fifteen (15) calendar days of the conclusion of the hearing. Time limits may be altered by mutual agreement between the employers and the Provincial Bargaining Council.

An Arbitrator has the authority to order pre-hearing disclosure and to act as a Mediator provided such action does not unduly delay a decision.

Each party will be responsible for its own costs. The costs of the Arbitrator will be shared by the parties.

7.5 Technical Information

The Employer agrees to provide to the Union such information that is available relating to faculty members in the bargaining unit, as may be required by the Union for collective bargaining purposes.

7.6 Human Resources Database

The Parties believe that their on-going and collective bargaining relationships are enhanced through useful, timely and accessible data on relevant human resources matters, including those listed below.

The Parties agree to provide and support the accumulation and dissemination of available data to the PSEA, which will be responsible for the management of the HRDB project including the gathering, analysis, and maintenance of such data. The Parties may undertake joint projects for the comparative analysis of such data.

The Parties agree that a Steering Committee will oversee this program. The Committee will include representatives designated by each party.

The Parties recommend that the Ministry of Advanced Education continue to provide funding to assist in the gathering, analysis, and maintenance of such data through the agreed-upon organization.

- (a) Relevant Matters include
 - (1) Health and Welfare
 - (i) Benefit plan designs
 - (ii) Participation rates
 - (iii) Premiums
 - (iv) Cost sharing
 - (v) Commission costs
 - (vi) Available studies commissioned by Government agencies
 - (e.g. comparative benefit analysis)
 - (vii) Carrier contracts
 - (2) Collective Bargaining
 - (i) Salary information by classification
 - (ii) Demographics: age, gender, salary, placement, status

- (iii) Analysis of local collective agreements within the system
- (iv) Pension plan participation rates
- (v) FTE, head count, placement on scale, appointment status
- (3) Contract Administration

(i) Arbitration, Labour Relations Board, JADRC, Harassment, Jurisdictional and other third-party decisions and costs thereof for the system

(ii) Local Letters of Understanding

ARTICLE 8 - RESPECTFUL WORKING ENVIRONMENT

8.1 Statement of Commitment

The Employer promotes teaching, scholarship and research and the free and critical discussion of ideas.

The Union and the Employer are committed to providing a working and learning environment that allows for full and free participation of all members of the institutional community. Harassment undermines these objectives and violates the fundamental rights, personal dignity and integrity of individuals or groups of individuals. Harassment is a serious offence that may be cause for disciplinary sanctions including, where appropriate, dismissal or expulsion.

The Employer has a responsibility under the BC's *Human Rights Code* to prevent harassment and to provide procedures to handle complaints, to resolve problems and to remedy situations where harassment occurs.

The Employer will offer educational and training programs designed to prevent harassment and to support the administration of the institutional policies and to ensure that all members of the institutional community are aware of their responsibility with respect to the policy. The Union and the Employer agree that attendance at these educational and training programs is mandatory.

8.2 Discrimination and Harassment (BC Human Rights Code)

(a) Harassment is a form of discrimination that adversely affects the recipient on one (1) or more of the prohibited grounds under the BC *Human Rights Code* [R.S.B.C. 1996 c.210].

Harassment as defined above is behaviour or the effect of behaviour, whether direct or indirect, which meets one (1) of the following conditions:

(1) is abusive or demeaning;

(2) would be viewed by a reasonable person experiencing the behaviour or effect of the behaviour, as an interference with their participation in an institution-related activity;

(3) creates a poisoned environment.

As of this date, the grounds protected against discrimination by BC's *Human Rights Code* [R.S.B.C. 1996 c.210] are age, race, colour, ancestry, place of origin, political belief, religion, marital status, family status, physical or mental disability, sex, sexual orientation and, in the case of employment, unrelated criminal convictions.

(b) Sexual Harassment

The Union and the Employer recognize the right of the faculty members to work and students to learn in an environment free from sexual harassment; therefore, the Employer will undertake to discipline a faculty member engaging in sexual harassment. Sexual Harassment is:

(1) behaviour of a sexual nature by a person who knows or ought reasonably to know that the behaviour is unwanted or unwelcome; and

(2) which interferes with another person's participation in an institution-related activity; or

(3) which leads to or implies employment, educational or academically-related consequences for the person harassed; or

(4) which creates a poisoned environment.

The parties agree that, where mutually agreeable, they shall first attempt to use Vancouver Island University Procedure No. 21.03.001 to resolve complaints prior to accessing the grievance process. A faculty member may initiate a grievance under this Clause. Grievances under this Clause will be handled with all possible confidentiality.

An alleged offender under this Clause shall be entitled:

(1) to be given notice of the substance of a grievance under this Clause;

(2) to be given notice and to attend, participate in and be represented in any arbitration hearing which is held as a result of a grievance under this Clause.

8.3 Workplace Bullying and Personal Harassment

(a) The parties support a working environment that is free from workplace bullying and personal harassment. For the purposes of this Article, the definition for workplace bullying and personal harassment will be as set out in Vancouver Island University Policy No. 21.05 and Procedure No. 21.05.001.

(b) The Union and the Employer agree that attendance in educational and training programs is mandatory for all employees, including the mandatory awareness and prevention training for all persons in a supervisory role, including but not limited to excluded personnel, Program Chairs, Union Shop Stewards and Executives.

(c) The procedures for addressing workplace bullying and personal harassment complaints will be in accordance with Vancouver Island University Policy No. 21.05 and Procedure No. 21.05.001.

The complainant and the respondent have the right to have a Union shop steward present at all times during the investigative process.

8.4 **Procedures for Discrimination and Harassment (BC Human Rights Code)**

The procedures for addressing human rights and sexual harassment complaints will be in accordance with Vancouver Island University Policy No. 21.03 and Procedure No. 21.03.001.

The complainant and the respondent have the right to have a Union shop steward present at all times during the investigative process.

(a) Informal Processes

The parties agree that, where mutually agreeable, they may first attempt to use Vancouver Island University policies or processes to resolve complaints of harassment and sexual harassment prior to accessing the following procedures in Article 8.4(c) Mediation and 8.4(d) Investigation.

(b) Right to Legal Counsel

The Union is the exclusive bargaining agent for the bargaining unit faculty member and as such has the exclusive right to represent the faculty member in all matters pertaining to their terms and conditions of employment, including matters that may lead to discipline by the Employer. An individual bargaining unit faculty member has no right to be represented by legal counsel during an Article 8 investigation involving an allegation of harassment.

(c) Mediation

When a complaint is received by the Employer involving an individual covered by this Collective Agreement, the parties will initiate a mediation procedure at the bargaining unit level. The mediation process is the recommended avenue of resolution. Consensual mediation will require the agreement of the complainant and the respondent to use the following process:

(1) the parties will discuss the nature of the complaint and agree upon who will conduct the mediation;

(2) the mediation process and resolution will be kept strictly confidential by all participants;

(3) where a resolution is reached, the complainant and the respondent must agree in writing to the resolution and the matter will then be considered concluded;

(4) no record of the mediation except the written agreed resolution will be placed on an faculty member's file. The written resolution will be removed from the faculty member's file after twelve (12) months unless there has been a subsequent complaint of harassment against the faculty member within the twelve (12) month period.

(d) Investigation

Where either the complainant or respondent does not agree to mediation, or no resolution is reached during the mediation, the complaint will be referred to an Investigator selected from a list of Investigators agreed upon by the parties. An Investigator will be appointed within ten (10) working days of referral.

Where the parties are unable to agree on a list of Investigators, JADRC will determine the list.

The referral should, where possible, include a written statement from the complainant and the respondent which succinctly outlines the issue(s) in dispute. The referral should be assembled by the Employer and forwarded to the Investigator with a copy sent to the Union(s).

The appointment of an Investigator does not preclude an Investigator from mediating the dispute where possible up to the time of submission of the Investigator's report to the parties pursuant to Article 8.4(e)(1) below.

Any complaint of harassment will be kept confidential except as is necessary to investigate and resolve the issue. Investigators will stress the confidentiality of the investigation with the person(s) interviewed.

- (e) Terms of Reference of the Investigator
 - (1) The purpose of the Investigator will be to ascertain facts.

(2) All persons quoted in the investigation will be named by code determined by the Investigator to preserve confidentiality.

(3) The report of the Investigator will be given, in confidence, to the Union(s) and the Employer. It is the responsibility of the Employer to forward a copy of the report to the complainant and the respondent. The Employer will state, in a covering letter, that the report is confidential. The report should refer to individuals involved by code only. However, a reference key will be provided to the Employer and the Union(s) for internal use. This practice should be repeated at any subsequent arbitral proceeding.

(4) The report will not be introduced as evidence or have standing in any arbitration, or other legal procedure. This does not preclude the parties from reaching an Agreed Statement of Fact based upon facts in the report in preparation for an arbitral proceeding.

(5) Reliance on Report of Third Party Investigator

Despite Article 8.4(c)(4), the Employer is entitled to rely on the fact of mediation or the report of a third party Investigator as evidence that may mitigate liability in a proceeding that follows receipt of the third party Investigator's report.

The Employer is entitled to rely on the Investigator's report as evidence that it acted in good faith in any disciplinary action that it undertook following receipt of the third party Investigator's report where the issue of good faith is raised by a grievor or the Union.

(6) The Investigator will not be compellable as a witness in any arbitration or other legal procedure which may result from the investigation.

(7) The Investigator will conclude their work within twenty (20) days of appointment and will render a report within a further ten (10) days. These timelines may be extended if deemed appropriate by the parties. If a dispute arises with respect to the extension, the matter will be referred to JADRC. If requested by the investigator, the Employer will provide meeting space and contact information about persons to be interviewed.

(8) The Investigator may, as part of their report, make recommendations for resolution of the complaint.

(9) The Investigator's report will not be placed on a faculty member's file.

(f) List of Local Investigators

The following list of local Investigators is attached for the use of the parties at their option under Article 8.4(a):

- Louise Pohl
- Betty Baxter
- Rebecca Frame
- Irene Holden
- Deborah Lovett
- Ana Mohammed
- John Sanderson

8.5 Findings

(a) The Employer will make a written determination based upon the facts and recommendation, if any, within ten (10) working days of the receipt of the Investigator's report. If necessary, this timeline may be extended by mutual agreement between the parties.

(b) The determination will:

- (1) state the action(s), if any, to be taken or required by the Employer;
- (2) include, where appropriate, a statement of exoneration.

8.6 Rights of the Parties

Should a complainant file a complaint under the provisions of the *Human Rights Code*, it is understood that the *Human Rights Code* complaint will be set aside until such time as the procedures under this Article have been completed.

Where an allegation includes both complaints under the *Human Rights Code* and a personal harassment complaint, the parties may agree to have the Investigator investigate all of the complaints, in order to relieve against expense and duality of process.

- (a) The above noted procedure does not restrict:
 - (1) The Employer's right to take disciplinary action;
 - (2) The Union's right to grieve such disciplinary action or to grieve an alleged violation of this Article.

(b) The report of the Investigator may be used in the development of an Agreed Statement of Fact for an arbitral proceeding.

8.7 False Complaints, Breaches of Confidentiality and Retaliatory Action

Frivolous, vexatious or malicious complaints of harassment or breaches of the confidentiality provisions of this clause or retaliation in respect of a complaint may result in discipline.

Should retaliation be alleged following the filing of a complaint, an Investigator may deal with that allegation and make a finding.

8.8 Discussion

The parties will meet as necessary to facilitate the administration and other aspects of the application of this Article including issues arising under Article 8.9 below. The parties may refer any differences over the administration or application of this Article to JADRC for resolution.

8.9 Relation to Other Agreements

Where a complaint under Article 8 involves individuals who are covered by another Collective Agreement, the parties will meet to clarify and agree upon a procedure.

ARTICLE 9 - GRIEVANCES

9.1 Definition

The Employer and the Union recognize that grievances may arise concerning:

(a) differences between the parties respecting the interpretation, application, operation, or any alleged violation of a provision of this Agreement; including a question as to whether or not a matter is subject to arbitration; or,

(b) the dismissal, discipline, or suspension of a faculty member bound by this Agreement.

The procedure for resolving a grievance shall be the grievance procedure in this Article.

9.2 Step 1

In the first step of the grievance procedure, every effort shall be made to settle the dispute with the designated appropriate administrator. The aggrieved faculty member shall have the right to have the steward present at such a discussion. If the dispute is not resolved orally, the aggrieved faculty member may submit a written grievance, through the Union steward, to Step 2 of the grievance procedure.

9.3 Time Limits to Present Initial Grievance

A faculty member who wishes to present a grievance at Step 2 of the grievance procedure, in the manner prescribed in Article 9.4, must do so no later than thirty (30) calendar days after the date:

(a) on which they were notified orally or in writing of the action or circumstances giving rise to the grievance;

(b) on which they first became aware of the action or circumstances giving rise to the grievance.

9.4 Step 2

(a) If the grievance is not resolved at Step 1, subject to the time limits in Article 9.3, the faculty member may present a grievance at this level by:

(1) recording this grievance on the appropriate grievance form, setting out the nature of the grievance and the circumstances from which it arose;

(2) stating the Article or Articles of the Agreement infringed upon or alleged to have been violated, and the remedy or correction required; and,

(3) transmitting this grievance to the Manager of Labour Relations through the Union steward.

(b) The Manager of Labour Relations shall:

(1) forward the grievance to the representative of the Employer authorized to deal with grievances at Step 2;

(2) provide the faculty member with a receipt stating the date on which the grievance was received.

9.5 Time Limit to Reply at Step 2

The representative designated by the Employer to handle grievances at Step 2 shall reply, in writing, to a faculty member's grievance within fourteen (14) calendar days of receiving the grievance at Step 2.

9.6 Step 3

If the grievance is not resolved at Step 2, the President of the Union, or designate, may present a grievance to the Vice-President Administration at Step 3:

(a) within fourteen (14) calendar days after the decision has been conveyed by the representative designated by the Employer to grievances at Step 2;

(b) within fourteen (14) calendar days after the Employer's reply was due.

9.7 Time Limit to Reply at Step 3

The representative designated by the Employer to handle grievances at Step 3 shall reply, in writing, to the grievance within thirty (30) calendar days of receipt of the grievance at Step 3.

9.8 Failure to Act

If the President of the Union, or designate, does not present a grievance to the next higher level within the prescribed time limits, the grievance will be deemed to be abandoned. However, the Union shall not be deemed to have prejudiced its position on any future grievance.

9.9 Time Limit to Submit to Arbitration

Failing satisfactory settlement at Step 3, and pursuant to Article 10, the President or designate, may inform the Employer of the intention to submit the dispute to arbitration within:

- (a) thirty (30) calendar days after the Employer's decision has been received; or
- (b) thirty (30) calendar days after the Employer's decision was due.

9.10 Amending of Time Limits

The time limits fixed in this grievance procedure may be altered by mutual consent of the parties, but the same must be in writing. Where a grievance or a reply is presented by mail, it shall be deemed to be presented on the day on which it is postmarked and it shall be deemed to be received on the day it was delivered to the appropriate office of the Employer or the Union.

9.11 Dismissal or Suspension Grievance

(a) In the case of a dispute arising from an faculty member's dismissal, rejection of probation, suspension greater than 20 days or suspension for just cause pending investigation, the grievance may be filed directly at arbitration, within 30 days of the date on which the dismissal, rejection on probation, or suspension occurred, or within 30 days of the faculty member receiving such notice.

(b) In the case of a dispute arising from other suspension, the grievance may commence at Step 3 of the grievance procedure within thirty (30) calendar days of the date on which the suspension occurred, or within thirty (30) calendar days of the faculty member receiving notice of suspension.

9.12 Deviation from Grievance Procedure

The Employer agrees that, after a grievance has been initiated by the Union, the Employer's representatives will not enter into discussion or negotiation with respect to the grievance, either directly or indirectly, with the aggrieved faculty member without the consent of the Union.

In the event that, after having initiated a grievance through the grievance procedure, a faculty member endeavours to pursue the same grievance through any other channel, then the Union agrees that, pursuant to this Article, the grievance shall be considered to have been abandoned.

9.13 Policy Grievance

Where either party disputes the general application, interpretation, or alleged violation of an Article of this Agreement, the dispute shall be discussed at Step 3 with the Vice-President Administration or the Union as the case may be.

Where no satisfactory agreement is reached, either party may submit the dispute to arbitration, as set out in Article 10 of this Agreement.

9.14 Technical Objections to Grievances

It is the intent of both parties to this Agreement that no grievance shall be defeated merely because of a technical error other than time limitations in processing the grievance through the grievance procedure. To this end, an Arbitration Board shall have the power to allow all necessary amendments to the

grievance, and the power to waive formal procedural irregularities in the processing of a grievance, in order to determine the real matter in dispute and to render a decision according to equitable principles and the justice of the case.

9.15 Retroactive Settlements

Settlements reached at any step of the grievance procedure shall be applied retroactively to the date of the occurrence of the action or situation which gave rise to the grievance or the date set by a Board of Arbitration.

ARTICLE 10 - ARBITRATION

10.1 Notification

Where a difference arising between the parties relating to the interpretation, application or administration of the Agreement, including any questions as to whether a matter is arbitrable or where an allegation is made that a term or condition of this Agreement has been violated, either of the parties may, after exhausting the grievance procedure in Article 9, notify the other party, within thirty (30) calendar days of the receipt of the reply at Step 3, of its desire to submit the difference or allegations to arbitration.

10.2 Appointment of a Single Arbitrator

When a party has requested that a grievance be submitted to arbitration, a single Arbitrator shall be selected from a list agreed to by the parties. The arbitrators on the agreed to list shall be selected through rotation and availability.

10.3 Appointment of a Board of Arbitration

Notwithstanding the provisions contained in Section 10.2, the parties shall, by mutual consent, have the option to appoint a three (3) member Board of Arbitration. The party requesting the grievance be submitted to arbitration shall indicate to the other Party of the Agreement the name of its nominee on an Arbitration Board. Within seven (7) calendar days thereafter, the other party shall indicate the name of its appointee to the Arbitration Board. The two (2) Arbitrators shall then meet to select an impartial Chairperson.

10.4 Failure to Appoint

If the recipient of the notice fails to appoint an Arbitrator; or, the two (2) appointees fail to agree upon a Chairperson within seven (7) calendar days of their appointment, the appointment shall be made in accordance with the *Labour Relations Code*.

10.5 Board Procedure

The Board may determine its own procedure in accordance with the *Labour Code* and shall give full opportunity to all Parties to present evidence and make representations. It shall hear and determine the difference or allegation and shall make every effort to render a decision within thirty (30) calendar days of its first meeting.

10.6 Amending Time Limits

The time limits fixed in the arbitration procedure may be altered by mutual consent of the parties but the same must be in writing.

10.7 Appointment of Arbitrator

The agreed to arbitrators pursuant to Article 10.2, shall be as follows:

- Chris Sullivan
- Colin Taylor
- John Hall
- Mark Brown
- Marguerite Jackson

10.8 Expedited Arbitrations

(a) Where a difference arises between the parties relating to the interpretation, application or administration of the collective agreement, including where an allegation is made that a term or condition of the collective agreement has been violated, either party may, after exhausting the steps of the grievance procedure under the agreement, notify the other party within thirty (30) calendar days of receipt of the last grievance step reply, of its desire to arbitrate and to submit the difference or allegation to expedited arbitration before a single arbitrator.

(b) Issues for Expedited Arbitration

(1) All grievances except those relating to the following shall be resolved by expedited arbitration:

- (i) Dismissals;
- (ii) Suspensions in excess of five (5) working days;
- (iii) Policy grievances;

(iv) Grievances requiring substantial interpretation of a provision of the collective agreement;

- (v) Grievances requiring the presentation of extrinsic evidence;
- (vi) Grievances where a party intends to raise a preliminary objection;
- (vii) Grievances arising from the duty to accommodate; and

(viii) Grievances arising from the interpretation, application and administration of the Common Agreement, including but not limited to, the application of Article 13.1 of the Common Agreement.

(2) Those grievances not suitable for resolution at expedited arbitration, as listed under Section 2(a) above, may be referred to arbitration under the provisions of the collective agreement.

By mutual agreement, a grievance falling into any of the categories as listed under Section
 above, may be placed into the expedited arbitration process.

(c) Expedited Arbitrators

The following arbitrators shall be selected on the basis of the person who is first available to hear the grievance on a rotating basis. It is understood that the same arbitrator will not be selected to hear consecutive grievances except by mutual agreement by the parties.

Chris Sullivan

- Colin Taylor
- John Hall
- Mark Brown
- Marguerite Jackson

If none of the listed arbitrators are available, the parties shall agree to another arbitrator.

(d) Process

As the process is intended to be expedited, lawyers shall not be retained to represent either party. This does not preclude either party from using staff who may be lawyers.

(e) Agreed Statement of Facts

The parties shall develop an agreed statement of facts and shall exchange reliance documents prior to the hearing. Disclosure of relevant or potentially relevant documents is a mutual and ongoing obligation before and during the particular hearing.

(f) Written Submission

By mutual agreement, written submissions may be used in place of a hearing. Submissions shall be in standard format and will not be more than ten (10) pages long.

(g) Procedure

All presentations shall be short and concise and are to include a comprehensive opening statement. The parties shall make limited use of authorities during their presentations.

(h) Mediation

(1) Prior to rendering a decision, the arbitrator may assist the parties in mediating a resolution to the grievance.

(2) Where mediation fails or is not appropriate in the opinion of the arbitrator, a decision shall be rendered as contemplated herein.

(i) Issuance of Decision

The Parties will request that the decision of the arbitrator is completed on the agreed form and mailed to the Parties within three (3) working days of the hearing.

(j) Status of the Decision

(1) All decisions, including mediated settlements, under this expedited arbitration process are limited in application the particular dispute and are without prejudice. The decisions shall have no precedential value and shall not be referred to in any subsequent proceeding. The expedited arbitrators shall include statements to this effect at the beginning of their decision.

(2) All settlements of proposed expedited arbitration cases made prior to an expedited hearing are also without prejudice and have no precedential value.

(3) The decision or award, including mediated settlements, is final, binding, and conclusive, and is not open to question or review in a court on any grounds whatsoever.

(4) Should the parties disagree as to the meaning of the decision or award, including mediated settlements, either party may request that the arbitrator clarify the decision.

(k) Costs

(1) The parties shall equally share in the costs of the fees and expenses of the expedited arbitrator.

(2) Hearings shall be conducted at the institution or at the offices of the union where possible to minimize costs.

(I) Authority of Arbitrator

The expedited arbitrator shall have the same powers and authority as an arbitration board established under the provisions under the *Labour Relations Code*.

ARTICLE 11 - DISMISSAL, SUSPENSION AND DISCIPLINE

11.1 Burden of Proof

In all cases of discipline, the burden of proof of just cause shall rest with the Employer.

11.2 Requirement of Writing

Notice of dismissal shall be in writing and shall set forth the reasons for dismissal. A faculty member who fails to report for duty for ten (10) consecutive working days, without informing the Provost and Vice-President, Academic, Dean, Director or Campus Administrator or in their absence the Associate Vice-President, Human Resources, or designate, of the reason for the absence, will be presumed to have abandoned the position. A faculty member shall be afforded the opportunity to rebut such presumption and demonstrate that there were reasonable grounds for not informing the Employer.

11.3 Dismissal and Suspension Grievance

All dismissals and suspensions will be subject to formal grievance procedure under Article 9 of this Agreement. A copy of the written notice of dismissal or suspension shall be forwarded to the President of the Union within five (5) calendar days of the action being taken.

11.4 Right to Grieve Other Disciplinary Action

Disciplinary action grievable by the faculty member shall include written censures, letters of reprimand, and adverse reports or performance evaluation. A faculty member shall be given a copy of any such document placed on the faculty member's file which might be the basis of disciplinary action. Should a faculty member dispute any such entry in their file, they shall be entitled to recourse through the grievance procedure and the eventual resolution thereof shall become part of their personnel record. Upon the faculty member's request, any such document, other than official evaluation reports, shall be removed from the faculty member's file after the expiration of eighteen (18) months from the date it was issued, provided there has not been a further infraction. The Employer agrees not to introduce as evidence in any arbitration any document from the file of a faculty member, the existence of which the faculty member was not aware at the time of filing.

11.5 Evaluation Reports

Where a formal appraisal of a faculty member's performance is carried out, the faculty member concerned shall be given the opportunity to read and review the appraisal. Provisions shall be made on any evaluation form for a faculty member's acknowledgement of receipt, which shall be signed by the faculty member upon receipt. The faculty member's signature on the evaluation form shall acknowledge only that the document has been received and read.

11.6 Union Representation

It is incumbent on a supervisor initiating an interview with a faculty member which is for the purpose of formal disciplinary action to notify the faculty member in advance of the purpose of the interview so the faculty member may contact the steward and invite Union attendance.

11.7 Access to Personnel Record

During normal working hours, and in the presence of a Human Resources Department staff member, every faculty member has the right of access to their personnel record at a time mutually agreed upon in writing by the Employee and the Employer. The Employer will provide copies of file entries as requested.

ARTICLE 12 - SENIORITY, LAYOFF AND RECALL, SEVERANCE, BUMPING RIGHTS

12.1 Seniority Defined

Definition of Service - Service shall be defined as the total hours paid, exclusive of overtime, converted to an FTE. 1.0 FTE, exclusive of overtime, shall represent one (1) year of service. (See Appendix I.)

(a) Regular Faculty Member Seniority

Seniority shall be defined as the number of accrued hours paid in the bargaining unit. Accrued hours for a regular faculty member shall include all regular hours paid, plus hours accrued per Article 12.1(c)(4) to a maximum of 1.0 FTE per year (equivalent to one [1] year seniority), exclusive of overtime and vacation pay-in-lieu.

(b) Term Faculty Member Seniority

(1) Term faculty members shall accumulate seniority based on the total number of hours paid, exclusive of overtime and vacation pay-in-lieu, by a faculty member in the bargaining unit pursuant to their term appointment.

(2) Term faculty members shall lose all seniority accumulated as a term faculty member in accordance with Articles 12.1(b)(1) and (2) if a period of twenty-four (24) months elapses from the date of the last term appointment to the date at which such a faculty member commences a subsequent term appointment.

(3) A term faculty member's seniority credits shall be applied only for the following purposes:

(i) determining the term faculty member's incremental entitlement in accordance with Article 24.5(d)(2) herein;

(ii) preference for term appointment vacancies in accordance with Article 30.3 herein;

(4) Any former term faculty member of the University hired on a subsequent term appointment more than twenty-four (24) months after the end of their previous term appointment shall accumulate term faculty member seniority in accordance with this Article 12.1(b) as a new faculty member starting at a base of zero hours and shall be placed on the salary scale in accordance with the initial placement provisions of Article 24.5(a).

(5) Seniority as applied to term faculty members in this Collective Agreement shall be limited to the application of seniority credits only for the purposes identified in Article 12.1(b)(4) above.

(c) Additional Hours

(1) Additional hours may be assigned to a regular part-time faculty member in the same instructional program (as listed in Appendix C), merit considered, without posting, on a seniority basis providing the hours do not conflict with the regular assignment or result in overtime.

(2) Regular part-time faculty members wishing to claim additional term or auxiliary hours in the same instructional program area (Appendix C) must apply to the BCGEU job competition posted by the Human Resources Department in September of each year, specifying which campus(es) are of interest. A list of regular part-time faculty members who apply to the posting will be distributed to Deans, Directors, Campus Administrator and the Union Chairperson.

(3) Qualified term faculty members shall be offered auxiliary work by seniority only after regular part time faculty members have been offered such work.

(4) Any regular part-time faculty member who is assigned additional hours shall have those term and auxiliary hours, exclusive of overtime and vacation pay-in-lieu, credited to their regular seniority bank.

(5) Any term faculty member who is assigned additional hours shall have those term and auxiliary hours, exclusive of overtime and vacation pay-in-lieu, credited to their term seniority bank.

(6) The parties agree that there shall be no additional cost to the Employer associated with faculty members exercising their seniority to claim extra available work on other than home campuses.

(d) Auxiliary faculty members shall not accumulate seniority.

12.2 Seniority List

(a) The Employer shall provide the Union with a seniority list for regular and term faculty members on January 31 and September 30 of each year.

(b) Before a change in workload can occur, including layoffs, claiming and posting and hiring of available work, the seniority list will be updated to include paid hours and accrued vacation. The seniority list will be updated to the date that the faculty member is exercising their rights for the above.

12.3 Loss of Seniority

(a) Regular Faculty Members

A faulty member shall lose seniority as a regular faculty member if:

- (1) discharged for just cause;
- (2) voluntary termination of employment or abandonment of position occurs;
- (3) laid off for more than twenty-four (24) months;
- (4) elects severance on layoff, pursuant to Article 12.9;
- (5) appointed to a Dean position as per Article 2.12;
- (6) they accept an appointment to a regular full-time position outside the bargaining unit.

(b) Term Faculty Members

A term faculty member shall lose seniority as a term faculty member:

(1) if a period of twenty-four (24) months elapses from the end date of the last term appointment, subject to Clause (4) below;

- (2) if discharged for just cause;
- (3) if voluntary termination of employment or abandonment occurs;

(4) if a period of twenty-four (24) months elapses from the date of termination of WorkSafe BC wage loss and vocational rehabilitation benefits for a claim incurred while on the job at Vancouver Island University;

(5) in the event that any of the circumstances in Article 12.3(a)(1), (2) and (4) occur.

(c) The acceptance of an auxiliary or term appointment by an faculty member on layoff shall not affect that faculty member's layoff or recall status.

12.4 Same Seniority Credits

When two (2) or more faculty members have the same seniority credits then seniority shall be determined by initial date of hire.

12.5 Pre-Layoff Canvass

(a) Where the Employer identifies to the Union a need to proceed with a layoff of regular faculty members pursuant to Article 12.7, the Employer shall, prior to issuing a layoff notice to any faculty member under Article 12:

(1) canvass the placement of regular faculty member(s) identified pursuant to Article 12.7, for layoff into a vacant position for which the faculty member(s) is qualified (the posting procedures in this Agreement shall be waived for such vacancies); or

(2) canvass any regular faculty member or group of regular faculty members within the Department identified (competency considered) for reduction in order to invite on a voluntary basis resignation with severance benefits of one (1) month for each full year of service up to a maximum of three (3) months.

(b) Where a faculty member selects an option or accepts an offer of placement, which shall be confirmed in writing by the Employer, such acceptance is final and binding on the faculty member.

(c) Responses from faculty members to the pre-layoff canvass will be received by the Employer for consideration only if submitted within fourteen (14) calendar days of issuance of a written notice to the faculty member or group of faculty members within the department identified for reduction and to the Union of the pre-layoff canvass.

(d) Where the number of volunteers exceeds the number of positions to be reduced, the determination shall be on the basis of program need and other relevant factors, one (1) of which is seniority.

12.6 Notice of Layoff

(a) In the event of a layoff of regular faculty members, notice shall be twenty (20) working days as the required written notice.

(b) Reduction, in whole or part, of a BCGEU faculty member's current regular appointment will constitute a layoff. Should it be necessary, layoff notice shall be served to a faculty member on assisted, unassisted or vacation leave subject to the conditions of this section.

(c) When the Senate approves a recommendation to suspend a program through the Summative Assessment process that may result in the layoff of a faculty member(s), the University shall notify the Union of the Senate decision no less than three (3) months prior to the formal notice of layoff.

(d) When written notice is given to the faculty member, the University shall provide the reasons for layoff and inform the faculty member(s) of their rights as outlined in the collective agreement.

12.7 Layoff, Recall, Bumping Rights of Regular Faculty Members

(a) A regular faculty member with three (3) or more years of regular service seniority who is to be laid off shall elect one (1) of the following:

(1) if applicable, accept a reduced assignment;

(2) fill a vacancy where they have the necessary skills, abilities, experience, education, qualifications and certifications;

(3) displace a regular faculty member with less regular service seniority where the laid off faculty member has the necessary skills, abilities, experience, education, qualifications and certifications. Where a regular faculty member elects to displace another regular faculty member with less regular service seniority, they shall notify the Employer within ten (10) calendar days of their receipt of notice of layoff; or where a regular faculty member exercises their right to displace a regular faculty member with less service seniority, the Union and the bargaining unit Chairperson shall be notified in writing by copy of the acknowledgement letter to the regular faculty member. The bargaining unit Chairperson or their designate, if requested by the regular faculty member, can sit as an observer at all interviews dealing with the displacement option exercised by the regular faculty member;

(4) elect severance benefits, if eligible, provided for in Article 12.9. A regular faculty member who elects severance pay and is subsequently re-employed shall not be entitled to re-credit seniority;

- (5) remain available for recall for twenty-four (24) months;
- (6) opt for retraining plan in Article 12.11.

(b) A regular faculty member, subject to layoff, who has less than three (3) years regular service seniority shall elect one (1) of the following:

(1) if applicable, accept a reduced assignment;

(2) fill a vacancy where they have the necessary skills, abilities, experience, education, qualifications and certifications;

(3) to displace a regular faculty member with less regular service seniority, within the same department listing in Appendix C where the laid off faculty member has the necessary skills, abilities, experience, education, qualifications and certifications. Where a regular faculty member elects to displace another regular faculty member with less regular service seniority, they shall notify the Employer within ten (10) calendar days of their receipt of notice of layoff; or where a regular faculty member esercises their right to displace a regular faculty member with less regular service seniority, the Union and the bargaining unit Chairperson shall be notified in writing by

copy of the acknowledgement letter to the regular faculty member. The bargaining unit Chairperson or their designate, if requested by the regular faculty member, can sit as an observer at all interviews dealing with the displacement option exercised by the regular faculty member. The regular faculty member who is displaced, in the event more than one (1) possibility exists, shall be the one (1) with the least regular service seniority; or

(4) remain available for recall for twenty-four (24) months.

(c) The right of recall is null and void if an opportunity is declined or not responded to within seven (7) calendar days after dispatched by registered mail to the last known address of the faculty member unless response was not reasonably possible. Recall rights are extended for a maximum of twenty-four (24) months from the date of layoff. Term appointments shall not affect this maximum. Salary upon recall shall be at the rate prior to layoff.

(d) Employees on recall who work in term or auxiliary positions during their recall period will earn term seniority. If an employee is recalled to a regular position, any term seniority earned during the recall period will transfer to their regular seniority. At the end of the recall period, when an employee has not been recalled, all regular and term seniority shall be converted to term seniority.

(e) Exceptional Efforts to Reduce Layoffs

(1) Where a layoff notice has been issued, temporary suspension of claiming rights will provide the Dean, Director, or Campus Administrator the opportunity to reassign the additional Term workload amongst the regular faculty members as a viable option to layoff.

(2) Claiming rights under Article 12.1(c) and preference for term appointment vacancies under Article 30.3 for additional Term workload in the same instructional program will be temporarily suspended when a layoff notice has been issued within that instructional program (see Appendix C).

(3) Reassignments would be in the normal course of action for regular faculty members within the same headquarters and/or department within that instructional program seniority group; however, where regular faculty members are impacted by a layoff notice from another headquarters and/or department within that instructional program, claiming rights are suspended.

(4) The suspension of claiming rights will be lifted when it is determined that the work cannot be used to mitigate a layoff.

12.8 Registry of Laid Off Faculty Members

(a) Electronic Posting of Available Positions

On behalf of the Parties, the PSEA will maintain a system-wide electronic Registry of job postings and the necessary supporting database.

(1) Employers are encouraged to use the Registry for the posting of all available positions.

(2) Employers will post on the Registry all employment opportunities of half-time ($\frac{1}{2}$) or more and longer than three (3) months in duration that are available to applicants beyond those employed by the institution by completing the PSEA Electronic Posting of Available Positions form (Appendix E1 - Form 1).

(3) Postings will be removed from the Registry and archived to the database one (1) week after the closing by the Employer that entered the posting.

(5) All faculty members covered by this Agreement may access the electronic Registry of job postings for purposes of review.

(6) Unions, Employers and eligible faculty members have the right to access the information on the Registry.

(b) Electronic Registry of Eligible Faculty Members (Registrants)

(1) Faculty members covered by this Agreement are eligible for listing on the Registry if they are faculty members who have received notice of layoff or have been laid off and are either:

(i) regular faculty members with one (1) calendar year of service working at fifty percent (50%) workload or greater, as defined in the applicable local agreement, or

(ii) term faculty members with two (2) calendar years of service working at fifty percent (50%) workload or greater, as defined in the applicable local agreements.

(2) Faculty members who meet the service requirements of Article 12.8(b)(1)(i) above and have not had appointments renewed are eligible for listing on the Registry.

(3) Length of Listing: A faculty member listed on the Registry may continue to be listed until the earlier of:

(i) recall or re-appointment to equivalent employment at the institution from which the person was laid off or was not reappointed;

(ii) obtaining equivalent employment as a result of being listed on the Registry;

(iii) the expiration of the faculty member's recall rights or two (2) years from the date of registration, whichever is later.

(4) Implementation

(i) A faculty member applies for listing through the Human Resources Department by completing the PSEA Registry of Eligible Employees form (Appendix E2 - Form 2).

(ii) The Employer will immediately forward the completed form to the PSEA who will list eligible faculty members on the Registry.

(iii) A registrant is responsible to ensure that the information on the Registry is current and to notify immediately the Employer and the local Union if they are no longer available for employment through the Registry.

(5) Faculty Members Not Eligible

Faculty members are not eligible for listing on the Registry if they have:

- (i) had their employment terminated for just and reasonable cause;
- (ii) accepted early retirement; or
- (iii) voluntarily resigned their employment.
- (c) Applying for Available Positions

(1) It is the responsibility of a faculty member listed on the Registry to enquire about and apply for available work as listed on the Electronic Posting of Available Positions.

(2) A faculty member applying for a posted position in the manner prescribed by the posting institution must tell the institution at the time of application that they are a registrant on the Registry.

- (d) Rights for Registrants
 - (1) Entitlement for Interview

Registrants applying for job postings at institutions who meet the hiring criteria as set by the search committee at the hiring institution will be short-listed and will be interviewed. In the event that more than five (5) qualified registrants apply, the institution shall interview the five (5) most qualified registrants.

The application of this language is subject to the provisions of the Collective Agreement in effect at the receiving institution.

(2) Entitlements for Successful Applicants

(i) Orientation/Training: A registrant who accepts an offer of available work shall be entitled to a reasonable amount of orientation and/or training.

(ii) Benefits: Registrants who are eligible for health and insurance benefits at the hiring institution shall have the waiting period(s) waived subject to carrier provisions.

(iii) Seniority: All registrants who accept an offer of available work will have their seniority recognized at the new institution for all purposes other than severance accrual for subsequent layoffs.

- a. In the case of the hiring from the Registry of an applicant represented by the BCGEU into another bargaining unit represented by the BCGEU, they will have their seniority recognized for all purposes other than severance accrual.
- b. FPSE local Unions may elect to participate in a reciprocal arrangement with other participating FPSE locals and with the BCGEU bargaining units for the purposes of recognition of seniority other than severance accrual. FPSE local Unions that elect to participate in such a reciprocal arrangement must indicate their participation through formal notification to JADRC.
- c. In the case of the hiring of an applicant from the Registry by and from institutions with bargaining units registered with JADRC, the successful applicant shall carry their seniority to that new institution for all purposes other than severance accrual.
- d. Relocation Costs for Registrants: Relocation costs for successful applicants who change residence as a result of the hiring that are supported by proper proof of expenditures within ninety (90) days of commencing employment, will be paid by the hiring institution in accordance with its relocation policies and practices for the position for which the registrant was hired. If funding is available, the costs will be reimbursed to the hiring institution from the Labour Adjustment Fund.

e. Recall and Repayment: A faculty member hired from the Registry who is recalled by an institution and returns to work at that institution will repay relocation costs received from the institution that hired them in accordance with its relocation policies and practices for the position for which the registrant was hired.

12.9 Severance Pay

A regular faculty member, subject to layoff, who has three (3) years or more of regular service seniority with the Employer, shall be entitled to receive severance pay of one (1) month salary for each completed year of seniority to a maximum of three (3) months' salary.

12.10 Reinstatement of a Regular Faculty Member

The following shall apply to regular faculty members:

(a) Should the position occupied by a regular faculty member prior to layoff be reinstated or a position with the identical terms of reference be vacated or created in the same department, within the twenty-four (24) month recall period, the faculty member will be offered the position. Article 30.1 will be waived in this instance.

(b) However, all other vacancies in the Union other than short-term vacancies (less than thirty [30] continuous working days) will be posted except those assignments accessed by regular part-time faculty members pursuant to Article 12.1(c)(1). The Employer shall send notice to laid-off faculty members. It is the responsibility of the faculty member to ensure the Human Resources Department is duly notified of the current mailing address.

- (c) Vacancies referred to in Article 12.10(b) shall be offered in the following order:
 - (1) regular laid-off faculty members within the twenty-four (24) month recall period;
 - (2) regular part-time faculty members;
 - (3) term faculty members.

(d) The consideration of regular and term faculty members as set out in Article 12.10(c) above shall be in accordance with the provisions of Article 30.3 for regular faculty members and Article 30.4 for term faculty members.

12.11 Retraining of Regular Faculty Members

Faculty Members who have been laid off may, under the provisions of this Article, elect to be retrained for specific future employment within the Employer, as outlined in Article 12.7(a)(7) opt for retraining plan.

At the time the retraining leave is approved, a specific position(s) will be identified as the future employment option(s). Upon successful completion of the approved retraining plan, the faculty member will be reinstated into the vacant position identified at the onset or during the period of the retraining leave.

If a retraining plan is agreed to between the Employer, Union and the faculty member involved, the faculty member shall receive an amount equivalent to the severance pay in Article 12.9. This amount shall be allocated in such a manner as to provide a payment plan during the retraining period. Continuation of Health and Insurance benefits shall be subject to:

(1) The faculty member meeting the requirements of current contracts with insurance carriers; and,

(2) The faculty member prepaying by way of post-dated cheques to the Employer, the total cost of such benefits.

The maximum period of leave for retraining shall be two (2) years except where the retraining programs involves a Degree program in which the maximum period of such leave shall be three (3) years.

A faculty member who chooses to be retrained shall accrue seniority for the first two (2) years of the retraining program, and shall, upon reinstatement, be placed with no loss of salary status.

12.12 Continuation of IT Privileges for BCGEU Term Faculty

BCGEU term faculty members shall be provided with continued access to VIU Information Technology services for up to six (6) months following the end of their term appointment to conclude instructional duties, or to prepare for subsequent term appointments.

ARTICLE 13 - REGULARIZATION

13.1 Regularization

(a) For the purpose of this Article, "term workload" means the direct instructional component or non-instructional assignment.

(b) Conversion of Faculty Members from Term to Regular Status

A term faculty member will be eligible for regularization if they have worked a minimum of 0.4348 FTE term workload in each of two (2) consecutive appointment years (see Appendix I). Regularization will be based on:

(1) the average term workload of the two (2) consecutive qualifying years, to a maximum of full-time, will be converted to an FTE value (see Appendix I); and

(2) through an annual review, the department will determine the allocation of annual workload (number of hours per day and months per year) to achieve that FTE.

Note: this could result in a regular appointment of less than twelve (12) months, with an annual scheduled break (lay-off notice not required, no provisions of lay-off apply).

(c) Conversion of Part-Time Term Appointments to Increased Regular Status

Increase to regular appointment will be based on:

(1) additional term workload will be converted to regular, based on the average term workload of the two (2) consecutive qualifying years, to a maximum of full-time, which will be converted to an FTE value (see Appendix I); and,

(2) through an annual review, the department will determine the allocation of annual workload (number of hours per day and months per year) to achieve that FTE.

- (d) Other Conditions
 - (1) Following review, conversions will be carried out no later than October 1 of each year.
 - (2) An appointment year is August 1 to July 31.

(3) In all cases, regularization or conversion is subject to satisfactory evaluation, seniority considerations if relevant, availability of ongoing work, and qualifications for the work available.

(4) Provided that all other conditions are met, the absence of an evaluation having been done shall not be a bar to regularization.

(5) The availability of such qualifying ongoing employment that is not for the purpose of leave replacement, such as under Articles 2.9, 11, 17, 18, 19, 22.6, 24.3, is confirmed no later than October 1 after completion of the two consecutive appointment years.

(e) Multiple Departments

In cases where term workload for regularization or conversion is accrued in more than one (1) department, the following sequence will apply:

(1) The workload will be first accommodated in the department(s) where they originated.

(2) If any portion of the workload cannot be accommodated by the originating department, the balance of that workload will be referred to the other departments, in descending order (order is based on the total FTE workload accrued in each department).

(f) Sequence of Action Regarding Regularization

(1) Conversion of term appointments to increased regular status in order of seniority, provided qualifications are met for the work available.

(2) Conversion of eligible term faculty members, in order of seniority, provided qualifications are met for the work available.

(3) Exercise of "right to claim" by regular faculty members including newly regularized faculty members under (e)(1) and (e)(2) above. (Article 12.1[c]).

(4) Posting of term vacancies to meet any work requirements still unfilled.

Note: The D. Munroe Award dated January 18, 2000, provides that nothing prohibits the Employer's right to regularization of any position as it deems necessary.

ARTICLE 14 - WORKLOAD

14.1 Workload Defined

(a) Instructional Workload

For the purpose of this Article, "instructional workload" means a direct instructional component and an indirect instructional component.

(b) Non-Instructional Workload

For the purpose of this Article, "non-instructional workload" means a Chair or non-teaching assignment, which is recognized as distinct from 14.1(a).

14.2 Contact Hours/Workweek

(a) The normal workweek shall be Monday to Friday.

(b) A full direct instructional workload shall be at a minimum of twenty (20) hours per week up to a maximum of twenty-five (25) hours per week and described as a percentage (%) of the Full Time

Equivalency (FTE) allocation, based on program norms (see Appendix I). Each program will establish an annual direct instructional workload. Health programs will be based on an average workload consisting of clinical and class theory. Any changes or exceptions to the program norms will be reviewed with the Union and require Executive approval. The rate of pay shall be as per the Salary Scale (Appendix A). There are two (2) components of instructional workload (whether face to face, online or blended):

- (1) Direct instructional component, which include the following responsibilities:
 - classroom lectures;
 - shop and laboratory supervision and instruction;
 - classroom learning sessions;
 - seminars/tutorials/discussion groups.
- (2) Indirect instructional component, which include the following responsibilities:
 - student interviews;
 - posted office hours;
 - instructional advising, evaluation, and preparation;
 - administrative responsibilities;
 - curriculum maintenance and program enhancement;
 - professional development activities;
 - other duties directly related to instruction, including department meetings and department initiatives;
 - liaising on student placements and assessments.

(c) Self direction is a key aspect to meet indirect instructional professional responsibilities.

(d) A Chair or non-instructional workload shall be up to a maximum of thirty-five (35) hours per week, and described as a percentage (%) of the Full Time Equivalency (FTE) allocation. Indirect instructional components may also be assigned to this workload. The rate of pay shall be as per the Salary Scale (Appendix A).

(e) Instructional and non-instructional assignments will be determined by the Dean, Director or Campus Administrator in consultation with faculty members. A faculty member may request a review of their workload. If disagreement exists the matter shall be referred to the Labour Management Committee for further consultation. The review process does not limit the grievance process.

(f) The number of hours per week for faculty members carrying both an instructional and noninstructional assignment shall be determined on a proportional basis.

(g) All faculty members are expected to attend department meetings and to participate in department initiatives. Department meetings and department initiatives will be arranged to accommodate the majority of faculty members' teaching schedules, not necessarily on the same day or time of the week.

(h) Subject to departmental needs, an alternative direct instructional schedule for a normal workweek or biweekly pay period to a maximum as set out by the program norms (Appendix I), or seventy (70) hours for a non-instructional assignment, may be instituted by mutual agreement of the Dean, Director, or Campus Administrator and the faculty member.

(i) If the department normally delivers direct instruction on an alternative four-day schedule, the fifth working day is normally scheduled for self-directed indirect instructional components.

(j) New or substantive curriculum development is recognized as beyond the workload stated above. New instructional assignments or substantive work projects beyond the normal department scope will be funded separately.

(k) Current program direct instructional workloads that are under twenty (20) hours per week will be maintained unless otherwise agreed to by the parties.

14.3 Curriculum Development and Professional Development

(a) Commitment to Curriculum Development and Professional Development

(1) As part of the regular faculty schedule, four (4) weeks of curriculum and professional development time shall be allocated to each regular full-time faculty member annually. Regular faculty members with part-time workloads shall have their curriculum and professional development time allocated on a pro rata basis.

(b) Plan Submission

(1) Each regular faculty member shall submit to the respective Dean or designate a detailed curriculum and professional development plan. Each Faculty Council shall determine the appropriate due date for submission of curriculum and professional development plans within the Faculty.

(2) Professional development activities will be directly relevant to the role of the regular faculty member including:

(i) Studies;

(ii) Experience in business, industry, community service, government or other educational institutions;

(iii) Preparing/reviewing course material outside of normal curriculum maintenance as per Article 14.2(j);

(iv) Workshops and conferences (presenting or attending);

(v) Other curriculum development and professional development activities as agreed to between the Dean and the regular faculty member.

(3) Any change to the agreed-to plan must be made by mutual agreement with the Dean and faculty member.

(4) In the absence of a curriculum and professional development plan, the Dean has the ability to assign curriculum and professional development activities.

(5) Subject to operational needs and mutual agreement, a regular non-instructional faculty member and/or the Dean may request an alternate curriculum and professional development schedule.

(c) Approval Process

(1) The Dean will track the usage of annual curriculum and professional development time.

(2) There will be no carry-over or unused curriculum and professional development time allocation from one (1) year to the next.

(3) Final approval of the plan is subject to the criteria set out in Article 14.3(f).

(d) Accountability

(1) It is recognized that knowledge sharing is an essential part of on-going learning. Each regular faculty member will provide a written summation of their curriculum and professional activities for reporting back to the Dean.

(2) Each Faculty will collaboratively determine how they wish to engage in peer sharing of their curriculum and professional development activities with their colleagues.

- (3) Program norms at Appendix I shall be respected.
- (e) Semester System/Year Scheduling and Development

Discussions in the development of the system/year schedule, shall include the Chair, Faculty and Dean in order to determine the appropriate annual instructional delivery schedule to allow for curriculum and professional development time.

(f) Dispute Resolution Mechanisms

Where an agreement cannot be reached, either party may refer to the Labour Management Committee. The criteria for resolution of any dispute shall:

- (1) Maintain the current level of student learning outcomes in programs;
- (2) Maintain the student FTE delivery targets;
- (3) Maintain comparable student completion rates;
- (4) Be consistent with the factors set out in Article 14.3(b)(2).

14.4 Designated Headquarters

(a) Every faculty member covered by this Agreement shall be assigned a designated headquarters. When temporarily assigned another work location, time spent in travel from the faculty member's residence to the new work location in excess of time normally spent in travel from the faculty member's residence to their designated headquarters shall be considered time worked, pursuant to Article 14.1, and if that time worked results in overtime, such overtime shall be paid at the straight time rate. It is understood that time spent in travel shall be reasonable.

(b) The parties recognize and agree that the Health programs require the assignment of dual-campus headquarters in order to remain viable. A faculty member who is assigned to teach theory and/or clinical classes at both the Nanaimo and Cowichan campuses in both the Fall and Spring semesters and commutes the equivalent of no less than two (2) days per week each semester to the other campus shall receive a stipend of five hundred dollars (\$500.00) per semester at the end of each semester to a maximum of one thousand dollars (\$1,000.00) per year. The parties agree there will be no additional cost to the Employer associated with faculty members with this dual-campus headquarters arrangement.

(c) The parties agree there will be no additional cost to the Employer associated with faculty members voluntarily applying for work in a different location.

(d) When faculty members are required to report to a central location in order to be assigned their work location, their workday shall commence from the time they are required to report for assignment.

14.5 Meal Periods

Meal periods shall, by mutual agreement between the Employer and the faculty members, be between thirty (30) and sixty (60) minutes in length.

14.6 Rest Periods

All faculty members shall have two (2) fifteen (15)-minute rest periods in each work period in excess of six (6) hours of direct instruction; one (1) rest period to be granted before and one (1) after the meal period. Faculty members working a shift of three and one-half (3½) hours of direct instruction, but not more than six (6) hours of direct instruction, shall receive one (1) rest period during such a shift. Rest periods shall not begin until one (1) hour after the commencement of work and not later than one (1) hour before either the meal period or the end of the shift. Rest periods shall be taken without loss of pay to the faculty member. Rest periods are not considered part of daily or weekly instructional hours.

14.7 MOU#1 Professional Development and Training

Faculty members covered by this Agreement may be given the opportunity to teach Professional Development and Training courses for which they are qualified. It is understood that such instruction will not qualify for premium rates.

ARTICLE 15 - OVERTIME

15.1 Definitions

(a) "Overtime" means direct instructional component or non-instructional assignment performed by a faculty member in excess, or outside, of the regularly scheduled hours of work.

- (b) "Straight-time rate" means the hourly rate of remuneration.
- (c) "Time and one-half" means one and one-half times $(1\frac{1}{2}x)$ the straight-time rate.
- (d) "Double-time" means twice (2x) the straight-time rate.
- (e) "Double-time and one-half" means two and one-half times (2¹/₂x) the straight-time rate.

15.2 Authorization and Application of Overtime

A faculty member who is required to work overtime shall be entitled to overtime compensation when:

- (a) the overtime worked is authorized in advance by the Employer; and
- (b) the faculty member does not control the duration of the overtime worked.

15.3 Right to Refuse Overtime

All faculty members shall have the right to refuse to work overtime, except when required to do so in emergency situations, without being subject to disciplinary action for so refusing.

15.4 Overtime Entitlement

- (a) A faculty member will be entitled to compensation for authorized overtime after:
 - (1) Five (5) hours of direct instructional work on a regular scheduled workday, or
 - (2) up to eight (8) hours per day if on an alternate workweek, or

(b) A faculty member will be entitled to compensation for authorized overtime after exceeding the direct instructional program norm (see Appendix I) for the week or biweekly period, or thirty-five (35) hours of work per week for a non-instructional assignment (as defined in Article 14.1[b]) or seventy (70) hours in a biweekly pay period. The rate of pay shall be maintained as per Appendix A.

(c) Overtime premiums for faculty members carrying an instructional and non-instructional assignment shall be determined on a proportional basis based on a combined full-time appointment.

(d) It is understood and agreed that the provisions made to this Agreement regarding noninstructional assignments shall become effective for any postings and subsequent appointments which occur after the effective date of the revised Collective Agreement.

15.5 Overtime Compensation

(a) Election of Time Off

If the faculty member elects to take compensatory time off for overtime compensation, they shall be entitled, within sixty (60) days, to schedule such compensatory time off by mutual agreement. This provision does not apply to a faculty member on an industry training or cost recovery assignment, unless agreed to by the Employer.

(b) No Carry Over of Overtime

Any overtime due at the fiscal year-end for that year, or prior to termination, shall be paid within the faculty member's pay.

(c) Minimum Overtime Increment

Overtime shall be calculated in thirty (30)-minute increments. However, the faculty member shall not be entitled to any compensation for periods of overtime of less than five (5) minutes per day.

15.6 Overtime Rates

Overtime worked shall be compensated at the following rates:

- (a) time and one-half (1¹/₂x) for the first four (4) hours of overtime on a regularly scheduled workday;
- (b) double-time (2x) for hours worked in excess of (a); and
- (c) double-time (2x) for all hours worked on a day of rest.

The compensation of overtime in (a) and (b) is to be on a daily basis, and not cumulative.

15.7 Overtime Exemption

The Employer and the Union agree, when the Employer offers part-time faculty members the opportunity to accept additional part-time auxiliary or term hours, merit considered, such assignments are exempt from overtime pay (as outlined in Article 15), provided they do not, in combination, exceed eight (8) direct instructional hours per day (as defined in Article 14.2[b]) and the program norms biweekly (see Appendix I).

The Employer may decide to offer such assignments subject to the following considerations:

(1) addresses the educational imperative of sound pedagogy;

- (2) results in effective delivery of instruction to students; and
- (3) is conducive to departmental effectiveness.

ARTICLE 16 - PAID HOLIDAYS

16.1 Paid Holidays

The following have been designated as paid holidays:

New Year's Day	Labour Day
British Columbia Family Day	National Day for Truth and Reconciliation
Good Friday	Thanksgiving Day
Easter Monday	Remembrance Day
Victoria Day	Christmas Day
Canada Day	Boxing Day
British Columbia Day	

Any other holiday proclaimed as a holiday by the Federal, Provincial, or Municipal Government for the locality in which a faculty member is working shall also be a paid holiday.

16.2 Holidays Falling on Saturday or Sunday

For a faculty member whose workweek is from Monday to Friday and, when any of the above-noted holidays falls on a Saturday and is not proclaimed as being observed on some other day, the following Monday shall be deemed to be the holiday for the purpose of this Agreement; and, when a holiday falls on a Sunday and it is not proclaimed as being observed on some other day, the following Monday (or Tuesday, where the preceding section already applies to Monday), shall be deemed to be the holiday for the purpose of this Agreement to be the holiday for the purpose of this Agreement to be the holiday for the purpose of this Agreement to be the holiday for the purpose of this Agreement unless alternate days are mutually agreeable.

16.3 Holiday Falling on a Day of Rest

When a paid holiday falls on a faculty member's day of rest, the Employer shall make every reasonable effort to give the faculty member a lieu day off with pay on the first regularly scheduled workday following the day of rest so affected. Where this is not possible, the lieu day shall be taken at a mutually agreeable time. A non-instructional day or an indirect instructional day is not a day of rest.

When a paid holiday is moved to another day under the provisions of this Clause:

(a) work performed by a faculty member on the day from which the holiday was moved shall be considered as work performed on a day of rest; and

(b) work performed by a faculty member on the day to which the holiday was moved shall be considered as work performed on a holiday.

16.4 Holiday Falling on a Scheduled Workday

A faculty member who works on a designated holiday which is a scheduled workday shall be compensated at the rate of double time (2x) for hours worked, plus a day off in lieu of the holiday; except for Christmas and New Year's when the compensation shall be at the rate of double time and one-half ($2\frac{1}{2}x$) for hours worked, plus a day off in lieu of the holiday.

16.5 Holiday Coinciding with a Day of Vacation

Where a faculty member is on vacation leave and a day of paid holiday falls within that period, the paid holiday shall not count as a day of vacation.

16.6 Paid Holiday Entitlement for Faculty Members Less than Full-time

Entitlement for faculty members who work less than full-time as defined in this Agreement shall be equal to an average of daily hours worked, including paid sick leave and paid vacation, but exclusive of overtime, in the four (4) week period immediately preceding the week in which the statutory holiday occurs.

ARTICLE 17 - ANNUAL VACATIONS

17.1 Annual Vacation Entitlement

The vacation year shall be defined as the twelve (12) month period commencing January 1 to December 31 each year.

A regular full-time faculty member will have an annual vacation entitlement of forty (40) working days per annum. Faculty members engaged on a part-time basis will be entitled to annual vacation on a pro rata basis.

Regular part-time faculty members shall have vacation paid out for all term and auxiliary hours worked at the rate of sixteen percent (16%) for all straight-time earnings.

This forty (40)-day entitlement shall be prorated in the event a faculty member is on leave pursuant to Articles 18.16, 18.20, and 22.6.

Term and auxiliary faculty members will be entitled to vacation pay at the rate of sixteen percent (16%) for all straight-time earnings.

17.2 Utilization of Vacation Credits

(a) Subject to the provisions of this Article, it is the intent of the parties that no faculty member shall be restricted in the time of year chosen to take vacation entitlement. However, all faculty members shall be allowed to take at least five (5) weeks of their vacation entitlement during the period May 1 to September 30 inclusive, which shall be defined as the prime-time vacation period.

(b) The full vacation entitlement would normally be scheduled to fit in with class schedules (low enrollment periods).

(c) A teaching schedule will be developed in September in order that faculty members may identify periods wherein they may schedule annual vacation. Where vacation requests are for periods other than low enrollment, they will be considered on an individual basis, taking into account the reasons for the request, budget limitations and course cost implications.

(d) A maximum of five (5) weeks of vacation in an unbroken period shall be granted on the basis of seniority within departments as first choice vacation period. Second periods of vacations shall be granted on seniority basis in a department but only after all "first" choices have been posted. The Employer shall make every reasonable effort to allow an eight (8)-week unbroken period if the faculty member desires.

(e) Vacation schedules will be circulated and posted by April 1 of each year.

(f) A faculty member who does not exercise seniority rights within two (2) weeks of receiving the vacation schedule shall not be entitled to exercise those rights in respect to any vacation time previously selected by a faculty member with less seniority.

(g) Should programs not be offered in the following calendar year during the period July 1 to August 30, Union members affected will be advised in writing no later than June 1 of the preceding calendar year of this intention.

17.3 Vacation Carry Over and Vacation Payout

A faculty member may carry over a maximum of fifteen (15) days' vacation leave to the next year. A faculty member may not carry vacation leave over for two (2) successive years.

The vacation entitlement for auxiliary and term faculty members, sixteen percent (16%) of regular straight-time earnings, shall be paid on each biweekly pay cheque.

17.4 Scheduled Vacations

(a) Vacation schedules, once approved by the Employer, shall not be changed, other than in cases of emergency, except by mutual agreement between faculty member and Employer.

(b) A faculty member recalled to duty while on vacation shall be reimbursed expenses incurred consistent with Article 32 (Travel Allowance). Travel time to return to duty and where necessary to return to the place from where recalled will not be considered vacation time.

17.5 Illness or Injury During Vacation

A faculty member who becomes ill or injured during the vacation period shall qualify for sick leave and be credited the appropriate number of hours to vacation leave credits. The faculty member must provide medical evidence where sick leave is claimed during vacation utilization. Employer will assume, where necessary, additional cost of obtaining the medical certificate.

17.6 Vacation Credits Upon Death

Earned but unused vacation entitlement shall be made payable, upon termination due to death, to the faculty member's estate.

ARTICLE 18 - SPECIAL AND OTHER LEAVE

18.1 Definitions

All leave entitlements are based on a calendar year (January 1 to December 31). All references to spouse within the leave provisions of this Agreement include heterosexual, common-law and same sex partners. References to family include spouse, children, children's spouses, step child, step child in law, siblings, in law siblings, parents, step-parents, parents-in-law, grandparents, grandchildren, children of siblings, Indigenous Elder, and any person living in the same household who is dependent upon the faculty member. For the purpose of Article 18.8 – Compassionate Care Leave – only, the definition of "family member" is as set out in Appendix H.

18.2 General Leave

The Employer may grant a leave of absence with or without pay to a faculty member for any reason for up to twenty-four (24) consecutive months. Such leaves shall not be unreasonably denied. Where an application for general leave is denied, the applicant will be provided with a written explanation for the denial of the leave.

Seniority Accrual

All paid leaves shall be treated as continuous employment for the purposes of seniority accrual. Unpaid leaves shall be treated as continuous employment for the purposes of seniority accrual for the duration of the leave, except for movement up the salary increment scale.

18.4 Retention of Status

18.3

A faculty member on approved paid or unpaid leave will retain their employment status for the duration of the leave.

18.5 Benefits While on Leave

A faculty member will continue to receive their salary and benefits while on paid leave under this Article. A faculty member on unpaid leave may arrange to pay the costs required to maintain benefit coverage.

18.6 Bereavement Leave

A faculty member not on leave of absence without pay or layoff will be entitled to five (5) days, equivalent to a faculty member's normal workweek as defined in Article 14.2, leave with no loss of pay and benefits in the case of the death of a family member and upon notification to the University. The University may grant additional leave with pay.

If a faculty member is on vacation leave at the time of bereavement, the faculty member shall be granted bereavement leave and be credited the appropriate number of days to vacation leave credits.

18.7 Family Illness Leave

A faculty member will be granted leave of absence for up to five (5) days, equivalent to a faculty member's normal workweek as defined in Article 14.2, per year without loss of pay or benefits for family illness for the purpose of actively caring for the member who is ill and requires the care of the faculty member. Additional family leave may be granted by the University.

18.8 Compassionate Care Leave

(a) Entitlement

A faculty member will be granted a compassionate care leave of absence without pay for up twentyseven (27) weeks to care for a gravely ill family member. For the purpose of this Article 18.8, "family member" is defined as one of the persons listed in Appendix H – Family Members for the purpose of Article 18.8 Compassionate Care Leave. In order to be eligible for this leave, the faculty member must provide a medical certificate as proof that the ill family member needs care or support and is at risk of dying within twenty-six (26) weeks.

A faculty member who is granted a compassionate care leave of absence to care for a gravely ill family member shall be entitled to the benefits as follows:

(1) The faculty member's benefit coverage will continue for the duration of the compassionate care leave, to a maximum of twenty-seven (27) weeks, and the premium payment shall be on the same basis as if the faculty member were not on leave.

(2) Where a faculty member elects to buy back pensionable service for part or all of the duration of the compassionate care leave, to a maximum of twenty-seven (27) weeks, the Employer will pay the Employer portion of the pension contributions in accordance with the Pension Plan regulations.

(3) Compassionate care leave, up to a maximum of twenty-seven (27) weeks, shall be treated as continuous employment for the purposes of seniority accrual under this Agreement.

(4) A faculty member who returns to work following a leave granted under this provision shall be placed in the position the faculty member held prior to the leave or in a comparable position.

(b) Additional Leaves

Should a faculty member require additional time to care for a gravely ill family member, additional leaves may be granted beyond the twenty-seven (27) week period specified in Article 18.8(a) above. Such additional leave shall be pursuant to Article 18.2 General Leave.

(c) Additional Bereavement Leave

Notwithstanding Article 18.1, a faculty member who has taken Compassionate Care Leave pursuant to this article, will also be eligible for any subsequent bereavement leave under Article 18.6.

18.9 Special Leave

Any term or regular faculty member not on leave of absence without pay or layoff shall be entitled to special leave at the regular rate of pay for the following:

- (a) marriage of faculty member three (3) days;
- (b) attend wedding of the faculty member's child one (1) day;
- (c) birth or adoption of the faculty member's child one (1)day;
- (d) serious household or domestic emergency one (1) day;
- (e) moving household furniture and effects one (1) day;
- (f) attend a formal hearing to become a Canadian citizen one (1) day;
- (g) attend funeral as pallbearer or mourner one-half (½) day.

Two (2) weeks' notice is required for leave under Subsection (a), (b), (e), and (f).

18.10 Full-Time Union or Public Duties

The Employer will grant a leave of absence without pay to a faculty member:

(a) to engage in election campaign activities in a municipal, provincial or federal election to a maximum of ninety (90) days. Such leaves will not be unreasonably denied;

- (b) to seek election in a municipal, provincial, or federal election to a maximum of ninety (90) days;
- (c) for a period of one (1) year for faculty members selected for a full-time position with the Union or any body to which the Union is affiliated;
- (d) where elected to public office, for up to two (2) consecutive terms.

18.11 Jury Duty and Court Appearances

(a) Leave of absence without loss of pay and benefits will be provided to a faculty member summoned to serve on a jury or when subpoenaed or summoned as a witness in a criminal or civil proceeding not occasioned by the faculty member's private affairs, or when the faculty member accompanies a dependent child when the child is subpoenaed or summoned to appear as a witness in a criminal or civil proceeding. A faculty member in receipt of pay or benefits under this Article has the

responsibility to reimburse the Employer all monies paid to them by the Court, except travelling and meal allowances not reimbursed by the Employer.

- (b) Time spent at court by a faculty member in an official capacity shall be at the regular rate of pay.
- (c) Court actions arising from employment requiring attendance at court shall be with pay.
- (d) In the event an accused faculty member is jailed pending a court appearance, such leave of absence shall be without pay.

18.12 Leave for Writing Examinations

Leave of absence with pay shall be granted to allow faculty members time to write examinations for courses approved by the Employer.

18.13 Leave for Taking Courses

(a) A faculty member shall be granted leave with pay to take courses at the request of the Employer. The Employer shall bear the full cost of the course, including tuition fees, entrance or registration fees, and other expenses consistent with Articles 18.19 and 32. Fees are to be paid by the Employer when due.

(b) A faculty member may be granted leave without pay, or leave with partial pay, to take courses in which the faculty member wishes to enroll.

18.14 Elections

A faculty member eligible to vote in a federal, provincial, or municipal election or a referendum shall have the required time off as declared by the appropriate government, during the hours in which the polls are open, in which to cast a ballot.

18.15 Exchange Leave

A faculty member holding a regular or continuous appointment may exchange their position and responsibilities with a qualified person from another institution for a fixed period of time with the agreement of the faculty member's department and the Employer.

The faculty member will continue to receive regular salary and benefits for the duration of the exchange. The exchanging individual will be paid by their institutional Employer.

Where there are large inequities in cost of living between the location of the exchanging individual, the Employer and the faculty member may discuss whether further assistance is required to facilitate the exchange.

18.16 Enrollment in VIU Courses

(a) Waiver of Tuition Fees for University Faculty Members Enrolling in Credit Courses

Faculty members who have a regular appointment may register in University courses, at no cost, provided that no fee-paying student is displaced. Such registration will be subject to admissions requirements and enrolment procedures, which will be set from time to time by the Registrar.

(b) Waiver of Tuition Fees for University Faculty Members Enrolling in Professional Development and Training.

A faculty member who is employed for a minimum of three (3) months will be allowed to enrol only during the period of employment in one (1) non-credit or part-time vocational course per term at no cost, subject to procedures which will be set from time to time by the Employer.

18.17 Leave of Absence for University Committees

A faculty member whose assigned work schedule would prevent them from attending meetings of a university committee to which they have been elected or appointed, will be granted a leave of absence from their regular duties without loss of pay or other entitlements to attend such meeting(s).

Where such leave is granted, the Employer will replace the faculty member as necessary. Costs arising from this provision will not be charged against the program area of the participating faculty member.

18.18 Provincial Articulation

In addition to the above, the Employer will cover expenses for one (1) faculty member for each provincially approved articulation meeting held (one [1] meeting per year).

18.19 On-the-Job Training

(a) Where deemed necessary by the Employer, all newly employed faculty members shall receive teaching instruction and orientation before being required to assume a normal instructional load.

(b) Courses leading to the Instructor's Diploma shall normally commence during the first year of employment.

Where the Instructor's Diploma is required by the Employer, the Employer agrees to reimburse to the faculty member-student one hundred percent (100%) of the Instructor's Diploma tuition fees; and if a faculty member-student taking a major course toward the Diploma outside their headquarters area, the following allowances:

- (1) equivalent return economy air fare from the faculty members headquarters;
- (2) equivalent dormitory fees for the duration of the course.

Term faculty members who attend the Instructor's Diploma course while not on an active term appointment and who are subsequently reappointed may be reimbursed as above upon application to and approval by the Dean, Director or Campus Administrator.

(c) Faculty members shall be on leave of absence with pay while attending such courses.

18.20 Unassisted Leave/Reduced Workload

(a) Unassisted Leave for Full-time or Part-time

Faculty members may apply to their Dean, Director or Campus Administrator (with a copy to the Human Resources Department) for unassisted leave for a period of up to one (1) year. Such applications may be for any purpose deemed justifiable by the faculty member. Applications should provide specifics and details of the duration and purpose of the leave. The parties agree that the applications for leave shall not be unreasonably withheld. If it involves industrial or business experience of particular value to the Employer, the faculty member shall receive an increment and accrual of seniority. A faculty member shall return to a faculty position with no loss of salary status and seniority. The faculty member must respond in writing to the Human Resources Department's request of their intent to return to work ninety (90) days prior to the anticipated return date or mid-term for leaves shorter than six (6) months. The Human Resources Department will send notification to the last known address provided by the

faculty member outlining the requirement to confirm the return-to-work date. Such leave may be extended up to an additional year.

Approved applications entailing industry or business experience of particular value to the Employer will be forwarded to the Board by the President for information only.

(b) Reduced Workload

Faculty members may apply for reduced workload on the same basis as they would apply for unassisted leave.

18.21 Foreign Contract Leave

(a) "Foreign Contract Leave" is defined as a reassignment to another employer who has contracted with Vancouver Island University for the delivery of instructional services outside Canada for a predetermined specified period of time.

(b) It is intended that Vancouver Island University, as the Employer of reference, retain responsibility for the payment of salary, related payroll procedures, seniority matters and incremental progression, and where possible, health and insurance coverage dependent upon the provisions of the applicable plans and the conditions of the carrier.

(c) The salary payable to the faculty member on a foreign contract shall be the salary commitment specified in the foreign contract less the cost of any applicable health and insurance coverage. Such salary shall be inclusive of vacation entitlement which shall be considered to be sixteen percent (16%) of the foreign contract salary amount.

(d) The faculty member on a foreign contract leave may elect to utilize accrued vacation credits to top up salary during the foreign contract leave period.

(e) For the purposes of seniority of faculty members whose seniority is based on hours completed, the hours completed on a particular foreign contract shall be calculated by dividing the foreign contract salary payable to the faculty member by the faculty member's hourly rate of pay which the faculty member held immediately prior to the commencement of the foreign contract leave. If the faculty member's overseas work is less than their normal full-time work but is supplemented by external work deemed to be of value to the University, then such external work shall account as credit toward seniority. In any event, the seniority of a faculty member accumulated pursuant to this article shall not exceed the seniority the faculty member would have gained as a full-time regular faculty member working at the University.

(f) It is recognized that a foreign contract would not include provision for overtime or other premium pay.

(g) (1) The faculty member on a foreign contract leave shall be subject to the working conditions of the contract employer.

(2) It is understood that various provisions of the Vancouver Island University/BCGEU Collective Agreement will not be applicable to the faculty member on a foreign contract leave in view of Article 18.21(g)(1) above as well as due to the University being unable to enforce aspects of the employment relationship, such as, but not limited to, occupational health and safety, class size, etc.

(3) It is recognized that a great deal of flexibility is required concerning travel expenses, hours/days of work, class and shift schedules, meal periods, statutory holidays, non-instructional duty days, professional development, processing of disciplinary action, evaluation methods and

the grievance procedure, as well as other matters dependent upon the nature of the foreign contract assignment.

(h) The faculty member will be reinstated to the faculty member's former position at the predetermined time upon the faculty member's return to the University unless the faculty member was a term faculty member.

(i) These foreign contract assignments are normally arranged at the faculty member's initiative. However, if not at the faculty member's initiative the assignment will be upon mutual agreement only between the faculty member and the University.

(j) It is recognized that the eligibility for Workers' Compensation coverage and claims remains in effect for a maximum of six (6) months when on assignment outside of Canada subject always to any changes in the applicable legislation that may impact the extent of the Workers' Compensation coverage that can be maintained.

18.22 Deferred Salary Leave

Deferred salary leave will be in accordance with Vancouver Island University Policy 43.18, and consistent with applicable legislation.

18.23 Leave for Domestic and Sexual Violence

Where leave from work is required due to an employee and/or employee's dependent child experiencing domestic violence or sexual violence, the employee shall be granted leave, in each calendar year, as follows and the leave will remain at least in accordance with *Employment Standards Act:*

- (a) Up to ten (10) days of leave, of which five (5) days are paid, and
- (b) Up to fifteen (15) weeks of unpaid leave.

Leave under (a) and (b) above may be taken in one continuous period or intermittently.

18.24 Cultural Leave for Indigenous Employees

(a) A self-identified Indigenous employee may request up to two days' leave with pay per calendar year to organize and/or attend Indigenous cultural event(s). Such leave will not be unreasonably withheld.

(b) Employees will provide the Employer with the dates of the days for which leave will be requested. Wherever possible, a minimum of two weeks' notice is required for leave under this provision.

ARTICLE 19 - MATERNITY AND PARENTAL LEAVE

19.1 Maternity Leave

A pregnant faculty member will qualify for maternity leave.

(a) Upon request, the faculty member will be granted a leave of absence, without pay, for a period of up to seventeen (17) consecutive weeks. In the case of a term faculty member, leave of absence without pay will be granted for a period of up to seventeen (17) consecutive weeks or for the duration of the term appointment if less than seventeen (17) consecutive weeks remains of the faculty member's term appointment from the date of the commencement of the above leave of absence.

(b) The period of maternity leave without pay may commence thirteen (13) weeks before the expected date of birth, and no later than the actual birth date. The Employer may require the faculty

member to commence the leave where the duties of the faculty member cannot reasonably be performed because of the pregnancy, and to continue the leave until the faculty member provides a certificate from a medical practitioner stating they are able to perform their duties.

(c) Regardless of the commencement date of the leave of absence, the leave shall not end before the expiration of six (6) consecutive weeks following the actual date of birth of the child, unless the faculty member requests a shorter period.

(d) A request for a shorter period must be submitted in writing to the Employer at least one (1) week before the date the faculty member indicates they intends to return to work, and the faculty member must furnish the Employer with a certificate from a medical practitioner stating that they are able to resume work.

(e) On return from maternity leave, a faculty member shall be placed in their former position or in a position of equal rank and salary.

19.2 Parental Leave

(a) Definitions

(1) A "parent" is either the: biological mother or father; adoptive mother or father; individual(s) with legal custody; or individual(s) with legal guardianship. An employee is entitled to apply for parental leave under 19.2, providing the employee with either the:

- Parent or an expected newborn child; or
- Adopting parent of a child placed or about to be placed with the parent for the first time.

(2) "Base Salary" is the salary that a faculty member would earn if working their full workload up to a maximum of a full workload (see Appendix I).

(b) Entitlement

Upon written request, a faculty member shall be entitled to a leave of absence without pay of up to six (6) consecutive months in addition to statutory requirements. Notwithstanding the foregoing, employees who elect the Standard EI Benefit as defined under the *Employment Insurance Act*, shall be entitled to up to twelve (12) consecutive months of additional leave without pay, provided these employees do not take a leave under *Employment Standards Act* that exceeds the Standard EI Benefit period.

(c) Commencement of Leave

Leave taken under Article 19.2(b) shall commence:

(1) for the parent who takes leave under 19.1 they shall be entitled to up to sixty-one (61) consecutive weeks of unpaid leave which must begin, immediately after the end of the leave taken under the maternity leave provisions or on a mutually agreed upon date within seventy-eight (78) weeks of the birth.

(2) for a parent who has not taken leave under 19.1 they shall be entitled to up to sixty-two
 (62) consecutive weeks of unpaid leave which must begin within seventy-eight (78) weeks of the birth.

(3) for an adopting parent, they shall be entitled to up to sixty-two (62) consecutive weeks of unpaid leave within seventy-eight (78) weeks after the child is placed with the parent.

(d) Graduated Return to Work

Upon written request, a faculty member on parental leave under Article 19 may return to work on a graduated basis. Upon receipt of a request, the local parties will mutually agree to an acceptable graduated parental leave return to work plan for the faculty member.

19.3 Benefits Continuation

(a) The Employer will maintain coverage for medical, extended health, dental, group life and disability benefits for leaves taken under Article 19. For the period of the leave, premium and pension contribution payment will be as follows:

(1) Premium payment for benefit coverage shall be on the same basis as if the faculty member were not on leave.

(2) Contributions for pensionable service shall be on the same basis as if the faculty member were not on leave. Where a faculty member elects to buy back pensionable service for part or all of the Article 19 leave, the Employer will pay the Employer portion of the pension contributions in accordance with the Pension Plan regulations.

(b) A faculty member who returns to work following a parental leave shall retain the seniority the faculty member had attained prior to the leave and shall accrue seniority for the period of leave.

A faculty member who returns to work following a parental leave shall be placed in the same position that faculty member held prior to the leave or in a comparable position.

A faculty member who has taken leave under this provision is entitled to all increases in wages and benefits the faculty member would have been entitled to had the leave not been taken.

Where the proposed commencement of the leave or return to work does not coincide with the instructional calendar, the local parties will negotiate mutually acceptable dates.

(c) If a faculty member maintains coverage for medical, extended health, dental and group life, the Employer agrees to pay the faculty member's share of these premiums. If a faculty member fails to return to work on the prearranged date, the Employer will recover monies paid under this section.

19.4 Supplemental Employment Benefit for Maternity and Parental Leave

(a) When on maternity or parental leave, a faculty member will receive a supplemental payment added to Employment Insurance benefits as follows:

(1) For the first week of the leave, a faculty member shall receive one hundred percent (100%) of their salary calculated on their average base salary.

(2) For a maximum of fifteen (15) additional weeks of maternity leave, the faculty member shall receive an amount equal to the difference between the Maternity Employment Insurance benefits and ninety-five percent (95%) of their salary calculated on their average base salary.

(3) For up to a maximum of thirty-five (35) weeks, the parent shall receive an amount equal to the difference between the Standard Parental Employment Insurance benefits and eighty-five percent (85%) of the faculty member's salary calculated on their average base salary.

(4) If the parent elects the Extended Parental El Benefit, for a maximum of sixty-one (61) weeks the parent shall receive the same total SEB benefit amount received under Article 19.4(3) when the employee opts for thirty-five (35) week Standard Parental El benefit, spread out and paid over the sixty-one (61) week period. Payroll will make this calculation.

(5) Provided the employee received SEB as per Article 19.4 (1), (2), (3) and/or (4), for the last week of the parental leave, where no EI benefit is paid, the employee shall receive one hundred percent (100%) of their salary calculated on their average base salary.

(6) The average base salary for the purpose of Article 19.4(a)(1) through (5) is the faculty member's average base salary for the twenty-six (26) weeks preceding the maternity or parental leave. If the faculty member has been on unpaid leave for part of the preceding twenty-six (26) weeks, then up to four (4) weeks of that unpaid leave will be subtracted from the twenty-six (26) weeks for the purpose of calculating the average base salary.

(b) A faculty member is not entitled to receive Supplemental Employment Benefits and Disability benefits concurrently. To receive Supplemental Employment Benefits, the faculty member shall provide the Employer with proof of application for and receipt of Employment Insurance benefits.

19.5 Seniority Rights on Re-employment

(a) The faculty member will confirm in writing to Human Resources Department their intention to return to University employ no later than thirty (30) days prior to the anticipated return to work date.

(b) A faculty member who makes application for re-employment prior to the expiration of maternity, parental or adoption leave and who returns to work following the leave, shall retain service credits and seniority rights accumulated prior to the leave and shall accrue seniority for the period of the leave.

(c) A faculty member who returns to work following a parental leave shall be placed in the position the faculty member held prior to the leave or in a comparable position.

(d) A faculty member who has taken leave under this provision is entitled to all increases in wages and benefits the faculty member would have been entitled to had the leave not been taken.

(e) Where the proposed commencement of the leave or return to work does not coincide with the instructional calendar, the local parties will negotiate mutually acceptable dates.

(f) The faculty member shall be deemed to have resigned on the date upon which leave of absence without pay commenced if an application for re-employment is not made prior to the expiration of the leave.

19.6 Extension Maternity Leave

Maternity leave shall be extended for up to an additional six (6) months for health reasons where a doctor's certificate is presented.

19.7 Sick Leave Credits

Illness arising due to pregnancy, during employment and prior to leave of absence, may be charged to normal sick leave credits.

ARTICLE 20 - HEALTH AND SAFETY

20.1 Conditions

The Union and the Employer agree that regulations made pursuant to any statute of the Province of British Columbia pertaining to the working environment shall be fully complied with.

20.2 Health and Safety Committee

The Employer and the Union agree to establish and operate Joint Health and Safety Committee, as required by WorkSafeBC. A copy of all minutes of the Committees shall be sent to the Union and the Employer.

20.3 Unsafe Work Conditions

(a) Consistent with specific requirements of WorkSafeBC, no faculty member shall be disciplined for refusal to work on a job that is deemed to be unsafe by:

- (1) a member of the bargaining unit who is a member of a Joint Health and Safety Committee; or
- (2) a person designated by a Joint Health and Safety Committee; or,
- (3) a Health and Safety Services Advisor.

(b) Where a condition has been deemed unsafe, in accordance with the above, faculty members will not be assigned to that site until conditions have been corrected.

20.4 Injury Pay Provision

A faculty member who is injured on the job during working hours and is required to leave for treatment, or is sent home for such injury, shall receive payment for the remainder of the shift without deduction for sick leave.

20.5 Transportation of Accident Victims

Transportation to the nearest physician or hospital for faculty members requiring medical care as a result of an on-the-job accident shall be at the expense of the Employer.

20.6 Pollution Control

The Employer and the Union agree to limit all forms of environmental pollution over which they have control.

20.7 Investigation of Accidents

The Joint Health and Safety Committee, as provided in Article 20.2, shall be notified in the event of any major accident or injury and may assist in the investigation as required by WorkSafeBC.

20.8 Pay Provisions

Any faculty member who serves on Joint Health and Safety Committee shall receive the regular rate of pay for attending meetings of the Committee held during working hours or for the assigned investigation of safety matters when assigned by the Committee.

ARTICLE 21 - CONTRACTING OUT

The Employer agrees not to contract out any work presently performed by faculty members covered by this Agreement which would result in the laying off of such faculty members.

ARTICLE 22 - HEALTH AND INSURANCE BENEFITS

There will be no change to the level of Health and Insurance benefits without prior consultation between the parties.

22.1 Eligibility

(a) Regular faculty members shall be eligible to participate in the Health and Insurance benefits as follows:

(1) A workload of at least 0.5 FTE per week or per biweekly period shall be eligible for all the health benefits described in Articles 22.2, 22.3, and 22.4.

(2) A minimum workload of 0.7143 FTE per week or per biweekly period is required for participation in the Insurance benefits provided for in Article 22.5.

(3) All regular faculty members are eligible for Disability benefits in Article 22.6.

(4) Further, a regular faculty member with a workload of 0.5 FTE per week or per biweekly period with additional term hours resulting in a total assignment of 0.7143 FTE per week or per biweekly period for a minimum of ten (10) months, will qualify for the Insurance benefits provided for in 22.5.

(b) Term faculty members shall be eligible to participate in the Health and Insurance benefits as follows:

(1) Appointments of at least five (5) months and a workload of at least 0.5 FTE per week or per biweekly period shall be eligible for the Health benefits described in Articles 22.2, 22.3, and 22.4.

(2) Appointments of at least ten (10)-months and a workload of at least 0.7143 FTE per week or per biweekly period shall be eligible for insurance benefits described in Article 22.5.

(3) A minimum workload of 0.5 FTE for sixteen (16) weeks or more shall be eligible for Disability benefits described in Article 22.6.

(4) The entitlement to the above benefits expires at the end of the term faculty member's period of appointment.

(5) If it is expected that a term faculty member, whose term appointment is ending, will be appointed to a new term appointment within four (4) months of the end of the immediately prior term appointment, the term faculty member may elect to continue to be covered for any or all of the benefits outlined in Articles 22.2, 22.3, and 22.4 provided the term faculty member was already being so covered and provided one hundred percent (100%) of each of the premiums is prepaid by the faculty member to the Employer prior to the end of the current term appointment. This clause also applies to regular part-year faculty members.

(6) If a new term appointment commences within four (4) months of the end of the previous term appointment, the faculty member's benefits that they carried in the term appointment immediately prior will be reinstated at the beginning of the month coincident with or immediately following the new term appointment, provided the term faculty member is eligible for such coverage during the new appointment period.

(c) Auxiliary faculty members are not entitled to the benefits outlined in Article 22.

22.2 Medical Services Plan

The parties recognize that the method of funding MSP has been changed from an individually paid premium system to a system funded by an employer paid payroll tax.

If the government, at any time in the future, reverts to an individually paid premium system for basic medical insurance, the parties agree that the Employer will pay 100% of the premium for employees on the same basis as exists in the 2019 to 2022 Collective Agreement.

22.3 Extended Health Benefits

The Employer pays, for eligible faculty members in accordance with Article 22.1, one hundred percent (100%) of the premium cost of the Extended Health Benefit plan.

(a) Extended Health Benefits coverage shall provide for ninety-five percent (95%) reimbursement for all covered expenses in excess of a fifty-dollar (\$50) deductible in a calendar year. Covered expenses include, but are not limited to, eligible prescription drugs, ambulance charges and emergency medical expenses while travelling outside Canada. The maximum lifetime benefit is unlimited.

(b) (1) Vision care shall provide one hundred percent (100%) reimbursement up to a maximum of six hundred and fifty dollars (\$650) in a twenty-four (24) month period.

(2) Eye vision exams shall be reimbursed to a maximum of one hundred and twenty-five dollars (\$125) every two (2) years.

(c) Hearing Aid benefit claims will be to a maximum of fifteen hundred dollars (\$1,500) every four (4) years.

(d) Emergency Travel Assistance coverage shall provide twenty-four (24)-hour assistance to the faculty member and/or dependent(s) for locating medical care and arranging medical transportation during a medical emergency occurring almost anywhere in the world.

(e) Medical Travel Referral Benefit shall be in accordance with the provisions set out in Appendix F.

(Important to refer to electronic brochure for specific details of terms and limits.)

22.4 Dental Care Plan

The Employer pays, for eligible faculty members in accordance with Article 22.1, one hundred percent (100%) of the premium cost of the Dental Plan. The Plan shall provide coverage including:

(a) one hundred percent (100%) of basic diagnostic, preventative, restorative and periodontal services (Plan A). Plan A includes revision of cleaning the teeth (prophylaxis and scaling) every nine (9) months except dependent children (up to age 19) and those with gum disease and other dental problems as approved by the Plan;

- (b) sixty percent (60%) of major restorative services such as crowns, bridges, and dentures (Plan B);
- (c) fifty percent (50%) of orthodontia (Plan C) to a maximum of three thousand dollars (\$3,000).
- (d) Dental Plan interpretation shall be in accordance with the provisions set out in Appendix H.

(Important to refer to electronic brochure for specific details of terms and limits)

22.5 Group Life Insurance/Accidental Death and Dismemberment Benefit Plan

All eligible regular faculty members must participate in the Group Life/Accidental Death and Dismemberment Insurance plans.

The Employer contributes one hundred percent (100%) of a Group Life insurance premium for eligible faculty members in accordance with Article 22.1, which provides for three times (3x) annual salary and Accidental Death and Dismemberment also three times (3x) annual salary each. (Important to refer to electronic brochure for specific details of terms and limits.)

22.6 Disability Benefits

(a) The Employer shall continue a single plan for the provision of Disability benefits for eligible faculty members.

(b) The Disability benefits plan will be as set out in the findings of the Joint Committee on Benefits Administration (JCBA) entitled Long-Term Disability Benefit Initiative, but will be an insured plan and will include the following elements:

(1) Benefit level of sick leave at one hundred percent (100%) for the first thirty (30) calendar days, Short-Term Disability at seventy percent (70%) weekly indemnity for the next twenty-one (21) weeks, and Long-Term Disability leave of seventy percent (70%) thereafter;

(2) Long-Term Disability as defined on the basis of two (2)-year own occupation and any other occupation thereafter as described by the JCBA plan;

(3) Health and Insurance benefit premiums will be paid by the Employer or the Plan for faculty members on sick leave, Short-Term Disability and Long-Term Disability;

(4) Employer payment of premiums for both Short-Term and Long-Term Disability benefits;

(5) Claims Review Committee made up of three (3) medical doctors (one [1] designated by the claimant, one [1] by the Employer and the third agreed to by the first two [2] doctors);

(6) mandatory rehabilitation as described in the JCBA plan;

(7) subject to provisions of the Plan, enrolment is mandatory for all active regular faculty members and for active non-regular faculty members employed on a continuing basis for at least a four (4) month period with fifty percent (50%) or more of a full-time workload as defined by local provisions.

(c) Faculty members shall retain any sick leave banks accrued up to but not beyond March 31, 2002, for the purpose of Short-Term Disability.

(d) The Employer shall provide each term faculty member who is not eligible for Disability benefits with sick leave to a maximum of ten (10) calendar days per illness. The pay shall be calculated on current term workload and pay rate.

(e) Medical Certificate

(1) The Employer may request a medical certificate for an absence in excess of three (3) days. The Employer will assume, where necessary, additional cost of obtaining the medical certificate.

(2) The Employer may request an independent medical certificate to confirm whether a faculty member is fit to return to work. The cost of securing such a certificate will be borne by the Employer.

(f) The faculty member shall make every reasonable effort to inform the appropriate Chair or the appropriate Dean, Director or Campus Administrator, if the appropriate Chair cannot be reached, of the inability to work because of illness or injury. Changes in anticipated date of return to duty shall be made

known to the appropriate Chair or the appropriate Dean, Director or Campus Administrator if the appropriate Chair cannot be reached.

(g) A faculty member is not eligible for sick leave with pay for any period of suspension, strike, layoff, or lock out.

(h) The Joint Committee on Benefits Administration (JCBA) shall oversee the continuation of the plan as described in Article 22.6(b) and shall address such matters pertaining to the plan as are included in the JCBA's mandate as set out in Article 22.11.

22.7 Medical Travel Expense Entitlement

Medical Travel Referral Benefit shall be in accordance with the provisions set out in Appendix F.

The medical travel referral benefit provides coverage to faculty members and their dependents for services and supplies while travelling in excess of a one hundred (100) kilometre radius of the faculty member's home campus for medical purposes where treatment is not available within the patient's own locale. The coverage limit for the medical travel referral benefit is one hundred and twenty-five dollars (\$125) per day for a maximum of fifty (50) days per calendar year for all expenses combined, on a per diem basis. Coverage will not be provided for travel and expenses incurred outside of BC except where the cost of travel to Alberta is less than the required travel within BC. Referral must be made by a physician to a specialist.

22.8 Coverage During Layoff

Coverage for the Health and Insurance benefits may be maintained by a faculty member who has been laid off for a maximum period of four (4) months provided the recall date is known to be within four (4) months at the time of layoff and the faculty member is willing to pay the Employer's share of the premiums in addition to the faculty member's share.

22.9 Absence Without Pay

Coverage for the Health and Insurance benefits may be maintained by a faculty member on unassisted leave, subject to the terms and conditions of the applicable Health and Insurance benefit, provided that the faculty member prepays both the Employer's and faculty member's share of the premiums by postdated cheques or payroll deduction. Any difference between the prepaid amounts and any increase in the premium costs during the leave will be reimbursed by the faculty member to the Employer upon termination of the leave.

22.10 Termination of Coverage

(a) Non-Retiring Faculty Members

Health and Insurance benefits coverage will cease on the day that a faculty member's employment terminates.

(b) Retiring Faculty Members

Retiring faculty members who are eligible for health benefits and who have applied for College Pension Plan benefits will maintain coverage until the commencement of Pension Plan Health benefits and, in any event, no later than ninety (90) calendar days following the date of the faculty member's retirement.

22.11 Joint Committee on Benefits Administration

(a) Committee Established

The Parties agree to participate in the post secondary sector's Joint Committee on Benefits.

(b) Committee Mandate

The Joint Committee on Benefits has a mandate to undertake tasks related to Health and Insurance benefits and Disability benefits including:

(1) Comparison and analysis of contract administration and costs, using criteria developed by the committee.

(2) Monitoring carrier performance including receiving reports from the plan administrator(s).

(3) Reviewing the cost effectiveness and quality of benefit delivery, service, and administration by carriers, including access issues.

(4) Tendering of contracts.

Participation in the existing Benefits User Group will continue. Participation is open to institutions that are not currently members of the Benefits User Group.

(c) Constraints (JCBA)

The Joint Committee on Benefits will not make any changes to the plan provisions that would increase the costs of Health and Insurance benefits or Disability benefits for individual Employers or reduce plan provisions without the agreement of the parties to this Agreement.

(d) Costs of the Joint Committee

The Employers will pay up to fifty thousand dollars (\$50,000) for the costs of the Committee's operations, exclusive of salaries and benefits.

22.12 ESA Sick Leave

Faculty who have completed 90 consecutive days of employment, and who are not eligible for benefits pursuant to Article 22.6, shall be entitled to up to five (5) paid days per calendar year of personal illness leave in accordance with Section 49.1 of the *Employment Standards Act*.

ARTICLE 23 - COLLEGE PENSION PLAN

Mandatory and optional enrolment in the College Pension Plan are based on eligibility rules as set out in the College Pension Plan regulations and applicable government legislation affecting the College Pension Plan.

ARTICLE 24 - WAGES AND ALLOWANCES

24.1 Wages

For wages effective April 1, 2022 to March 31, 2025 and for the duration of this contract refer to Appendix A.

24.2 Paydays

(a) Payment of wages will be processed on a biweekly basis and will be directly deposited to the credit of the faculty member's account at a financial institution in Canada of the faculty member's choice.

(b) A comprehensive statement shall accompany each paycheque detailing all payments, allowances and deductions.

(1) The distribution of paycheques shall be done in such a manner that the details of the paycheque shall be confidential.

24.3 Chairs Stipend and/or Release Time

(a) Chairs will receive Chair Stipends as set out below.

(b) Release time for Chair duties may be provided, at the discretion of the Employer, taking into account the workload of the individual Chair.

	April 1,	April 1,	April 1,
	2022	2023	2024
More than 0.9286 FTE per week	\$133.50	\$142.51	\$146.79*
Teaching Assignment	bw	bw	bw
More than 0.8571 FTE per week	\$66.76	\$71.27	\$73.41*
Teaching Assignment	bw	bw	bw

24.4 First Aid

The Union and the Employer agree that first aid regulations made pursuant to the *Workers' Compensation Act* shall be fully complied with. Where the Employer requires a faculty member to obtain or renew the faculty member's survival or occupational first aid certificate, the cost shall be borne by the Employer. Designated faculty members will be granted time off with pay during normal working hours to renew their certificate. A biweekly premium shall be paid to faculty members required to possess a certificate under this Article. The amount of the premium shall be:

Level 1	\$ 20 biweekly
Level 2	\$ 40 biweekly
Level 3	\$ 45 biweekly

24.5 Salary Placement

(a) Initial Salary Placement

(1) Credentials: For initial placement on the scale, credentials shall be recognized for placement consideration, subject to the following criteria:

(i) Credentials, for the purpose of this article, refer to the following accredited education credentials or equivalency (as confirmed by the Human Resources Department):

- Vocational Instructor's Diploma (VID) or Certificate in Adult and Continuing Education (CACE) or Provincial Instructor's Diploma (PID), where stated as a preferred qualification
- Two (2)-year Diploma
- Bachelor's Degree
- Post Degree Diploma, or B.Ed., or five (5)-Year Degree
- Master's Degree
- Doctorate

(ii) A qualified candidate will receive placement recognition of one (1) step for each credential they hold above the minimum required qualifications, as stated in the job posting, to a maximum of three (3) steps for all credentials.

(iii) If the credential(s) is included in the job posting as the minimum required qualification(s) for the position, there will be no placement recognition, except in the case of a Master's Degree.

(iv) An extra step(s) for a credential(s) will not be awarded where this would result in exceeding the initial maximum salary placement.

(2) Experience: For initial placement on the scale, the following experience shall be recognized, providing minimum job requirements or the equivalent are met, by giving one (1) step on the scale for:

(i) Teaching Experience:

An equivalent of one (1) step for each year of teaching experience beyond two (2) in an approved university, university-college, college, technical institution, or vocational school or, in the case of high school related instruction in vocational, commercial, remedial and upgrading subjects.

All directly relevant teaching experience in the public school system will receive full credit.

Teaching experience less than three (3) years will be added to any industry experience and be subject to the four (4) year minimum experience and the two (2) years for one (1) increment.

Teaching non-credit courses is not included.

(ii) Work Experience:

An equivalent of one (1) step for each two (2) years of relevant work experience beyond four (4) in industry or business or related experience.

Part-time experience will be prorated on the two (2) semester year. Other part-time work experience prorated on the basis of a thirty-five (35) hour workweek and twelve (12) month year. Maximum credit for one (1) year is one (1) year.

(3) Salary Placement Review:

All salary placement reviews must be filed with the Associate Vice-President, Human Resources within ninety (90) days of receiving the Initial Salary Placement form completed by the Human Resources Department. The Salary Placement Review Committee will be comprised of two (2) members appointed by the Union and two (2) members appointed by the Employer.

The decision of the Salary Placement Review Committee will be provided to the faculty member in writing and is final and binding.

Notwithstanding 24.5(a)(1) and (2), non-Western credentials or lived experience may be considered when placing an Indigenous employee on the BCGEU wage scale, in relation to specific positions as mutually agreed to by the Union and the Employer.

(4) Forms: The Salary Data Collection form will be given to the faculty member at the offer of employment and will be submitted as soon as possible to the Human Resources Department.

(5) The above criteria shall be used to a maximum placement of Step 6 on the salary scale. Notwithstanding the above, in departments where it is difficult to recruit faculty, placement above Step 6 shall be referred to the Salary Placement Review committee under Article 24.5(a)(3) for a final and binding decision.

(6) Any former faculty member of the Employer, hired on a subsequent appointment more than twenty-four (24) months after the end of a previous appointment, shall be placed on the salary scale in accordance with the initial salary placement provisions of Article 24.5(a).

(b) Post Placement Credential Progression

(1) A faculty member shall qualify for one (1) additional increment above their existing salary upon successfully qualifying for the following credential or equivalency (as confirmed by the Human Resources Department), provided it is above the minimum required qualifications in their current position:

- Two (2)-year Diploma
- Bachelor's Degree
- Post Degree Diploma of B.Ed., or five (5)-year Degree
- Master's Degree
- Doctorate

(2) The effective date for such changes in category shall be effective from the first day of the next pay period following formal notification received by the Employer of the successful completion of the credential.

(3) A faculty member will receive incremental progression for credentials to a maximum of three (3) increments for the duration of their employment with Vancouver Island University.

(c) Maintenance of Placement

Where a qualified candidate previously employed by another institution as referred to under Article 12.8(d), or an equivalent institution outside of British Columbia and/or the PSEA sector, becomes employed within two (2) years by Vancouver Island University, initial placement shall be made at the higher of the placement formula at Vancouver Island University or their most recent salary step from the other institution. This will only apply when the faculty member becomes employed in the same or a substantially similar field. The normal probation provisions of Vancouver Island University will apply.

(d) Service Progression

Faculty members shall progress through all incremental steps of the salary scale.

(1) Regular Faculty Members

(i) For each year of service to the Employer, a regular faculty member is entitled to one (1) increment on the salary scale until the maximum is attained. It is understood that a year's service is equivalent to 1.0 FTE (see Appendix I) worked (i.e. 1.000 increment credit) and at least one (1) calendar year since the start of service or date of last increment. An increment will be effective on the first day of the pay period following the date on which the year's service is reached.

(ii) Any auxiliary hours worked by term or regular faculty members shall be recognized for incremental purposes.

(iii) Effective October 1, 1989, a faculty member who has term seniority in accordance with this Agreement who is appointed to a regular appointment shall have their first increment adjustment as a regular faculty member based on accrued increment hours accumulated as a term faculty member and service as a regular faculty member. Thereafter, the faculty member's incremental progression shall be applied in accordance with the provisions of Article 24.5(d)(1)(i) and (ii).

(2) Term Faculty Members

(i) For each year of service to the Employer, a term faculty member is entitled to one (1) increment on the salary scale until the maximum is attained. It is understood that a year's service is equivalent to 1.0 FTE (see Appendix I) worked (i.e. 1.000 increment credit) and at least one (1) calendar year since the start of service or date of last increment. An increment will be effective on the first day of the pay period following the date on which the year's service is reached.

(ii) Any auxiliary hours worked by term or regular faculty members shall be recognized for incremental purposes.

(3) Auxiliary Faculty Members

(i) New auxiliary faculty members (hired after date of signing) will be placed at the minimum step for the first thirty (30) calendar days.

(4) Notwithstanding the above and where required by job posting, all faculty must complete the Provincial Instructor Diploma or a Master's degree before progressing to Step 1 of the salary scale after time of hire. This clause shall not apply to employees hired prior to the date of ratification this Agreement.

24.6 Clothing Supply

(a) If a particular type of work clothing or special apparel is required by the nature of the faculty member's job, such clothing or apparel shall be provided by the Employer.

(b) The Employer agrees to supply at no cost to faculty members all pieces of health and safety apparel and equipment required by Workers' Compensation.

24.7 Cleaning

The Employer shall be responsible for laundering, dry cleaning and maintenance of all apparel supplied by the Employer.

24.8 Health and Safety Apparel and Equipment

The Employer will supply, at no cost to employees, all health and safety apparel and equipment that WorkSafeBC requires the Employer to provide.

In areas designated by the WorkSafeBC or the Employer where safety footwear is required to be worn, the Employer will pay one hundred percent (100%) of the cost for safety footwear.

24.9 Independent Study Stipend

Independent study shall be provided on a voluntary basis in addition to a faculty member's assigned workload. These studies will be provided by agreement of the faculty member, the Chair and the Dean, Director or Campus Administrator and will have a maximum of five (5) students. The faculty member will

receive a stipend of two hundred and fifty dollars (\$250) per student per term or semester. The Dean, Director or Campus Administrator will identify independent studies in the workload report.

24.10 Professional Association Fees

Where, as a condition of employment and as specified in the job posting, a regular faculty member is required to belong to a professional association, Vancouver Island University shall reimburse the faculty member for the fee required to maintain membership in that professional association. This provision shall be effective January 1, 2024.

ARTICLE 25 - TECHNOLOGICAL CHANGE

The procedure to be followed by the Employer and the Union concerning technological change shall be in accordance with the British Columbia *Labour Relations Code*.

ARTICLE 26 - PROGRAM TRANSFERS AND MERGERS

26.1 Notice of Program Transfer / Merger

When the Employer decides to transfer or merge a program or a partial program and the transfer or merger will result in the transfer or layoff of one (1) or more faculty members, the Employer will provide written notice to the Union(s) as soon as possible, but no less than sixty (60) calendar days prior to the date of transfer or merger.

26.2 Transfer/Merger Agreements

When notice is served, a Committee composed of equal representation from the Employer and the Union(s) representing faculty members affected by the transfer or merger will be formed to negotiate a transfer/merger agreement.

The transfer/merger agreement will address all relevant matters and will be signed by each of the parties.

A copy of the agreement will be provided to each affected faculty member.

26.3 Disputes

Grievances arising prior to the transfer/merger date remain the responsibility of the sending institution.

If a dispute arises as a result of a program transfer/merger and/or its faculty members being transferred the matter will be referred to the JADRC for resolution.

ARTICLE 27 - COPYRIGHT AND INTELLECTUAL PROPERTY

27.1 Copyright Ownership

The copyright or patent for any work product, including creative work, instructional strategies or curriculum/instructional material, software or any other material or technology that may be copyrighted or patented:

(a) belongs to the faculty member(s) where the work product has been prepared or created as part of assigned duties, other than the duties listed in Article 27.1(b) below, and the copyright to all copyrightable material shall be the sole property of the faculty member(s) and shall be retained throughout their lifetime and upon their death by their heirs or assigns; and

(b) belongs to the Employer where one (1) or more faculty members:

(1) have been hired or agree to create and produce copyrightable work product(s) for the Employer, or

(2) are given release time from usual duties to create and produce copyrightable work product(s), or

(3) are paid, in addition to their regular rate of pay, for their time in an appointment to produce copyrightable work product(s).

27.2 Employer Rights to Materials Copyrighted by Faculty Member(s)

Where the faculty member holds the copyright pursuant to Article 27.1(a), the Employer shall have a right to use their copyrighted material in perpetuity for institutional purposes. The Employer may amend and update the copyrighted material with the approval of the faculty member(s) holding the copyright to the material. Such approval will not be unreasonably withheld.

27.3 Faculty Member Rights to Materials Copyrighted by the Employer

Where the Employer holds the copyright pursuant to Article 27.1(b), the faculty member(s) shall have the right to use in perpetuity, free of charge, such copyrighted material. The faculty member may amend and update the copyrighted material with the approval of the Employer holding the copyright to the material. Such approval will not be unreasonably withheld.

27.4 Joint Review

JADRC may, at the request of either party, review issues arising from the application of this article.

ARTICLE 28 - DISTRIBUTED LEARNING

(a) Distributed learning includes, both direct and indirect instruction, but is not limited to, print based education courses, online or web-based instruction, video-conferencing, teleconferencing, instructional video and audio tapes, hybrid or mixed-mode programs and courses.

(b) In developing and offering distributed learning programs and courses, the Employer will plan in collaboration with the department or functional area and the faculty member(s) who will develop and/or deliver the program or course.

(c) For the purposes of this article, departments or functional areas are defined as the operational or administrative sub-division of the Employer within which a faculty member is appointed and assigned workload and may include geographic limitations.

(d) Subject to mutual agreement, the parties may develop criteria for the determination of the appropriate release time for the development, delivery and revisions of distributed learning programs or courses.

(e) The Employer will provide the necessary technological and human resources for faculty members assigned to develop and deliver the program and courses.

(f) The Employer will provide the necessary and appropriate training in the use of relevant educational technology for faculty members assigned to deliver distributed learning programs and courses.

(h) Faculty members shall not be required to deliver distributed learning programs/courses from their home. Faculty members delivering or developing distributed learning courses shall be provided with office space and the appropriate technology to support them in their work.

(i) Where a faculty member has been assigned an online course and agrees to the Employer's request to teach all or part of that course from home, the Employer shall provide the appropriate technology and pay for the reasonable and approved costs of delivering those courses from home.

(j) No regular faculty member will be laid off as a direct result of the introduction of distributed learning or education technology.

ARTICLE 29 - NON-INSTRUCTIONAL DUTY DAYS FOR FACULTY MEMBERS

In order to allow faculty members time to prepare for incoming classes and to conclude the affairs of previous classes, provisions shall be made for the scheduling of non-instructional duty days to a maximum of ten (10) per year.

For the period between Christmas and New Years Day when institutional services have been minimized, the normal attendance expectations for BCGEU members will be relaxed provided that they have completed their duties.

ARTICLE 30 - POSTED JOB COMPETITIONS

30.1 Job Postings

(a) Where a regular or term job vacancy within the bargaining unit except short term vacancies not to exceed thirty (30) continuous working days is to be filled, notice shall be posted on the University's website for a minimum period of seven (7) calendar days.

(b) A copy of the appointment letter will be forwarded to the bargaining unit chairperson, or designate, after the close of each competition.

(c) Stewards will be given access to applicable HRIS reports. When a steward accesses the name, seniority, position, pay step, or other employee information, it is subject to privacy legislation.

(d) Work that faculty members are regularized into is not subject to posting.

(e) Work will not be posted until eligible faculty members who are affected by layoff under Article 12.10 are considered.

30.2 Qualifications

The Employer has the responsibility to set position qualifications. Position qualifications will be set in good faith and reflect the program needs.

30.3 Transfers

Current regular faculty members will be given seniority preference in respect of transferring into vacancies within the bargaining unit, merit considered. Factors used to determine merit will be educational qualifications, skills, ability and experience and any other matters which are necessary or desirable, having regard to the nature of the duties to be performed and consistent with the job requirements.

30.4 Term Appointment Vacancies

Once regular full-time faculty members have exercised their seniority in accordance with Clauses 12.1(c) (Seniority Defined-Additional Hours), 12.10 (Reinstatement of Regular Faculty Member) and 30.3 (Transfer), regular part-time faculty members may claim additional term hours (per Clause 12.1[c]), to a maximum of full-time work, as available in the same instructional program (as listed in Appendix C).

Term faculty members will then be given preference in respect of consideration for term appointment vacancies only, merit considered. Factors used to determine merit will be education, qualifications, skills, ability and experience and any other relevant matters which are necessary or desirable, having regard to the nature of the duties to be performed and consistent with the job requirements.

In the event that merit is assessed as relatively equal, seniority shall be the determining factor. The Employer agrees that the provisions of Clause 30.4 will not be utilized to avoid the Employer's obligation to convert temporary appointments to regular status.

30.5 Equity Deserving Groups

The Employer will consult with the Union prior to posting positions where equity deserving candidates are being sought, subject to rights being exercised under Article 30.3. In such cases Article 30.4 may be waived with a mutual agreement of the Union and the Employer.

ARTICLE 31 - TRANSFERS AND SECONDMENTS

31.1 Short Term Transfers

(a) The Employer agrees short-term transfers will be by mutual agreement.

(b) Employer initiated transfer to a campus other than the one which a faculty member was originally assigned will be assisted by the Employer in the form of, but not necessarily limited to, moving expenses which shall include the cost of moving and insuring the move of household furniture and personal effects.

31.2 Secondments

A process by which the Employer may assign a faculty member to another agency, board, society, commission or employer. The Employer agrees that faculty members will not be seconded against their will. The Employer will seek to provide a faculty member with four (4) weeks written notice of secondment. Where possible, the written notice of secondment shall indicate the term of secondment.

ARTICLE 32 - TRAVEL

32.1 Travel on University Business

Faculty members required to travel on Employer business will be reimbursed for meals, lodging, and transportation as per the policy of the University Board. Receipts for actual reasonable expenses are to be submitted for meals or a flat per diem rate as set out in the Board Policy.

ARTICLE 33 - INDEMNITY/INSURANCE

(a) Where a faculty member is charged with an offence resulting directly from the proper performance of duties and is subsequently found not guilty, the faculty member shall be reimbursed for reasonable legal fees.

(b) Any damages shall be paid up to the policy limit required by the Ministry of Education due to liability imposed by law, including cost of defence, resulting from injury and/or damage to the person or property of others including personal injury, professional errors and omissions.

(c) Providing the Employer requests the use of tools, reference texts and instruments, and the declared value is recorded in writing with the Dean, Director or Campus Administrator at the time the items are brought on campus, coverage is for, "All risks of direct physical loss or damage consistent with policy provisions".

ARTICLE 34 - CLASS SIZE

Where a faculty member's class size is to be increased due to extraordinary circumstances, the maximum class size will be determined through consultation with the Faculty Members, Chair, Dean, Director or Campus Administrator. Where a determination is not made through consultation in accordance with the above, the faculty member may refer the matter in writing (copy to Dean, Director or Campus Administrator and the Joint Labour Management Committee) to the Vice-President, or President, as applicable, for a final determination.

The Employer agrees to adhere to WorkSafeBC regulation.

ARTICLE 35 - THE HIGH SCHOOL AT VIU

35.1 Field Trips; Extra Curricular Programs and Activities; Student Teachers; Volunteers

The Employer and the Union agree to facilitate and enhance the operation of The High School at VIU acknowledging the particular pedagogical dimensions of The High School at VIU instructional services delivered to this unique international, school-aged population.

(a) Field Trips

Field trips inherent in this program require that instructional duties be, on occasion, performed away from Employer premises and as such are not voluntary.

(b) Extra Curricular Programs and Activities

The Union and the Employer recognize that the involvement of faculty in extra-curricular programs and activities contribute significantly toward a positive, integrated learning environment. Extra curricular programs and activities are those which normally occur beyond the assigned contact hours.

Faculty are encouraged to participate in extra curricular activities; however, the Employer recognizes that such involvement is voluntary.

(c) Student Teachers

In consultation with the faculty, the Principal of The High School at VIU will place student teachers.

(d) Volunteers

The Parties agree to recognize and encourage the presence of host parents and families, community and University faculty member volunteers as a support to instruction, both in the classroom and on field trips.

35.2 Principal's Teaching Assignment

The Employer and the Union agree that the Principal of The High School at VIU may be expected to teach in the program to a maximum of one (1) course in two (2) of the semesters each academic year. This requirement is in recognition of the following points:

(1) represents the current operating practices of both public and private small secondary schools;

(2) addresses an economic and educational imperative of sound school administration at this level;

(3) is advocated by the The High School at VIU Board.

It is agreed that the performance of this teaching assignment will not result in the incumbent being placed in the bargaining unit or receiving BCGEU payment for this instruction.

The Employer agrees that the teaching assignment for the Principal may be reduced or eliminated in an academic year were such assignment to result in a reduction to the annual teaching workload for BCGEU faculty engaged in The High School at VIU instruction.

It is agreed that the Principal will not teach in an area for which existing faculty are qualified and enabled by the timetable to teach the course. The timetable will be developed so as not to limit teaching opportunities.

ARTICLE 36 - LABOUR MANAGEMENT COMMITTEE

36.1 Composition of Labour Management Committee

There shall be established a Labour Management Committee composed of members equal in number, represented by the Employer and the Union. The minimum size of this Committee shall be two (2) Union representatives and two (2) Employer representatives, and the maximum size shall be four (4) Employer representatives and four (4) Union representatives. This Committee may call upon additional persons for technical information or advice. The Committee may establish sub-committees or "ad hoc" committees as it deems necessary and shall set guidelines and operating procedures for such committees.

36.2 Meetings

The Labour Management Committee shall meet at the call of either Party at a mutually agreeable time and place. Faculty members shall not suffer any loss of basic pay for time spent on this Committee.

36.3 Chairperson

An Employer representative and a Union representative shall alternate in chairing meetings.

36.4 Committee Jurisdiction

This Committee shall not have jurisdiction over wages or any other matter of collective bargaining including the administration of this Agreement.

The Committee shall have the power to make recommendations to the Union and the Employer on the following general matters:

(a) reviewing matters, other than grievances, relating to the maintenance of good relations between the parties;

(b) correcting conditions causing grievances and misunderstandings.

36.5 Labour Management Minutes

Once the Labour Management Committee minutes have been adopted by the Committee, the minutes may then be posted on the Union Board(s) and may be posted in an employee accessed electronic location.

ARTICLE 37 - INTERNATIONAL EDUCATION

The Parties agree that participation in international education is important and valuable, enhancing student and faculty opportunities.

The Parties agree that this Article shall govern the terms and conditions for faculty who travel outside Canada and the U.S. to perform assigned work pursuant to the Collective Agreement.

37.1 General

- (a) Faculty participation in international education is voluntary.
- (b) Subject to Article 37.1(d) below, the terms and conditions of the Collective Agreement will apply.

(c) The Employer will meet and review the terms and conditions for each assignment outside Canada and the U.S. with the faculty member participating in an international education project.

(d) Should an international education project require the interpretation of the workload provisions of the Collective Agreement, the Employer will apply to the faculty such workload expectations as are reasonably equivalent to those workload expectations that would normally apply.

37.2 Expenses

(a) The Employer will reimburse, pursuant to Policy 42.08, receipted expenses incurred by a faculty member while on Employer business. The Employer may grant a sufficient travel advance to cover those expenses that can reasonably be anticipated prior to travel, including appropriate transportation, accommodation and meal expenses.

(b) The Employer will waive the requirement that receipts be provided in situations where these are not reasonably obtainable. In such cases, a per diem allowance consistent with Policy 42.08 will be provided to the faculty member.

37.3 Health and Insurance Benefits

(a) Subject to eligibility pursuant to the Collective Agreement, the Employer will provide current Health and Insurance benefits coverage for faculty working under this Article. Premiums for this coverage will continue to be paid as if the faculty member was continuing to work for the Employer in British Columbia.

Limitations:

(1) Dental expenses incurred will be reimbursed based on the British Columbia fee schedule in effect under the Employer's group policy.

(2) Benefit coverage will not extend beyond the date the policy or any benefits terminate with the Employer's insurance carrier.

(3) When a faculty member is working in countries where payment for medical services may require cash payment, the faculty member will submit their claim to the insurance carrier for

reimbursement of such expenses. The Employer will advance monies in such instances if there is an anticipated delay from the insurance carrier.

(4) A faculty member will be referred to the Human Resources Department to clarify the benefit and travel medical insurance coverage.

(5) In no event will a faculty member who is eligible for benefits receive greater benefits than they would have been covered for while employed at their home campus.

(b) Notwithstanding the provisions of 37.3(a), in cases where a faculty member does not have Extended Health Benefit coverage, the Employer will supply travel medical insurance. Article 37.3 also applies to travel in the United States.

37.4 Emergencies and Emergency Evacuation

The Faculty of International Education shall maintain procedures for dealing with emergencies while working overseas and shall provide same to a faculty member as part of their orientation.

37.5 Orientation

Faculty working under this Article will receive an orientation prior to departure that includes but is not limited to:

- (a) the project;
- (b) the culture and country;
- (c) travel, safety or medical concerns, benefits issues;
- (d) emergency procedures; and
- (e) other issues related to the work.

37.6 Return

The Employer will arrange the scheduling of international work in such a way that faculty members will be provided up to three (3) days inclusive of required travel time between the completion of their international education assignments before assuming regular duties at the institution. The duration of travel shall be a factor in determining the number of days off upon return. This will not apply in situations where a faculty member elects to extend their stay through the use of vacation.

ARTICLE 38 - TERM OF AGREEMENT

38.1 Duration

This Agreement shall be binding and remain in effect from 12:01 a.m., April 1, 2022 to midnight, March 31, 2025. The Union and the Employer desire every faculty member to be familiar with the provisions, rights and obligations inherent in this Agreement. For this reason, the Employer shall print sufficient copies of the Agreement for distribution to faculty members within sixty (60) calendar days from the signing of the Agreement.

38.2 Notice to Bargain

(a) This Agreement may be opened for collective bargaining by either Party giving written notice to the other Party on or after December 1, 2024 but in any event, not later than December 31, 2024.

(b) Where no notice is given by either Party prior to December 31, 2024, both Parties shall be deemed to have been given notice under this Article on December 31, 2024, and thereupon Article 38.3 of this Agreement applies.

(c) All notices on behalf of the Union shall be given by the President of the Union and similar notices on behalf of the Employer shall be given by the Vice-President or President as applicable.

38.3 Commencement of Bargaining

Where a Party to this Agreement has given notice under Article 38.2 of this Agreement the Parties shall, within ten (10) calendar days after the notice was given, commence collective bargaining.

38.4 Changes in Agreement

Any changes deemed necessary under this Agreement may be made by mutual agreement at any time during the life of this Agreement. Such agreed changes shall be incorporated into this Agreement as an addendum.

38.5 Agreement to Continue in Force

Both Parties shall adhere fully to the terms of this Agreement during the period of bona fide collective bargaining.

The Parties hereby agree to recommend to their respective principals an April 1, 2022 March 31, 2025 Agreement as amended from the current contract, subject to ratification by the University Board and the BCGEU membership.

Page 68

SIGNED ON BEHALF OF	SIGNED ON BEHALF OF	
THE UNION:	THE EMPLOYER:	
Paul Finch	Deborah Saucier	
President	President & Vice-Chancellor	
Doreen Smith	Patricia Elliott	
Staff Representative	Director, Employee Relations	
Dean Cadieux, Bargaining Unit Chair Bargaining Committee Chairperson	EMPLOYER BARGAINING COMMITTEE:	
Malena Acosta	Glynis Steen	
Bargaining Committee	Dean, Trades & Applied Technology	
Brad Jannaway	Jean Maltesen	
Bargaining Committee	Dean, Academic and Career Preparation	
SIGNED ON BEHALF OF PSEA:	Lynda Phillips Associate Dean, Health Sciences and Human Services	
Doug Campbell, Chair Post-Secondary Employers' Associate Board of Directors	Jessie Magee-Chalmers Campus Administrator, Cowichan	
Original convisioned on February 4	2025	

Dated Original copy signed on February 4, 2025

APPENDIX A

All salary scales for classifications or positions in the collective agreement shall be increased by the following percentages effective on the dates indicated.

(a) Effective April 01, 2022, all annual salary scales in the Collective Agreement which were in effect on March 31, 2022 shall have each step increased by \$455. The resulting rates of pay will then be increased by a further 3.24%. The new rates of pay shall be rounded to the nearest whole cent or dollar as applicable.

(b) Effective April 1, 2023, all salary scales in the Collective Agreement which were in effect on March 31, 2023 shall be increased by 6.75%. The new rates of pay shall be rounded to the nearest whole cent or dollar as applicable. (Note: This increase includes the COLA amount 1.25% in addition to the 5.5% general increase.)

(c) Effective April 1, 2024, all salary scales in the Collective Agreement which were in effect on March 31, 2024, shall be increased by two percent (2%). The new rates of pay shall be rounded to the nearest whole cent or dollar as applicable.

(d) The salary increase in (c) above may be adjusted pursuant to Letter of Understanding 4 Re: Cost of Living Adjustment.

These salary increases shall apply to all current employees who are members of the bargaining unit on the date of ratification of the MOA and any employees who retired on or after April 1, 2022. Retroactive payments for current employees will be paid within 4 pay periods of the date of ratification.

	01–Apr-22	01-Apr–23	1-Apr-24
Step	to	to	to
	31–Mar-23	31-Mar-24**	31-Mar-25**
1	\$102,655	\$109,584	\$112,872
2	\$96,195	\$102,688	\$105,769
3	\$89,635	\$95,685	\$98,556
4	\$85,990	\$91,794	\$94,548
5	\$82,854	\$88,447	\$91,100
*6	\$79,725	\$85,106	\$87,659
7	\$76,590	\$81,760	\$84,213
8	\$73,459	\$78,417	\$80,770
9	\$70,326	\$75,073	\$77,325
10	\$67,192	\$71,727	\$73,879
11	\$64,061	\$68,385	\$70,437

Table A1 Provincial Salary Scale

* Maximum initial placement in category

** These rates have been adjusted pursuant to Letter of Understanding 4 Re: Cost of Living Adjustment

Table A2 2% Salary Stipend

	01–Apr-22	01-Apr–23	1-Apr-24
Step	to	to	to
	31–Mar-23	31-Mar-24**	31-Mar-25**
1	\$104,708	\$111,776	\$115,129
2	\$98,119	\$104,742	\$107,884
3	\$91,428	\$97,599	\$100,527
4	\$87,710	\$93,630	\$96,439
5	\$84,511	\$90,216	\$92,922
*6	\$81,320	\$86,808	\$89,412
7	\$78,122	\$83,395	\$85,897
8	\$74,928	\$79,985	\$82,385
9	\$71,733	\$76,574	\$78,872
10	\$68,536	\$73,162	\$75,357
11	\$64,061	\$68,385	\$70,436

*Maximum initial placement in category.

** These rates have been adjusted pursuant to Letter of Understanding 4 Re: Cost of Living Adjustment

APPENDIX B EMPLOYEE AND FAMILY ASSISTANCE PROGRAM

The current Employee and Family Assistance Program will remain in place.

APPENDIX C DEPARTMENTAL LISTING

Faculty	Departmental Listing	Including
Academic and Career	Aboriginal Bridging Programs	University and College Entrance
Preparation		Preparation, Aboriginal
		University Bridging Program
	Adult Basic Education	
	Adult Special Education	Employment and Life Skills Training (ELST), Workplace Essential Skills and Training (WEST), Employability Skills
	English as a Second Language Program (ESL)	
	Literacy	Volunteer Tutor Program
	Post-Secondary Education Programming	Cooperative Entrepreneur Training Program, Foundations for Success

BCGEU (Faculty) and Vancouver Island University (03/2025)

Faculty	Departmental Listing	Including
	The High School	
Health Sciences & Human Services	Dental Assistant	
	Health Care Assistant	
	Practical Nursing	
	Professional Development & Training	Activity Assistant Animal Care Aide Community Mental Health Worker Dental Office Receptionist Foundations in Mental Health Hospital Unit Coordinator Medical Office Assistant Medical Device Reprocessing Technician
Trades & Applied Technology	Automotive Service Technician	Foundation & Apprenticeship
	Business Programs	Office Administration
	Carpentry	Foundation & Apprenticeship
	Culinary Arts	Foundation & Apprenticeship
	Electrical Program	Foundation & Apprenticeship
	Esthetics	
	Hairstylist	Foundation & Apprenticeship
	Refrigeration Air-Conditioning Mechanic (RACM)	Foundation and Apprenticeship
	Heavy Equipment Operator	
	Heavy Mechanical Trades	Foundation and Apprenticeship
	Horticulture	
	Information Technology and Applied Systems (ITAS)	
	Marine and Motorcycle Technician	Foundation and Apprenticeship
	Power Engineering	
	Professional Baking	Foundation & Apprenticeship
	Welding	Foundation and Apprenticeship
Various	Professional Development & Training (as covered under Memorandum of Understanding #1)	

APPENDIX E1 REGISTRY OF LAID OFF EMPLOYEES – FORM 1

PSEA ELECTRONIC POSTING OF AVAILABLE POSITIONS

- 0. (For PSEA use only)
- 1. College/University College/Institute and Location:
- 2. Job Title:
- 3. Area/Program/Discipline(s):
- 4. Job Description:
- 5. Minimum and Preferred Qualifications:
- 6. Start Date:
- 7. Close Date:
- 8. Contact Person and Address:

APPENDIX E2 REGISTRY OF LAID OFF EMPLOYEES – FORM 2

PSEA REGISTRY OF ELIGIBLE EMPLOYEES

(For PSEA use only:)

- 1. College, University College, Institute:
- 2. Registrant:
- 3. Service Date (length of service):
- 4. Program/Area:
- 5. Date of Availability (Lay-off or End of Contract):

Registrant Electronic Resume available at:

College/University College/Institute Contact Person:

College/University College/Institute Contact Phone Number:

Bargaining Unit Contact Person:

Bargaining Unit Contact Phone Number:

Information Release Waiver for the purposes of the Freedom of Information and Protection of Privacy:

I agree that the above personal information including my Resume (if available) can be made available to prospective Institutional Employers and Union via the internet or other means.

Signature of Registrant

Date

APPENDIX F MEDICAL TRAVEL REFERRAL BENEFIT

Benefit Summary

Deductible Amount: None

Benefit Amount: One hundred percent (100%) of eligible expenses

Individual Maximum: Ten thousand dollars (\$10,000) per year

Coverage Limitations:

- One hundred and twenty-five dollars (\$125) per day for a maximum of fifty (50) days per calendar year for all eligible expenses combined;
- Where an employer requires it, receipts must be submitted with the expense claim;
- Where the eligible expenses exceed one hundred and twenty-five dollars (\$125) per day, but do not exceed the average of one hundred and twenty-five dollars (\$125) per day for the year, the average will be paid. For example, where the expenses claimed in a given calendar year are one hundred and fifty dollars (\$150) day one (1), one hundred and twenty-five dollars (\$125) day two (2) and one hundred and sixty dollars (\$160) day three (3), a total of three hundred and seventy-five dollars (\$375) will be paid. Where the expenses claimed in a given calendar year are one hundred and fifty dollars (\$150) day one (1), seventy-five dollars (\$75) day two (2) and three hundred and fifty dollars (\$150) day one (1), seventy-five dollars (\$75) day two (2) and three hundred dollars (\$300) day three (3), a total of three hundred and seventy-five dollars (\$375) will be paid;
- Coverage will not be provided for travel and expenses incurred outside of BC except where the cost of travel to Alberta is less than the required travel within BC;
- Referral must be made by a physician to a specialist (a physician whose specialized services and treatments routinely performed are those that general practitioners do not perform).

List of Eligible Expenses

Medical Travel: When ordered by the attending physician because in their opinion adequate medical treatment is not available within a one hundred (100) kilometre radius of the faculty member's home campus, the following are included as eligible expenses:

- Public transportation (e.g. scheduled air, rail, bus, taxi and/or ferry);
- Automobile use as set out in the Policy or Collective Agreement (as applicable).

Accommodation: Where transportation has been provided as outlined above, accommodation in a commercial facility, Easter Seal House, Heather House, Vancouver Lodge, Ronald McDonald House, or other similar institutions as approved by the administrator, before and after medical treatment.

Meals: Where transportation has been provided above, reasonable and customary expenses for meals as set out in the Policy or Collective Agreement (as applicable).

Attendant: Where necessary, and at the request of the attending physician, transportation and accommodation of an attendant (e.g. family member or registered nurse) in connection with expenses incurred under items 1 and 2 above.

Exclusions

No benefit shall be payable for:

- Charges which are considered an insured service of any provincial government plan;
- Charges which are considered an insured service under the Extended Health benefits, or any other group plan in force at the time;
- Charges for a surgical procedure or treatment performed primarily for beautification, or charges for hospital confinement for such surgical procedure or treatment;
- Charges for medical treatment, transport or travel, other than specifically provided under eligible expenses;
- Charges not included in the list of eligible expenses;
- Charges for services and supplies which are furnished without the recommendation and approval of a physician acting within the scope of their license;
- Charges which are not medically necessary to the care and treatment of any existing or suspected injury, disease or pregnancy;
- Charges which are from an occupational injury or disease covered by any Workers' Compensation legislation or similar legislation;
- Charges which would not normally have been incurred but for the presence of this coverage or for which the faculty member or dependent is not legally allowed to pay;
- Charges which the administrator is not permitted, by any law to cover;
- Charges for dental work where a third party is responsible for payments of such charges;
- Charges for bodily injury resulting directly or indirectly from war or act of war (whether declared or undeclared), insurrection or riot, or hostilities of any kind;

- Charges for services and supplies resulting from any intentionally self-inflicted wound;
- Charges for experimental procedures or treatment not approved by the Canadian Medical Association or the appropriate medical speciality society;
- Charges made by a physician for travel, broken appointments, communication costs, filling in forms, or physician's supplies.

Claims Adjudication

To claim benefits, the faculty member or dependent must:

- Submit original receipts or photocopies of receipts if accompanied by an explanation of benefits from another carrier, and a claim form;
- Provide explanation and proof to support the claim including itemized bills and the attending physician's statement that the referral to the location where treatment was received was medically necessary;
- Provide explanation and proof to support the claim that an attendant (if any) was necessary and made at the request of an attending physician.

APPENDIX G DENTAL PLAN

The nine (9) month limitation applies to 1) polishing, 2) the application of fluoride, and 3) the recall itself. The nine (9) month limitation does not apply to scaling; any current scaling limits in dental contracts apply.

The process for an individual faculty member to have their teeth cleaned more frequently than every nine (9) months as provided by Article 22.4 is as follows:

- Faculty member visits dentist as usual.
- Dentist advises that the faculty member has gum disease or other dental problem which requires cleaning more frequently than every nine (9) months.
- Dentist fills in the usual claim form, but in addition notes that the faculty member has gum disease or specifies the other dental problem that requires more frequent cleaning.
- Faculty member or dentist submits the form to the insurance carrier as normal.
- The insurance carrier determines if the reasons set out by the dentist fit within the approved reasons under the Dental plan for having teeth cleaned more frequently than every nine (9) months.

The Employers' approval of the more frequent cleaning is not required.

APPENDIX H FAMILY MEMBERS FOR THE PURPOSE OF ARTICLE 18.8 COMPASSIONATE CARE LEAVE

1. The following "family members" are persons identified through their relationship to the faculty member.

• Spouse (includes heterosexual, common-law, and same-sex relationships)

- Children
- Children's spouses
- Step-children
- Step-children-in-law
- Siblings
- In-law siblings
- Parents
- Step-parents
- Parents-in-law
- Grandparents
- Grandchildren
- Nieces/Nephews
- Guardians
- Step-siblings
- Aunts/Uncles
- Current or former foster-parents
- Current or former foster children
- Current or former wards
- Current or former guardians
- Spouse of sibling or step-sibling
- Spouse of child or step-child
- Spouse of a grandparent
- Spouse of a grandchild
- Spouse of an aunt or uncle
- Spouse of a niece or nephew
- Spouse of a current or former foster child
- Spouse of a current or former guardian
- Spouse of a faculty member's current or former foster parent
- Spouse of a faculty member's current or former ward
- Spouse of a person who is living with the faculty member as a member of the faculty member's family

2. The following "family members" are persons identified through their relationship to the faculty member's spouse

- Spouse's parents or step-parents
- Spouse's siblings or step-siblings
- Spouse's children
- Spouse's grandparents
- Spouse's grandchildren
- Spouse's aunts or uncles
- Spouse's nieces or nephews
- Spouse's current or former foster parents
- Spouse's current or former wards
- 3. The following "family members" are deemed family members
- Any other person in the same household who is dependent upon the faculty member
- Any person who lives with the faculty member as a member of the faculty member's family

• Whether or not related to a faculty member by blood, adoption, marriage or common-law partnership, an individual with a serious medical condition who considers the faculty member to be, or whom the faculty member considers to be, like a close relative

APPENDIX I BCGEU Program Norms

Program	Teaching Norms (hpw)
Academic and Career Preparation (ACP):	
Adult Special Education	24
Adult Basic Education	24
Aboriginal Bridging Programs	24
Literacy	24
English as a Second Language	20
The High School	25
Foundations for Success	24
Cooperative Entrepreneur Training Program	24
Health Sciences & Human Services:	
Activity Assistant (PD&T)	28***
Dental Assistant	20
Health Care Assistant	25*
Practical Nursing	22**
Trades & Applied Technology:	
Office Administration	25
Automotive	25
Carpentry	25
Culinary Arts	25
Electrical	25
Esthetics	25
Hair Stylist	25
Heavy Mechanic Trades	25
Heavy Equipment Operator	25
Horticulture	18***
Refrigeration Air-Conditioning Mechanic (RACM)	25
ITAS	20
Marine and Motorcycle Technician	25
Power Engineering	25
Professional Baking	25
Welding	25
Professional Development & Training (MOU#1)	
PD&T MOU#1 (Various Programs)	28***

* Health Programs Teaching Norms are based on an annual weighted average workload consisting of practical/clinical and class theory.

** 22 hours is the program norm for the Practical Nursing Program. This is based on a weighted formula of combined direct instruction comprised of theory and practicum with a maximum of 832 workload hours annually for direct instruction over an annual 38-week period. For example, faculty may work up to 20 hpw theory or up to 28 hpw practicum/clinical to reach an annual weighted average of 22 hours per week as the program norm. The program norm should not be met by practicum supervision only, and the department chair will endeavor to distribute the workloads equitably.

Annual Workload = FTE value (See Definitions). Hours paid will be converted to an FTE value.

Formula: Number of Direct Hours/Program Norm x Common Denominator Factor = FTE value.

*** Approved exception to Program Norm.

MEMORANDUM OF UNDERSTANDING 1 For Courses and Programs Not Approved by Senate

BCGEU jurisdiction extends to some courses and programs which have not been approved by Senate.

- (1) (a) Courses and programs appropriate for this memorandum are:
 - (i) developed for a sponsoring agency;
 - (ii) for upgrading of skills for employability or skill enhancement;
 - (iii) not a pre-requisite for regular programming;

(iv) not connected to, nor form a part of, courses and/or programs approved by Senate.

(b) For people employed in these courses and/or programs, only the following terms and conditions will apply:

- (i) Seniority
 - a. Employees and faculty members (whether regular, term or auxiliary) employed under this memorandum shall accrue seniority only for the purpose of advancing on the salary scale as set out at (1)(b)(ii).
 - b. Seniority accrued under this memorandum is not transferable.
- (ii) Salary
 - a. Employees and faculty members (whether regular, term or auxiliary) shall be placed at the minimum step of the current salary grid for courses and programs under this Memorandum of Understanding.
 - b. Upon accruing 1820 hours (as accrued in the Payroll system) the employee shall advance one increment step on the salary scale. An increment will be effective on the first day of the pay period following the date on which the increment is earned.
 - c. Any work under this memorandum resulting in overtime shall be paid at straight time.
- (iii) Assignment of work
 - a. Courses and programs include direct and indirect instructional components.
 - b. The workload shall be based on the program norm of 28 hours per week or 56 hours bi-weekly.
 - c. Seniority accrued by regular and term faculty members employed under this memorandum is not transferable to their regular position and will only be applicable as stated at (1)(b)(ii) of this memorandum.

- d. Regular or term faculty members may be employed under this memorandum, providing the hours do not conflict with the regular assignment. Any work resulting in overtime shall be paid at straight time.
- (iv) A travel stipend may be paid at the sole discretion of the employer.

(2) (a) Community interest courses and/or programs are excluded from this memorandum. Community interest courses and programs have:

- (i) No assessment.
- (ii) No certificate, other than for participation.
- (iii) No record of learning outcomes.

(b) Employees hired to instruct these courses and/or programs are outside the jurisdiction of the Collective Agreement.

- (3) Exceptions
 - (a) Fee for service or contractor status
 - (b) Special expertise/"name" speaker
 - (c) Brokered programs for other institutional education providers
 - (d) Co-sponsored/partnership courses
 - (e) Guest lecturers

It is recognized that these exceptions should be reviewed, from time to time, by the Labour Management Committee. Any changes to the list of exceptions shall be by mutual agreement.

(4) Accountability

In order to satisfy the intent of this memorandum, the Labour Management Committee will ensure accountability through reporting and transparency. The Committee will be provided with the following:

- (a) Annually, a report consisting of the following:
- (i) List of people (names)
- (ii) Hours worked
- (iii) List of courses and programs
- (iv) Rate of pay
- (v) Start and end dates
- (vi) Exceptions and rationale
- (b) Quarterly, the following information will be provided:

(i) All courses and programs offered in the previous quarter. Quarterly reporting dates are October, January, April and July.

The Committee, by mutual agreement, may vary these reports.

MEMORANDUM OF UNDERSTANDING 2 ESL Programs

The Employer and the Union agree to allow volunteers to participate in the ESL classes under the conditions as set out below.

(1) Volunteer(s) will only attend regularly scheduled classes. For example, a volunteer would not conduct a class or run a learning centre on their own.

(2) Volunteer(s) will only be in the classroom at the faculty member's discretion.

(3) Volunteer(s) will take direction from faculty members.

(4) Any orientation or training that is identified as required will be undertaken prior to the volunteer's participation in classroom activity.

(5) Volunteer(s) will not be part of an ESL training program to certify new ESL faculty members.

(6) The use of volunteers will not reduce or cause a reduction to normal faculty members' workload assignments.

This Agreement will in no way prejudice any other Agreement between the Union and the Employer.

LETTER OF UNDERSTANDING 1 Prior Learning Assessment

Effective August 1, 2002

Definition:

Prior Learning Assessment (PLA) is the assessment by some valid and reliable means, of what has been learned through formal and non-formal education/training or experience, that is worthy of credit in a course or program offered by the institution providing credit.

The assessment and evaluation of prior learning and the determination of competency and credit awarded, will be done by faculty members who have the appropriate subject matter expertise but other staff in an institution may have a supporting role in the process.

The work required for prior learning assessment includes but is not limited to: classroom-based and individual advising; classroom-based and individual assessment, training and upgrading; development of assessment tools; and training in the use of flexible assessment.

Training in Prior Learning Assessment:

A faculty member required to perform prior learning assessment responsibilities as part of their workload, has a right to employer-paid training time and expenses, in the methodology and application of prior learning assessment as necessary for the assigned task.

Process:

PLA assignments for up to two (2) courses may be assigned to a full-time faculty member as part of regular indirect instructional responsibilities (14.2[b]). The number of PLA assignments included in regular indirect instructional responsibilities will be pro-rated for less than full-time appointments (e.g., if half time one [1] PLA assignment). PLA assignments which are part of normal indirect instructional responsibilities are

considered part of normal workload for full-time faculty with no additional compensation to individual faculty members.

NB: A PLA assignment is considered to be an assessment of one (1) student for one (1) course by one (1) faculty member.

PLA assignments beyond two (2) courses a year may be provided on a voluntary basis. Any individual full time faculty member may accept up to a maximum of ten (10) PLA assignments per year. Acceptance of PLA assignments for more than ten (10) courses in a year requires the written approval of the BCGEU shop steward and the Provost and Vice-President, Academic.

Additional PLA assignments (beyond two [2] courses) will be compensated as follows:

1. (a) A stipend of one hundred and twenty-five dollars (\$125.00) for a Full PLA Assignment (assignment includes advising, support, development of assessment tools and assessment) for one (1) student in one (1) course under seventy-five (75) hours.

(b) A stipend of two hundred and fifty dollars (\$250.00) for a Full PLA Assignment for one (1) student in one (1) course of seventy-five to one hundred and fifty (75 - 150) hours.

(c) A stipend of three hundred and seventy-five dollars (\$375.00) for a Full PLA Assignment for one (1) student in one (1) course over one hundred and fifty (150) hours.

2. A stipend of two hundred and fifty dollars (\$250.00) for one (1) student in a Clinical Workplace Based Assignment.

3. A stipend of four hundred and fifty dollars (\$450.00) for one (1) Comprehensive Assessment of one (1) student for three (3) or more courses.

4. A stipend of fifty dollars (\$50.00) for a PLA Challenge Examination involving a pre-existing format (with minor revisions to existing assessment tool) for one (1) course.

5. A stipend of one hundred dollars (\$100.00) for a PLA Challenge Examination involving a customized format (exam designed specifically for individual assessment) for one (1) course.

The faculty member will be paid the full amount of the appropriate stipend whether or not the student completes the PLA process. Payment to the faculty member will be made on the first pay period following the confirmation of the student's PLA registration.

When PLA assignments in an instructional area exceed ten (10) course assignments, the Dean, Director or Campus Administrator may, at their discretion and with the mutual agreement of the faculty member and the Department Chair, provide release time in lieu of payment to a faculty member qualified to accept these assignments. Release time shall be negotiated by the Dean, Director, or Campus Administrator and faculty member involved, and will normally occur within the fiscal year in which it was earned.

LETTER OF UNDERSTANDING 2 Partial Sick Leave and Partial Disability Benefits

The parties agree that it is in the interests of both the faculty member and the Employer to enable an faculty member to remain at work when the faculty member is only partially disabled.

"Partially disabled" for the purpose of this Letter of Understanding means that the faculty member is unable to do a portion of their normal workload where such portion is agreed by the Employer to conform to the configuration of faculty workload in the faculty member's instructional or non-instructional areas and where the partial sick leave is in any event no greater than eighty percent (80%) of a full-time workload in that area. The application of this definition is subject to the Employer's legal duty of accommodation.

Determination of whether the faculty member is partially disabled as defined above shall be by the Short-Term Disability benefits carrier.

A faculty member who is determined to be partially disabled will be entitled to sick leave under Article 22.6 on a pro-rated basis until the faculty member has satisfied the qualifying period for Short-Term Disability benefits of the equivalent of thirty (30) complete calendar days. In any event, to qualify for Short-Term Disability benefits the faculty member must complete the qualifying period within six (6) months of the date the faculty member commenced part-time sick leave.

Should the faculty member return to their full normal duties of their own occupation during this qualifying period for Short-Term Disability benefits and then become disabled from the same or related disability within fourteen (14) consecutive calendar days after returning to full active employment, they will be considered to be within the same qualifying period.

The faculty member is required to meet all application, reporting, and other requirements provided for in this Short-Term and Long-Term Disability benefits plans as applicable.

The carrier's approval of a partial Disability claim for sick leave continuation on a pro-rata basis does not in itself mean that the faculty member's subsequent claims for Short-Term Disability benefits will be automatically approved, nor does approval for Short-Term Disability benefits mean that the faculty member's subsequent claim for Long-Term Disability benefits will be automatically approved.

Additional information on the processes and criteria for partial sick leave and partial Disability benefits are set out in the document titled "Administration of Partial Sick Leave and Partial Disability Benefits", which the parties agree shall be part of the "Policies and Procedures" sections of the Disability Management Handbook for the Disability benefits plan set out in Article 22.6 of the Collective Agreement.

LETTER OF UNDERSTANDING 3

Education Support and Professional Development Fund (ESPD)

Between

Vancouver Island University

and

BCGEU

Collectively, the "Parties"

To maintain and promote high quality education, diversity and access to vocational programs, Vancouver Island University shall provide both students and instructors with additional supports to promote teaching and learning:

The following funding allocations shall be made available:

Year	Amount
Year 1 – April 1, 2022	\$126,000
Year 2 – April 1, 2023	\$90,000
Year 3 – April 1, 2024	\$70,000

The above amounts represent ongoing additional funding. For clarity, the total annual funding allocation will be \$70,000 in Year 3 and beyond.

The Education Support and Professional Development Fund will be available to non-regular and regular BCGEU instructors and used for the following:

1. Professional development activities directly relevant to the role of the faculty members within the University, including:

- 1. studies, and
- 2. experience in business, industry, community service, government or other educational institutions.

2. Instructor Supports – Related and tangible classroom supports will be developed to improve the learning experiences of students, for example:

- a. creative application of technology;
- b. in-class cultural liaisons;
- c. training and awareness of positive mental health and

d. the ability to develop enhancements to current programs, or create new programs that improve the level of success of the diverse student.

3. Student Supports – To achieve greater levels of success within VIU programs, student support initiatives might include:

- a. peer support programs;
- b. numeracy and literacy supports;
- c. communication skills workshops;
- d. tutor supports; and
- e. study skills enhancements and time management guidance.

Applications must be approved by the faculty member's senior administrator.

Subject to the provisions of this LOU, the Education Support and Professional Development Fund Committee may establish, publish, and work within their own guidelines.

The Committee shall comprise two BCGEU members appointed by the Union and two Vancouver Island University administrators.

The Education Support and Professional Development Fund Committee shall review and approve applications for funding. The parties agree that there will be no expenditure of funds if the committee members fail to reach consensus. Funds not allocated within a calendar year will be carried over for one year only.

Information regarding the use of the funds shall be made available on a yearly basis to the bargaining unit Chair and Provost.

LETTER OF UNDERSTANDING 4 Cost of Living Adjustment

Definitions

"General Wage Increase" or "GWI" means the overall general wage increase expressed as a percentage.

"Cost of Living Adjustment" or "COLA" means a percentage-based general wage increase adjustment provided in accordance with this Memorandum of Agreement. COLA is an upward adjustment applied to and folded into all wage rates.

The "annualized average of BC CPI over 12-months" (AABC CPI) means the *Lastest 12-month Average Index % Change* reported by BC Stats in March for British Columbia for the 12-months starting at the beginning of March in the preceding year and concluding at the end of the following February.

The "Latest 12-month Average Index", as defined by BC Stats, in a 12-month moving average of the BC consumer price indexes of the most recent 12-months. This figure is calculated by averaging index levels over the applicable 12-months.

The Latest 12-month Average Index % Change is reported publicly by BC Stats in the monthly BC Stats *Consumer Price Index Highlights* report. The BC Stats *Consumer Price Index Highlights* report released in mid-March will contain the applicable figure for the 12-months concluding at the end of February. The percentage change reported by BC Stats that will form the basis for determining any COLA increase is calculated to one decimal point. For reference purposes only, the annualized average of BC CPI over 12months from March 1, 2021 to February 28, 2022 was 3.4%.

COLA

The COLA will be applied as applicable to the GWI effective on April 1, 2023 and April 1, 2024. The COLA will be calculated by determining the difference between the AABC CPI and the annual general wage increase to the maximum COLA prescribed that year in Wage Schedule – Grids.

April 2023

If the 2023 AABC CPI exceeds the April 2023 GWI of 5.5%, then, on April 1, 2023 the April 2023 GWI will be adjusted upwards to reflect a COLA equal to the difference between the April 2023 GWI and the 2023 AABC CPI up to a maximum of 1.25%.

April 2024

If the 2024 AABC CPI exceeds the April 2024 GWI of 2.0%, then, on April 1, 2024 the April 2024 GWI will be adjusted upwards to reflect a COLA equal to the difference between the April 2024 GWI and the 2024 AABC CPI up to a maximum of 1.0%.

LETTER OF AGREEMENT 1 Suspension of Seniority Credit Adjustments for faculty members on LTD beyond Change of Definition

The parties agree to suspend seniority credit adjustments, as provided for under BCGEU Article 18.3, for those BCGEU regular faculty members who are in receipt of Long Term Disability (LTD) benefits and have been deemed to be disabled from all occupations as a result of the "Change of Definition" (COD) decision by the disability benefits carrier, Manulife Financial. This seniority suspension is in effect commencing the first pay period in January 2009.

Should an LTD-COD faculty member subsequently return to active VIU employment as a BCGEU regular member, the Human Resources Department will calculate and apply BCGEU seniority credits that would have otherwise been earned from the period that seniority credit adjustments were suspended to the date the faculty member returns to active employment.

The intent of this agreement is not to remove any rights of the BCGEU member as a result of suspending their seniority credit adjustments.

LETTER OF AGREEMENT 2

The parties agree that to support future local bargaining, discussions regarding contact time/instructional hours, including the use of distributed learning methodologies, it would be beneficial for the Employer and the bargaining unit to discuss matters of mutual concern at the regular Labour Management Committee meetings. These discussions may include the following: development time, class size, contact hours, non-contact hours (email correspondence, marking, exam invigilation, etc.), intake methodologies, delivery methods, and training for instructors.

LETTER OF AGREEMENT 3 Vacation Carry Over

Notwithstanding Article 17.3, vacation leave banked prior to the date of ratification of this Agreement shall remain available to faculty members until such times as it is utilized in accordance with Article 17.2.

LETTER OF AGREEMENT 4 FTE Calculations

During collective bargaining, the parties discussed the calculation of FTE value and Common Denominators Factor as set out in Appendix I of the Collective agreement. The parties agree it would be beneficial to discuss those calculations further as part of BCGEU-VIU Labour Management during the life of this agreement.

SECTION II COMMON COLLECTIVE AGREEMENT

The Common Agreement between the following Employers (as represented by the Post-Secondary Employers' Association):

Camosun College Northern Lights College Coast Mountain College Okanagan College Selkirk College And the B.C. General Employees' Union

For your reference the current Common Agreement is available on the VIU (Vancouver Island University) website.

Vancouver Island University is not a party to the Common Agreement and it has no affect at Vancouver Island University

07020125v2

docusign

Certificate Of Completion

Envelope Id: F2BA1C6D-3DC7-4BDE-A4A9-459C4E52E33D Subject: VIU Collective Agreement (April 1, 2022 to March 31, 2025) for Signatures Source Envelope: Document Pages: 94 Signatures: 12 Initials: 0 Certificate Pages: 6 AutoNav: Enabled EnvelopeId Stamping: Enabled Time Zone: (UTC-08:00) Pacific Time (US & Canada)

Record Tracking

Status: Original 1/30/2025 10:03:48 AM

Signer Events

Brad Jannaway

Brad.Jannaway@viu.ca

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure:

Accepted: 1/30/2025 10:21:09 AM ID: 9e5a8b10-afaf-4fff-8e96-d35e68c9c487

Dean Cadieux

Dean.Cadieux@viu.ca Security Level: Email, Account Authentication

(None)

Electronic Record and Signature Disclosure: Accepted: 1/30/2025 10:36:31 AM

ID: 8ba37c0a-2692-4ad9-bdda-5d6ec7fb543c

Malena Acosta Malena.Acosta@viu.ca Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 1/30/2025 10:48:51 AM ID: c84cc086-0925-4dec-bb00-24ca0abce04d

Deborah Saucier Deborah.Saucier@viu.ca President and Vice Chancellor Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 1/30/2025 2:38:39 PM ID: cfbcde3d-9ae3-441d-b2f5-44f7b4225a11 Holder: Doris Stowe doris.stowe@bcgeu.ca

Signature Signed by: Brad Jannaway

F4E4FF8B5DD7421...

Signature Adoption: Pre-selected Style Using IP Address: 104.128.253.250



Signed by

Malena Acosta

6307961562E0484

Signature Adoption: Pre-selected Style Using IP Address: 184.151.230.164 Signed using mobile

Signature Adoption: Pre-selected Style Using IP Address: 104.128.253.255

Sent: 1/30/2025 10:23:38 AM Viewed: 1/30/2025 10:36:31 AM Signed: 1/30/2025 10:38:29 AM

Sent: 1/30/2025 10:38:35 AM Viewed: 1/30/2025 10:48:51 AM Signed: 1/30/2025 10:49:33 AM

Sianed by: Neborali Saucier E0BEBD7B52274FC

Signature Adoption: Pre-selected Style Using IP Address: 104.28.116.2

Sent: 1/30/2025 10:49:38 AM Viewed: 1/30/2025 2:38:39 PM Signed: 1/30/2025 2:38:55 PM

DocuSigned by



Status: Completed

Envelope Originator: Doris Stowe HQ IT Services 4911 Canada Way Burnaby, BC V5G 3W3 doris.stowe@bcgeu.ca IP Address: 20.236.201.102

Location: DocuSign

Timestamp

Sent: 1/30/2025 10:15:00 AM Viewed: 1/30/2025 10:21:09 AM Signed: 1/30/2025 10:23:33 AM

Signer Events

Patricia Elliott Patricia.Elliott@viu.ca Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 1/30/2025 2:40:56 PM ID: 31995f20-1701-4dfa-a137-a56a49b63d5d

Glynis Steen

Glynis.Steen@viu.ca

Dean, Trades and Applied Technology Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 1/31/2025 9:20:47 AM ID: 3e507f2e-4d4c-416f-b254-63a307fa0c7d

jeannie Maltesen

Jean.Maltesen@viu.ca

Provost and VP Academic

Vancouver Island University

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 1/31/2025 2:16:08 PM

ID: bf85742b-1dc3-4c4c-a27d-6f49c13bec47

Jessie Magee-Chalmers

Jessie.MageeChalmers@viu.ca

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 2/1/2025 10:44:16 AM

ID: d3d828ce-5363-41b2-b853-fa511320412e

Lynda Phillips

Lynda.Phillips@viu.ca

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 2/3/2025 7:55:05 PM

ID: d5d1c731-a667-4cae-beaf-c78d5bf4841c

Doreen Smith

Doreen.Smith@bcgeu.ca Staff Representative BCGEU Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure:

Signature

TP F299D8154F44A1

Signature Adoption: Drawn on Device Using IP Address: 104.128.253.249

Signed by: Glynis Steen 4FREEED9804E8

Signature Adoption: Pre-selected Style Using IP Address: 104.128.253.16

Sent: 1/30/2025 2:41:59 PM Viewed: 1/31/2025 9:20:47 AM Signed: 1/31/2025 9:21:19 AM

DocuSigned by AMalteser 936B2162D0A74DE

Signature Adoption: Uploaded Signature Image Using IP Address: 24.69.69.166

Sent: 1/31/2025 9:21:26 AM Viewed: 1/31/2025 2:16:08 PM Signed: 1/31/2025 2:17:44 PM

Signed by: Jessie Magee-Chalmers -A1B20A2224324DF...

Signature Adoption: Pre-selected Style Using IP Address: 173.180.87.234

Sent: 1/31/2025 2:17:50 PM Viewed: 2/1/2025 10:44:16 AM Signed: 2/1/2025 10:44:31 AM

lynda phillips

Signature Adoption: Pre-selected Style Using IP Address: 24.84.164.123

Sent: 2/1/2025 10:44:36 AM Viewed: 2/3/2025 7:55:05 PM Signed: 2/3/2025 7:55:49 PM

DocuSigned by: Doreen Snoth F9AD968E78E24DD..

Signature Adoption: Pre-selected Style Using IP Address: 184.66.98.234

Sent: 2/3/2025 7:55:56 PM Viewed: 2/4/2025 8:22:11 AM Signed: 2/4/2025 8:24:25 AM

Timestamp

Sent: 1/30/2025 2:39:00 PM Viewed: 1/30/2025 2:40:56 PM Signed: 1/30/2025 2:41:54 PM

Signer Events

Not Offered via Docusign

Doug Campbell Doug.dmconsulting@outlook.com Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 2/4/2025 8:35:07 PM ID: 0cd6afa0-9f1e-466d-83a4-bfce8d2cc88f

Paul Finch

Paul.Finch@bcgeu.ca president BCGEU

Security Level: Email, Account Authentication (None)

Electronic Record and Signature Disclosure: Accepted: 2/4/2025 9:02:02 PM ID: 2f8a5e4c-d5b1-4617-abc6-beef15d0fe7d

In Barcon Signar Eventa

Signature

DocuSigned by: Doug Campbell 16AB74F0439

Signature Adoption: Pre-selected Style Using IP Address: 206.116.247.249

Timestamp

Sent: 2/4/2025 8:24:31 AM Viewed: 2/4/2025 8:35:07 PM Signed: 2/4/2025 8:35:22 PM

DocuSigned by: Decu Science 0B69046EF98F41B...

Signature Adoption: Drawn on Device Using IP Address: 209.53.26.194 Signed using mobile Sent: 2/4/2025 8:35:28 PM Viewed: 2/4/2025 9:02:02 PM Signed: 2/4/2025 9:02:11 PM

In Person Signer Events	Signature	Timestamp
Editor Delivery Events	Status	Timestamp
Agent Delivery Events	Status	Timestamp
Intermediary Delivery Events	Status	Timestamp
Certified Delivery Events	Status	Timestamp
Carbon Copy Events	Status	Timestamp
Witness Events	Signature	Timestamp
Notary Events	Signature	Timestamp
Envelope Summary Events	Status	Timestamps
Envelope Sent Certified Delivered Signing Complete Completed	Hashed/Encrypted Security Checked Security Checked Security Checked	1/30/2025 10:15:00 AM 2/4/2025 9:02:02 PM 2/4/2025 9:02:11 PM 2/4/2025 9:02:11 PM
Payment Events	Status	Timestamps
Electronic Record and Signature Disclosure		

ELECTRONIC RECORD AND SIGNATURE DISCLOSURE

From time to time, British Columbia General Employees' Union (we, us or Company) may be required by law to provide to you certain written notices or disclosures. Described below are the terms and conditions for providing to you such notices and disclosures electronically through the DocuSign system. Please read the information below carefully and thoroughly, and if you can access this information electronically to your satisfaction and agree to this Electronic Record and Signature Disclosure (ERSD), please confirm your agreement by selecting the check-box next to 'I agree to use electronic records and signatures' before clicking 'CONTINUE' within the DocuSign system.

Getting paper copies

At any time, you may request from us a paper copy of any record provided or made available electronically to you by us. You will have the ability to download and print documents we send to you through the DocuSign system during and immediately after the signing session and, if you elect to create a DocuSign account, you may access the documents for a limited period of time (usually 30 days) after such documents are first sent to you. After such time, if you wish for us to send you paper copies of any such documents from our office to you, you will be charged a \$0.00 per-page fee. You may request delivery of such paper copies from us by following the procedure described below.

Withdrawing your consent

If you decide to receive notices and disclosures from us electronically, you may at any time change your mind and tell us that thereafter you want to receive required notices and disclosures only in paper format. How you must inform us of your decision to receive future notices and disclosure in paper format and withdraw your consent to receive notices and disclosures electronically is described below.

Consequences of changing your mind

If you elect to receive required notices and disclosures only in paper format, it will slow the speed at which we can complete certain steps in transactions with you and delivering services to you because we will need first to send the required notices or disclosures to you in paper format, and then wait until we receive back from you your acknowledgment of your receipt of such paper notices or disclosures. Further, you will no longer be able to use the DocuSign system to receive required notices and consents electronically from us or to sign electronically documents from us.

All notices and disclosures will be sent to you electronically

Unless you tell us otherwise in accordance with the procedures described herein, we will provide electronically to you through the DocuSign system all required notices, disclosures, authorizations, acknowledgements, and other documents that are required to be provided or made available to you during the course of our relationship with you. To reduce the chance of you inadvertently not receiving any notice or disclosure, we prefer to provide all of the required notices and disclosures to you by the same method and to the same address that you have given us. Thus, you can receive all the disclosures and notices electronically or in paper format through the paper mail delivery system. If you do not agree with this process, please let us know as described below. Please also see the paragraph immediately above that describes the consequences of your electing not to receive delivery of the notices and disclosures electronically from us.

How to contact British Columbia General Employees' Union:

You may contact us to let us know of your changes as to how we may contact you electronically, to request paper copies of certain information from us, and to withdraw your prior consent to receive notices and disclosures electronically as follows: To contact us by email send messages to: christopher.firstner@bcgeu.ca

To advise British Columbia General Employees' Union of your new email address

To let us know of a change in your email address where we should send notices and disclosures electronically to you, you must send an email message to us at christopher.firstner@bcgeu.ca and in the body of such request you must state: your previous email address, your new email address. We do not require any other information from you to change your email address.

If you created a DocuSign account, you may update it with your new email address through your account preferences.

To request paper copies from British Columbia General Employees' Union

To request delivery from us of paper copies of the notices and disclosures previously provided by us to you electronically, you must send us an email to christopher.firstner@bcgeu.ca and in the body of such request you must state your email address, full name, mailing address, and telephone number. We will bill you for any fees at that time, if any.

To withdraw your consent with British Columbia General Employees' Union

To inform us that you no longer wish to receive future notices and disclosures in electronic format you may:

i. decline to sign a document from within your signing session, and on the subsequent page, select the check-box indicating you wish to withdraw your consent, or you may;

ii. send us an email to christopher.firstner@bcgeu.ca and in the body of such request you must state your email, full name, mailing address, and telephone number. We do not need any other information from you to withdraw consent.. The consequences of your withdrawing consent for online documents will be that transactions may take a longer time to process..

Required hardware and software

The minimum system requirements for using the DocuSign system may change over time. The current system requirements are found here: <u>https://support.docusign.com/guides/signer-guide-signing-system-requirements</u>.

Acknowledging your access and consent to receive and sign documents electronically

To confirm to us that you can access this information electronically, which will be similar to other electronic notices and disclosures that we will provide to you, please confirm that you have read this ERSD, and (i) that you are able to print on paper or electronically save this ERSD for your future reference and access; or (ii) that you are able to email this ERSD to an email address where you will be able to print on paper or save it for your future reference and access. Further, if you consent to receiving notices and disclosures exclusively in electronic format as described herein, then select the check-box next to 'I agree to use electronic records and signatures' before clicking 'CONTINUE' within the DocuSign system.

By selecting the check-box next to 'I agree to use electronic records and signatures', you confirm that:

- You can access and read this Electronic Record and Signature Disclosure; and
- You can print on paper this Electronic Record and Signature Disclosure, or save or send this Electronic Record and Disclosure to a location where you can print it, for future reference and access; and
- Until or unless you notify British Columbia General Employees' Union as described above, you consent to receive exclusively through electronic means all notices, disclosures, authorizations, acknowledgements, and other documents that are required to be provided or made available to you by British Columbia General Employees' Union during the course of your relationship with British Columbia General Employees' Union.